JONATHAN

AN DER KAMP

SOCIAL COST-BENEFIT ANALYSIS OF AIR POLLUTION CONTROL MEASURES



Advancing environmental-economic assessment methods to evaluate industrial point emission sources



Jonathan van der Kamp

Social cost-benefit analysis of air pollution control measures

Advancing environmental-economic assessment methods to evaluate industrial point emission sources

Social cost-benefit analysis of air pollution control measures

Advancing environmental-economic assessment methods to evaluate industrial point emission sources

by Jonathan van der Kamp



Dissertation, Karlsruher Institut für Technologie KIT-Fakultät für Wirtschaftswissenschaften

Tag der mündlichen Prüfung: 03. Mai 2017 Gutachter: Prof. Dr. rer. nat. Ute Karl Prof. Dr. rer. pol. Frank Schultmann

Impressum



Karlsruher Institut für Technologie (KIT) KIT Scientific Publishing Straße am Forum 2 D-76131 Karlsruhe

KIT Scientific Publishing is a registered trademark of Karlsruhe Institute of Technology. Reprint using the book cover is not allowed.

www.ksp.kit.edu



This document – excluding the cover, pictures and graphs – is licensed under a Creative Commons Attribution-Share Alike 4.0 International License (CC BY-SA 4.0): https://creativecommons.org/licenses/by-sa/4.0/deed.en

The cover page is licensed under a Creative Commons Attribution-No Derivatives 4.0 International License (CC BY-ND 4.0): https://creativecommons.org/licenses/by-nd/4.0/deed.en

Print on Demand 2017 – Gedruckt auf FSC-zertifiziertem Papier

ISBN 978-3-7315-0697-3 DOI 10.5445/KSP/1000072046

Social cost-benefit analysis of air pollution control measures

Advancing environmental-economic assessment methods to evaluate industrial point emission sources

zur Erlangung des akademischen Grades eines

Doktors der Wirtschaftswissenschaften (Dr. rer. pol.)

von der Fakultät für Wirtschaftswissenschaften des Karlsruher Instituts für Technologie (KIT)

genehmigte

Dissertation

von

Dipl.-Wi.-Ing. Jonathan van der Kamp

aus Karlsruhe

Tag der mündlichen Prüfung: 03. Mai 2017 Erster Gutachter: Prof. Dr. rer. nat. Ute Karl Zweiter Gutachter: Prof. Dr. rer. pol. Frank Schultmann

Kurzfassung

Luftschadstoffe erhöhen das Risiko für Atemwegs- und Herz-Kreislauf-Erkrankungen sowie das Sterblichkeitsrisiko der Bevölkerung und führen somit zu gesellschaftlichen Kosten. Wohlfahrtsorientierte Politik zielt darauf ab, diese Kosten in Marktpreise zu integrieren, z.B. durch Grenzwerte für industrielle Luftschadstoffemissionen. Gemäß der neoklassischen ökonomischen Theorie sollten Emissionen verringert werden, bis die damit verbundenen Grenzkosten und -nutzen gleich groß sind. Dieses Gleichgewicht kann durch die gesellschaftliche Kosten-Nutzen-Analyse (KNA) bestimmt werden, welche auf eine effiziente Verteilung knapper Ressourcen und die Vermeidung unverhältnismäßiger Kosten abzielt. Hierdurch wird die KNA auch für betroffene Industriesektoren interessant.

Ansätze für die gesellschaftliche KNA von Emissionsminderungsmaßnahmen sind im politischen Entscheidungskontext weit verbreitet, da entsprechende Analysen verpflichtend zur Vorbereitung neuer Gesetze sind. Ansätze zur Entscheidungsunterstützung auf der betrieblichen Standortebene sind jedoch rar. Existierende Bewertungsmodelle erfassen die zeitliche und räumliche Variabilität von Eingangsdaten nur unzureichend.

Die vorliegende Arbeit beabsichtigt diese Forschungslücke durch methodische Weiterentwicklungen für die gesellschaftliche KNA von Emissionsminderungsmaßnahmen zu schließen. Der Fokus liegt auf der Bewertung von Gesundheitsschäden. Zentrale Zielsetzungen sind der Transfer der gesellschaftlichen KNA auf die Standortebene, die Analyse des Einflusses von Modell- und Methodenparametern auf die Ergebnisse und die Ableitung methodischer Empfehlungen für private Entscheidungsträger unter Berücksichtigung von Unsicherheiten.

Hierzu wird der sogenannte Wirkungspfadansatz detailliert untersucht und mittels eines neuen methodischen Ansatzes zur Gesundheitsschadensbewertung implementiert. Dieser beruht auf der Einbindung von zeitlich und räumlich hoch aufgelösten Schadstoffausbreitungsrechnungen. Für die gesellschaftliche KNA wird die Gesundheitsschadensbewertung mit einem betriebswirtschaftlichen Ansatz zur Abschätzung der privaten Emissionsminderungskosten kombiniert.

Die zeitliche und räumliche Modellauflösung, in Verbindung mit Emissionsquellcharakteristiken und der Bevölkerungsdichte, werden durch Sensitivitätsanalysen als zentrale, ortsspezifische Einflussfaktoren auf die Gesundheitsschäden identifiziert. Die Berücksichtigung NO₂-expositionsbezogener Sterblichkeit führt zu einer deutlichen Erhöhung der Schäden in Metropolregionen, hängt jedoch vom Konzentrationsschwellwert, der Emissionsquelle und der Modellauflösung ab. Der größte Teil der quantifizierten Schäden betrifft Sterblichkeitsrisiken durch eine langfristige Feinstaubexposition, weshalb damit verbundene methodische Aspekte besonders einflussreich sind.

Die neu entwickelte Methodik verringert Unsicherheiten, stellt jedoch hohe Ressourcenanforderungen. Das vertretbare Unsicherheitsniveau und der zu bevorzugende methodische Ansatz richten sich somit nach dem Entscheidungskontext. Die vorliegende Arbeit zeigt, dass Sensitivitätsanalysen die Belastbarkeit der Ergebnisse erhöhen. Die Ergebnisse unterstreichen weiterhin die Bedeutung methodischer Leitlinien, um Willkür bei der gesellschaftlichen KNA zu vermeiden und somit die Berücksichtigung von Gesundheitsschadenskosten in Entscheidungssituationen zu erleichtern.

Abstract

Exposure to ambient air pollution increases the risk for humans of developing respiratory or cardiovascular diseases as well as of dying prematurely, thus imposing costs on society. Welfare-oriented policy-making aims at integrating these costs into market prices, e.g. by reducing harmful atmospheric emissions from industrial sources. Following neoclassical economic theory, emissions should be reduced up to the point where the marginal costs equal the marginal benefits of emission control. Social cost-benefit analysis (CBA) is a method that serves to approximate this equilibrium, aiming at spending scarce resources efficiently and avoiding disproportionate costs. This principle also bears an interest for concerned industry sectors, as it can under certain conditions be used to apply for exemptions from too stringent regulations.

Scientific methods for social CBA related to air pollution at an aggregate level are well developed in the public policy context, as they are a mandatory support for the preparation of new regulations. Yet, approaches for decision-support at the level of industrial point emission sources are rare. Existing models do not properly capture temporal variations in emission patterns and spatial variations in population density.

This thesis aims at filling this gap by enhancing economic methods for social CBA of emission control measures, notably regarding health damage cost assessment. The main objectives are to transfer the social CBA method to the site level, test the influence of key modelling and methodological choices on health damage costs, and derive recommendations with regard to uncertainty and private sector decision-making.

To achieve these objectives, the so-called impact pathway approach is comprehensively reviewed and implemented into a new modelling framework for health damage cost assessment of classical air pollutants. It makes use of air pollutant dispersion modelling of high temporal and spatial resolution. For the purpose of social CBA, health damage cost assessment is combined with a private cost characterisation method.

Case study results confirm that temporal and spatial atmospheric modelling resolution, in conjunction with emission source characteristics and population distribution, are key site-dependent influencing factors on health damage costs. Including NO₂ exposure-related mortality considerably increases results in metropolitan areas, however depending on further factors such as concentration threshold, emission source characteristics, and modelling resolution. The largest share of health damage costs is linked to premature

mortality from long-term PM_{2.5} exposure, making related methodological choices particularly influential.

The newly developed approach decreases uncertainty, however at the expense of considerable resource requirements. The tolerable level of uncertainty and the preferred assessment approach thus depend on the decision context. This thesis shows that sensitivity analyses of key influencing factors increase the robustness of results. The results underline the importance of methodological guidance from official bodies to avoid arbitrary choices in social CBA, thus facilitating the integration of health damage costs into decision-making.

Table of Contents

Ku	rzfass	ung		i
Ab	stract			iii
Lis	t of Fi	gures		xi
Lis	t of Ta	ables		xvii
Ab	brevia	ations		xxi
Pre	eface.			xxv
1	Intro	ductio	n	1
	1.1	Contex	xt and problem definition	1
	1.2	Goal d	efinition	3
	1.3	Appro	ach and related contents	4
2	Back	ground	: fossil fuel-fired power generation under changing economic	
_	and e	environ	imental conditions	7
	2.1	The Eu	ropean climate and energy policy framework	7
	2.2	Histori	ic strengths of fossil fuel-fired power generation	9
		2.2.1	Security of supply	9
		2.2.2	Competitiveness	10
	2.3	Supply	and demand, structural changes and a tendency of increasing	
		flexibil	lity	12
	2.4	Atmos	pheric emissions and environmental impacts	14
		2.4.1	Greenhouse gas emissions: contribution of fossil fuel power	
			plants and regulatory framework	15
		2.4.2	Classical air pollutant emissions: contribution of fossil fuel power	
			plants	16
		2.4.3	Formation mechanisms of air pollutants	18
		2.4.4	Environmental and health impacts caused by classical air	
			pollutants	21
		2.4.5	The regulatory framework targeting classical air pollutant	
			emissions	22
		2.4.6	Best available techniques, the industrial emissions directive and	
			the concept of disproportionate costs	24
	2.5	Summ	ary of chapter 2	25

3	Met	hods: e	economic assessment of emission control measures	27
	3.1	Enviro	onmental economic theory underlying the assessment of emission	
		contro	ol measures	28
		3.1.1	Defining environmental economics	28
		3.1.2	The neoclassical approach to dealing with environmental	
			problems	29
		3.1.3	Rationale for employing environmental-economic methods in the	
			context of industrial production activities	30
		3.1.4	Limitations of the neoclassical approach	30
		3.1.5	Definitions related to externalities theory	31
		3.1.6	Production externalities and internalization through policy	
			instruments	32
	3.2	Cost-k	penefit analysis of emission control measures	34
		3.2.1	The socially "optimal" level of pollution	34
		3.2.2	The decision rule in social CBA	35
		3.2.3	Discounting in CBA	36
	3.3	Techn	o-economic assessment of emission control measures	37
		3.3.1	Cost characterization of emission control measures using the	
			annuity method	38
	3.4	Enviro	onmental-economic assessment of emission control measures	39
		3.4.1	Estimating marginal damage costs using the impact pathway	
			approach	40
		3.4.2	Distinguishing two policy-relevant types of health assessment	42
		3.4.3	Uncertainty underlying the environmental-economic assessment	
			of health damage costs	45
	3.5	State-	of-the-art review including gap analysis: economic assessment of	
		emiss	ion control measures	48
		3.5.1	Early pioneering work	50
		3.5.2	Costing methodologies	51
		3.5.3	The Externalities of Energy project series	52
		3.5.4	The GAINS model and its role in European policy-making	52
		3.5.5	Use of CBA for policy support and at power plant level	53
		3.5.6	Social CBA in the presence of variable operation profiles	54
		3.5.7	Models for health damage cost assessment related to	
			atmospheric emissions from point sources	55
		3.5.8	Overview: transferring a decision-making framework from the	
			public policy to the business domain	58
	3.6	Summ	nary of chapter 3	59

4	Health damage cost assessment: in-depth methodology review				
	and	adapta	tion	61	
	4.1	Conte	xt and objectives	61	
	4.2	Appro	pach	62	
	4.3	Mode	Is and data used	63	
		4.3.1	The tool EcoSenseWeb, its case study application and updates	63	
		4.3.2	Exposure modelling	63	
		4.3.3	Particle toxicity	64	
		4.3.4	Impact functions	65	
		4.3.5	Monetary valuation	69	
	4.4	Result	S	71	
		4.4.1	Development of health damage costs over time	71	
		4.4.2	Quantitative analysis on the influence of single elements in the		
			assessment chain	72	
	4.5	Summ	nary of chapter 4	74	
5	Exen	nplary	application of social cost-benefit analysis using an existing model		
	and	weak p	point analysis	75	
	5.1	Introd	luction	75	
	5.2	Mode	I: damage cost assessment using EcoSenseWeb	75	
		5.2.1	Classical air pollutants	76	
		5.2.2	Trace pollutants	77	
		5.2.3	Greenhouse gases	77	
		5.2.4	Settings for calculations	78	
	5.3	Analysis of the influence of the geographic location on the damage costs 80			
		5.3.1	Case study: retrofitting a coal-fired power plant with a DeNOx		
			system	80	
		5.3.2	Scope and discounting	82	
		5.3.3	Determining private costs	83	
		5.3.4	Determining damage costs	83	
	5.4	Result	s and discussion	84	
		5.4.1	Costs	84	
		5.4.2	Benefits (avoided marginal damage costs)	84	
		5.4.3	CBA results	85	
		5.4.4	CBA results – sensitivity analysis	86	
	5.5	Limita	tions and weak point analysis	88	
		5.5.1	Practical considerations regarding the application of the		
			"disproportionate" cost criterion related to industrial emission		
			control technologies	88	

		5.5.2	Weak point analysis of the EcoSenseWeb model	89
	5.6	Summ	ary of chapter 5	91
6	Deve	elopme	nt of a cost and benefit assessment methodology for emission	
	cont	rol mea	asures at point sources	93
	6.1	Develo	opment of a health damage cost assessment framework	93
		6.1.1	Air quality modelling	96
		6.1.2	Exposure assessment	. 103
		6.1.3	Health impact assessment	. 106
		6.1.4	Monetary valuation	111
	6.2	Costin	g methodology	. 115
		6.2.1	Approach taken for techno-economic cost assessment	. 115
		6.2.2	Cost characterisation of potential emission control measures at a	
			heavy fuel oil-fired power plant	. 116
	6.3	Discou	unt rates used in case studies	. 121
	6.4	Summ	ary of chapter 6	. 121
7	Appl	ication	of the extended methodology framework and results	. 123
	7.1	Explor	ing the influence of emission patterns and atmospheric modelling	
		featur	es	. 124
		7.1.1	Emission intensity	. 124
		7.1.2	Background emission inventory	. 126
		7.1.3	Temporal modelling resolution	127
		7.1.4	Spatial modelling resolution	. 130
	7.2	Exploring the influence of methodological choices in health impact		
		assess	ment	. 133
		7.2.1	Inclusion of NO ₂ -related health impacts	. 134
		7.2.2	Using an alternative approach for long-term exposure mortality	40.0
			impact assessment	. 136
		7.2.3	Lowering the concentration threshold for ozone-related health	420
		с · і	impacts	. 138
	7.3	Social	cost-benefit analysis of emission control measures	. 140
		7.3.1	Social CBA of primary and secondary emission control measures	. 140
		7.3.2	Comparison with simplified approaches for benefit assessment	. 145
	7 4	7.3.3	Sensitivity analysis	. 150
	7.4	Summ	ary or chapter 7	153
8	Discu	ussion .		. 155
	8.1	Discus	sion of the case study results	. 155
		8.1.1	Influence of atmospheric modelling features	. 155

		8.1.2	Influence of methodological choices in health impact assessment	159	
		8.1.3	Social CBA of emission control measures	161	
	8.2	Uncer	tainties in health damage cost assessment: quantitative and		
		qualita	ative evidence	162	
		8.2.1	Existing evidence on uncertainties underlying health damage cost		
			assessment	163	
		8.2.2	Quantitative uncertainty characterisation based on the results of		
			this thesis	165	
		8.2.3	Qualitative assessment of uncertainties underlying the social CBA		
			on emission control measures	172	
	8.3	Implic	ations of uncertainty on private and public decision-making	175	
		8.3.1	Implications of uncertainty for private decision-making	176	
		8.3.2	Implications of uncertainty for policy decision-making	177	
	8.4	Critica	l appraisal of the developed methodology framework and		
		associ	ated research opportunities	178	
		8.4.1	Health damage cost assessment	178	
		8.4.2	Private cost assessment	183	
		8.4.3	Social CBA of emission control measures	184	
	8.5	Using	the social CBA methodology in the business context:		
		metho	odological recommendations	185	
	8.6	Summ	ary of chapter 8	191	
9	Sumi	mary a	nd outlook	193	
Ар	Appendix199				
Re	References				

List of Figures

Figure 1.1:	General approach and underlying methodological developments of the current thesis
Figure 2.1:	Supply and consumption of non-renewable energy resources; reprinted with permission from and based on BGR (2016)
Figure 2.2:	Levelized cost of electricity of different electricity generation technologies as a function of full load hours per year; reprinted with permission from and based on Schröder et al. (2013)
Figure 2.3:	Gross EU28 electricity generation by energy carrier; adapted from eurostat (2017), using information from European Environment Agency (2015) for data aggregation purposes
Figure 2.4:	Annual EU28 greenhouse gas emissions per sector; adapted from eurostat (2016a)
Figure 2.5:	Annual EU28 NO _x emissions derived from Nomenclature For Reporting (NFR) 2014 sector classification; based on European Environment Agency (2016)17
Figure 2.6:	Annual EU28 SO _x emissions derived from NFR 2014 sector classification; based on European Environment Agency (2016)
Figure 2.7:	Annual EU28 PM _{2.5} emissions derived from NFR 2014 sector classification; based on European Environment Agency (2016)
Figure 3.1:	Economic methods for assessing emission control measures; derived from European Commission (2006a)27
Figure 3.2:	Illustration of a negative production externality; left: quantity (q) and price (p) under a competitive equilibrium (E); right: new equilibrium (E*) when internalising a negative externality, also showing associated welfare losses (w) if internalisation did not take place; derived from Pigou (1932)
Figure 3.3:	Illustration of the optimal level of pollution (O), associated marginal costs (c), marginal benefits (b) and level of emission abatement (a); derived from Samuelson and Nordhaus (2007)

Figure 3.4:	Impact pathway approach for assessing health-related damage costs following the DPSIR scheme (oval shapes = input data; rectangles = output data; rounded rectangles = assessment steps); derived from European Commission (2005a)41
Figure 4.1:	Marginal damage costs of atmospheric emissions from a coal-fired power plant unit, estimated with different IPA implementations72
Figure 4.2:	Influence of exposure modelling, particle toxicity and long-term mortality assessment on health-related damage costs of atmospheric emissions from a coal-fired power plant unit73
Figure 5.1:	Damage costs of a coal-fired power plant located in Brussels (Belgium), Cartagena (Spain), or Helsinki (Finland) without (wo) and with (w) DeNOx, calculated by means of EcoSenseWeb v1.3 (default assessment)
Figure 6.1:	Schematic overview of input data, assessment steps and output of the health damage cost assessment framework; own conception based on the impact pathway approach
Figure 6.2:	Daily SO ₂ emissions per unit in scenario S2, reflecting the operation profile of the oil-fired power plant throughout the year; own compilation
Figure 6.3:	Modelling domain "France", displaying the "canton" level, with enlarged section of "Île-de-France" modelling domain, displaying "communes" per (colour-differentiated) NUTS3 zone; emission source located at white dot; maps based on IGN (2013)
Figure 6.4:	Modelling domain "Europe", displaying NUTS3 administrative zones per (colour-differentiated) country; map based on eurostat (2015a), © EuroGeographics for the administrative boundaries
Figure 6.5:	Atmospheric processes considered in Eulerian chemistry-transport models; reprinted with permission from and based on Sportisse (2007)
Figure 6.6:	Population number per commune in France in 2009 and emission source location (white dot); based on IGN (2013) 104
Figure 6.7:	Data processing during exposure assessment, featuring the zonal statistics function; exemplary scheme for the European domain (own compilation)

Figure 6.8:	Cost curve (quasi-marginal abatement costs) for installing a low NO _x burner at an oil-fired power plant unit; dotted line at 500 h/a; based on own calculations
Figure 6.9:	Cost curve (quasi-marginal abatement costs) for installing an electrostatic precipitator at an oil-fired power plant unit; dotted line at 500 h/a; based on own calculations
Figure 7.1:	Annual health damage costs related to the normal emission intensity scenario (S2b) and the high emission intensity scenario (S2c) at different modelling domains
Figure 7.2:	Annual health damage costs related to emission scenario S2c using the default background emission inventory (left) and a hypothetical future background emission inventory 2020 (right) at the France modelling domain
Figure 7.3:	Annual health damage costs related to the variable emission scenario (S2) and the equivalent constant emission scenario (S2b) at different modelling domains
Figure 7.4:	Change in annual ambient mean PM _{2.5} concentration due to the variable operation scenario S2 (left) and the equivalent constant emission scenario S2b (right) at the France modelling domain
Figure 7.5:	Wind directions and wind speed simulated using the WRF model at emission release height over the France modelling domain during the variable operation scenario (left) and during the constant operation scenario (right); Source: Legorgeu (2016)
Figure 7.6:	Annual health damage costs related to emission scenario S2 at different modelling domains and assessed using different spatial modelling resolutions
Figure 7.7:	Health damage costs caused by emission scenarios S1 and S2 using the standard or the plume-in-grid (PIG) modelling approach at the France and Île-de-France modelling domain
Figure 7.8:	Change in annual ambient mean PM _{2.5} concentration due to operation scenario S2 at the Île-de-France modelling domain using a medium (left) or high (right) spatial modelling resolution
Figure 7.9:	Change in annual ambient mean PM _{2.5} concentration due to the variable operation scenario S2 using the standard (left) or plume- in-grid (PIG) modelling approach (right) at the Île-de-France modelling domain

Figure 7.10:	Annual health damage costs related to emission scenarios S2 (variable) and S2b (constant) showing the percentage contribution of NO ₂ -related mortality impacts due to long-term exposure (annual average above 20 μ g/m ³) at different modelling domains 135
Figure 7.11:	Change in annual ambient mean NO ₂ concentration (above 20 μ g/m ³) due to operation scenario S2b at the Île-de-France (left) and France (right) modelling domain
Figure 7.12:	Change in annual ambient mean NO ₂ concentration (above 20 μ g/m ³) due to operation scenario S2b at the European modelling domain (excerpt of modelling domain)
Figure 7.13:	Annual mortality-related health damage costs due to long-term PM _{2.5} exposure of the emission scenarios S2 (variable) and S2b (constant) at different modelling domains and estimated using two alternative approaches
Figure 7.14:	Annual ozone-related health damage costs for emission scenarios S2 (variable), S2b (constant, normal emission intensity) and S2c (constant, high emission intensity) at different modelling domains and using two alternative concentration thresholds
Figure 7.15:	Annual health damage costs related to the initial emission level (S1) and after installing primary emission control measures for NO _x and SO ₂ (S2) at different modelling domains
Figure 7.16:	Annualised costs of primary emission control measures and related benefits through avoided health damage costs per power plant unit and for the European modelling domain
Figure 7.17:	Annual health damage costs related to the initial emission level with primary NO _x and SO ₂ control measures installed (S2) and after installing an additional secondary emission control measure for PM (S3) at different modelling domains
Figure 7.18:	Annualised costs of secondary emission control and related benefits through avoided health damage costs per power plant unit and for the European modelling domain
Figure 7.19:	Annualised costs of primary emission control measures and related benefits through avoided health damage costs per power plant unit and for the European modelling domain using different approaches for benefit assessment

Figure 7.20:	Annualised costs of secondary PM emission control and related benefits through avoided health damage costs per power plant unit and for the European modelling domain using different approaches for benefit assessment	49
Figure 7.21:	The capital recovery factor of the annuity method as a function of project lifetime and weighted average cost of capital for typical ranges of values	51
Figure 7.22:	Annualised costs of primary emission control measures and related benefits through avoided health damage costs per power plant unit and for the European modelling domain using different assumptions and parameters for the cost and benefit assessment 15	52
Figure 8.1:	Health damage costs per tonne of substance emitted in 4 German sub regions (DE1 – DE4) and per height of release; adapted from van der Kamp et al. (forthcoming)	66
Figure 8.2:	Flowchart representing the choice of the basic methodological approach to be used for health damage cost assessment at the level of an industrial point emission source (own conception)	90

List of Tables

Table 2.1:	European (EU), German (DE) and French (FR) energy policy targets in 2020, 2030 and 2050 (Assemblée Nationale 2015, BMBU 2016, BMWi 2014, Bundesregierung 2010, European Commission 2008a, 2011, 2014b, 2015b, 2016, European Parliament and Council of the European Union 2009)
Table 2.2:	Potential emissions related to fuel cycle stages enabling electricity generation in large combustion plants; adapted from European Commission (2006b)
Table 2.3:	Main quantifiable environment and health impacts caused by classical air pollutant emissions (European Commission 2005a, Preiss and Klotz 2008, WHO 2013a)21
Table 2.4:	Regulations targeting classical air pollutants and relevant to fossil fuel-fired power generation, differentiated by category and scope (own compilation)
Table 3.1:	Comparison of the main features of health impact assessment and health risk assessment applied to emission sources; based on CDC (2015)
Table 3.2:	Uncertainty classification including the nature/type of uncertainty and associated sources; adapted from (Institute of Medicine 2013, IPCC 1996, Swart et al. 2009)45
Table 3.3:	Overview on exemplary studies assessing emission control measures of classical air pollutants in the energy sector, including information on their scope and models used (own compilation)
Table 3.4:	Comparative overview on selected assessment tools to quantify human health-related damage costs from point emission sources (own compilation)
Table 3.5:	Proposal of elements to consider when transferring a social CBA methodology for the assessment of emission control measures from the public policy to the business domain (own compilation)
Table 4.1:	Air quality and exposure modelling characteristics for classical air pollutants except ozone for the considered IPA implementations (European Commission 1999b, 2004b, Preiss and Klotz 2008a)

Table 4.2:	Health-related impact function slopes (s_{CR}) and corresponding risk/age group, expressed as 'effect increase per (10 µg/m ³ x person x year)' and grouped by pollutant (European Commission 2004, Torfs et al. 2007, WHO 2013a)	66
Table 4.3:	Monetary values (m) expressed in €2000 for health endpoints (European Commission 1999a, 2004, Holland 2014a, Preiss and Klotz 2008)	70
Table 5.1:	Settings of EcoSenseWeb as used in the analysis of the current chapter; based on Preiss and Klotz (2008)	79
Table 5.2:	Data used as input into EcoSenseWeb to define the damage costs of a coal-fired power plant unit without and with a selective catalytic reduction (SCR) and primary abatement NO _x measures installed: adapted from European Commission (2004).	80
Table 5.3:	Geographic coordinates of the three investigated power plant locations	82
Table 5.4:	Private costs of installing a SCR and primary abatement measures at a coal-fired power station, according to EGTEI (2012)	84
Table 5.5:	Annual damage costs without (wo) and with (w) DeNOx and resulting annual benefits (default assessment)	85
Table 5.6:	Discounted benefits, discounted costs and net present values of the CBA of a SCR installation at a coal-fired power plant at three different locations (default assessment)	86
Table 5.7:	Discounted benefits, discounted costs and net present values for 2500 varying full load hours per year (otherwise default settings)	87
Table 5.8:	Discounted benefits, discounted costs and net present values as a function of default and sensitivity assumptions regarding particle toxicity and air quality modelling	88
Table 6.1:	Data input and processing steps used in the health damage cost assessment (own compilation)	95
Table 6.2:	Power plant operation scenarios and related emission levels, valid for units 1-4 of the power plant respectively; parameter changes from one scenario to the next in bold; own assumptions	97
Table 6.3:	Elements to be considered in the scope definition of atmospheric modelling and specific choices made in the current thesis (grey shaded); own compilation	98

Table 6.4:	Characteristics and data sources of the atmospheric modelling; based on Legorgeu (2016)	101
Table 6.5:	Data used for exposure assessment (own compilation)	103
Table 6.6:	Data used for health impact assessment at the three assessment domains (own compilation)	106
Table 6.7:	Health endpoints and data used in the assessment framework developed in this thesis; NO ₂ -related endpoints (grey) for sensitivity analysis; based on WHO (2013a)	108
Table 6.8:	Default monetary valuation factors (net of income growth) and included components used in the assessment framework developed in this thesis (own compilation)	114
Table 6.9:	General parameters used for cost characterisation of emission control measures (own assumptions)	117
Table 7.1:	Overview on research objectives and corresponding case studies treated in the current chapter	123
Table 7.2:	Objective and research approach to assess the influence of emission intensity at source level on health damage costs	125
Table 7.3:	Objective and research approach to assess the influence of the background emission inventory on health damage costs	126
Table 7.4:	Objective and research approach to assess the influence of temporal resolution on health damage costs	128
Table 7.5:	Objectives and research approach to assess the influence of spatial resolution on health damage costs	130
Table 7.6:	Objective and research approach to assess the influence of additionally quantifying direct NO2-related health effects	134
Table 7.7:	Objective and research approach to assess the influence of different approaches for long-term exposure mortality impact assessment	137
Table 7.8:	Objective and research approach to assess the influence of different concentration threshold levels for ozone-related health effects	139
Table 7.9:	Objectives and research approach to assess the private costs and health-related benefits of primary and secondary emission control measures	140

Table 7.10:	Objectives and research approach for the social CBA of primary emission control measures using simplified methods for health benefit assessment	146
Table 7.11:	Objectives and research approach for the sensitivity analysis of input parameters used in the social CBA of primary emission control measures	150
Table 8.1:	Overview on influencing factors, underlying types of uncertainty and sensitivity of health damage costs with regard to changing the influencing factors; based on case studies within this thesis or related work	167
Table 8.2:	Review of unquantified elements with an impact on the social CBA of emission control measures	173
Table A.1:	Overview on selected environmental pressures and related impacts caused by fossil fuel-fired power generation; adapted from Bachmann and van der Kamp (2014)	199
Table A.2:	Data input requirements in EcoSenseWeb for a fossil-fired power plant; based on Preiss and Klotz (2008)	201
Table A.3:	Age group fractions and corresponding years used in the health impact assessments of this thesis; sources: Île-de-France and France: (INSEE 2009c); EU-average: (WHO 2016)	203
Table A.4:	EU and French baseline rates for 2009 per health endpoint used in the health impact assessments of this thesis	203
Table A.5:	Life expectancy in EU28 member states in the year 2009 and YOLL (population aged 30+) per 1-year increase of $10 \ \mu g/m^3 \ PM_{2.5}$ concentration (life expectancy data from HFA DB (WHO 2016); YOLL estimates based on Miller et al. (2011))	205

Abbreviations

AAQD	Ambient Air Quality Directive (Directive 2008/50/EC)
BAT	Best Available Techniques
BREF	BAT REFerence (document)
CAFE	Clean Air For Europe
CAPEX	Capital Expenditures
CBA	Cost-Benefit Analysis
CCS	Carbon Capture and Storage
CH4	Methane
CLRTAP	Convention on Long-Range Transboundary Air Pollution
CO2	Carbon dioxide
CRF	Concentration-Response Function
CVM	Contingent Valuation Method
DeNOx	De-Nitrification (system for reducing NOx emissions)
DPSIR	Driver-Pressure-State-Impact-Response
EAP	Environmental Action Programme
EEA	European Environment Agency
EGTEI	Expert Group on Techno-Economic Issues
EIAD	Environmental Impact Assessment Directive (Directive 2014/52/EU)
ELV	Emission Limit Value
EMEP	European Monitoring and Evaluation Programme

ESP	Electrostatic Precipitator
ETS	Emission Trading Scheme
EU	European Union
ExternE	Externalities of Energy (EU research project series)
FR	France
GAINS	Greenhouse Gas and Air Pollution Interactions and Synergies
GDP	Gross Domestic Product
GHG	Greenhouse Gas(es)
GIS	Geographic Information System
HEIMTSA	Health and Environment Integrated Methodology and Toolbox for Scenario Assessment (EU research project)
HRAPIE	Health Risks of Air Pollution In Europe
IDF	Île-de-France
IED	Industrial Emissions Directive (Directive 2010/75/EU)
IPA	Impact Pathway Approach
IPPC	Integrated Pollution Prevention and Control (Directive 2001/80/EC)
LAU	Local Administrative Units
LCOE	Levelized Cost Of Energy
LNB	Low NOx Burner
MAC	Marginal Abatement Cost (approach)
MCPD	Medium Combustion Plants Directive (Directive EU/2015/2193)
MDC	Marginal Damage Cost (approach)
MRAD	Minor Restricted Activity Days

MW(h)	Megawatt (hour)
NECD	National Emission Ceilings Directive (Directive 2016/2284/EU)
NEEDS	New Energy Externalities Developments for Sustainability (EU research project)
NewExt	New Elements for the Assessment of External Costs from Energy Technologies (EU research project)
NFR	Nomenclature For Reporting
NH3	Ammonia
NMVOC	Non-Methane Volatile Organic Compounds
NO _x /NO ₂	Nitrogen oxide/Nitrogen dioxide
NUTS	Nomenclature of Territorial Units for Statistics
OPEX	Operational Expenditures
PIG	Plume-in-Grid
PM	Particulate Matter
PM _{2.5} /10	Particulate Matter with an aerodynamic diameter below 2.5/10 μm
RAD	Restricted Activity Days
SCR	Selective Catalytic Reduction
SIA	Secondary Inorganic Aerosols
SIREAM	SIze REsolved Aerosol Mode
SO2	Sulphur dioxide
SOMO35	Sum Of Means Over (35 parts per billion)
TFTEI	Task Force on Techno-Economic Issues
TSAP	(EU) Thematic Strategy on Air Pollution

Abbreviations

UNECE	United Nations Economic Commission for Europe
VOLY	Value Of a Life Year
VSL	Value of a Statistical Life
WACC	Weighted Average Cost of Capital
WHO	World Health Organization
WLD	Work Loss Days
WRF	Weather Research and Forecasting (model)
YOLL	Years Of Life Lost

Preface

Publishing this thesis has taken several years and involved lessons on scientific rigour, collaborative work, intrinsic motivation etc. Having finished this process (but not the learning), I would like to thank several people and institutions for their valuable support:

My supervisor Prof. Ute Karl shaped my early professional development. Her experience helped me to distinguish the academic from the business world, which is particularly useful when transferring academic methods to a business context. Many thanks also to my co-supervisor Prof. Frank Schultmann (and the Institute for Industrial Production (IIP) at the KIT) for the assistance and for giving me the opportunity to learn from and exchange with academic peers. With Till Bachmann I share fruitful working years - through his profound scientific background he provided valuable support to this thesis and beyond. Robin Lorenz did a great job in supporting me with programming tasks. Thanks as well to my collaboration partners at CEREA for their contribution and, of course, to EIFER, including my colleagues, for the productive atmosphere that enabled me to pursue this thesis. Last but not least, I would like to mention my family: apart from gently pushing for progress from time to time, what matters most is their mere existence.

I gratefully acknowledge support from the Karlsruhe House of Young Scientists which enabled me to broaden my horizon during a research period in Singapore. The Stiftung Landesbank Baden-Württemberg generously supported the publication of this thesis.

Finally, I would like to express some motives underlying this thesis. The fact of being a business engineer partly explains the contents (i.e. a bit of everything, aka a transversal and multidisciplinary work). The topic is founded in my conviction that our environment should be preserved. Accordingly, I oriented my academic career towards energy (a life supporting good but also cause of environmental harm) and environmental economics. There are certainly more obvious and effective ways of protecting the environment than writing a thesis that aims at establishing environmental economic methods in an industrial context. Yet, industry should be part of the solution. Advancing scientific methods that aim at spending scarce resources efficiently, whilst at the same time integrating the environment into decision-making is thus my contribution. Though acknowledging limitations related to these methods, I consider that applying and discussing them in a sense of continuous improvement is globally more helpful than rejecting them.

Karlsruhe, June 2017
1 Introduction

1.1 Context and problem definition

A welfare optimizing allocation of resources in markets is often inhibited by the existence of so-called externalities or, when expressed in monetary terms, external costs. External costs comprise those positive or negative effects related to producing a good or delivering a service which are not reflected in its market price. A typical example of externalities and a key element henceforth are unpriced human health impacts caused by atmospheric emissions that originate from fossil fuel combustion, e.g. in the public power or transport sector. Another prominent example related to combustion processes, though not at the centre of this thesis, are anthropogenic greenhouse gas emissions that contribute to global warming (IPCC 2014b).

Policy-making concerned with reducing market distortions and increasing welfare aims at internalising external costs. This involves their integration into market prices through incentive-based instruments or quality standards. In doing so and with regard to environmental policy-making, the European Union (EU) applies the "polluter pays principle", requiring that polluters take financial responsibility for the damage caused by their activities (article 191 (2), consolidated version of the Treaty on the Functioning of the European Union, 2012). Moreover, the EU demands that "available scientific and technical data" as well as "the potential benefits and costs of action or lack of action" are accounted for in environmental policy-making (article 191 (3), ibid.). This points to the need to balance environmental with wider economic indicators, e.g. by simultaneously considering the private costs and societal benefits (i.e. avoided damages) related to air pollution control measures. To fulfil this task, reliable science-based methods are required, enabling policymakers to

1) assess air pollution-related external costs or benefits, and

2) put external costs into perspective with private effects.

These methods are the centrepiece of this thesis. Their respective state of scientific advancement shall be briefly summarised in order to motivate the methodological developments carried out in the following. Despite early pioneering work by the economist Pigou (1912) and first environmentaleconomic approaches related to energy emerging in the 1980s (Hohmeyer 1988, Mendelsohn 1980), it was only in the 1990s that the first comprehensive methods for the monetary assessment of externalities in the energy sector were developed in a joint US-European research initiative (European Commission 1995, Oak Ridge National Laboratory and Resources for the Future 1992). A great deal of work was devoted to human health impacts caused by atmospheric emissions that proved to represent a substantial share of quantifiable damages and that are still a cause of concern. Since then, corresponding assessment methods have evolved continuously, reflecting updated scientific evidence and improved modelling capabilities (cf. chapter 4).

Yet, when aiming to assess health impacts caused by individual fossil fuel power plants, more variable operation patterns, induced by recent market developments (cf. section 2.3), require the corresponding assessment methods to be adapted. Most existing European-wide assessment models fail to properly account for input data variability, since they are based on a roughly resolved and averaged modelling in terms of space and time (cf. sections 3.5 and 5.5.2). This neither allows assessing spatial variations in population exposure properly, nor does it account for temporal variations in emissions and meteorological background conditions during the year. Using a highly-resolved space- and time specific atmospheric modelling is thus identified as a way of improving the health impact assessment of industrial point emission sources (cf. chapter 6).

Social cost-benefit analysis (CBA) is frequently applied to compare private and external effects, particularly in the public policy context. In the energy sector and with regard to air quality, social CBA has been regularly used as a support to define appropriate levels of policy intervention at an aggregated level, e.g. the country or European level. The overall efficiency of policy scenarios is assessed, essentially by balancing private emission control costs and societal health (or other) benefits from reduced pollution levels. Whilst acknowledging some limitations of CBA (cf. section 8.4), its use in the regulatory context is nowadays well accepted. Applications of social CBA at the level of a particular point emission source are rare (cf. section 2.4.6) and more generally as a means to foresee and cope with increasing environmental requirements, the use of CBA presents opportunities for the private sector. The social CBA methodology therefore needs to be transferred from the public policy domain to the business domain, which is another key undertaking of the current thesis.

As a first step towards this objective, an exploratory social CBA of emission control at a coal-fired power plant is conducted in chapter 5, focusing on variability in damage costs

due to geographical settings. However, the employed model uses partly outdated input data and, most critically, does not account for increasingly variable operation profiles. Other existing models rely likewise on averaged damage cost factors. This lack of a social CBA methodology applicable to industrial point emission sources is tackled by the dedicated methodological advancements of this thesis (cf. chapter 6). Even though these developments focus on the energy sector, the general approach to social CBA can be easily transferred to other industrial sectors concerned with emission control measures.

Through practical case studies within this thesis, new insights are gained on the influence of modelling features and methodological choices on health damage costs. These insights feed into a thorough discussion of uncertainty underlying social CBA and health damage costs in particular. To conclude the methodological developments and based on the insights generated within this thesis, specific recommendations for the application of social CBA in the business context are derived (cf. section 8.5).

1.2 Goal definition

The overarching goals and corresponding sub-objectives of this work are to:

- Extend the existing scientific methods for the assessment of human health damage costs caused by atmospheric emissions from fossil fuel power plants in order to cope with temporal variability in operation profiles and spatial variability in population densities. This shall be achieved by developing a new assessment framework, building upon a highly resolved atmospheric modelling.
- Transfer the methodology for social cost-benefit analysis of emission control measures from the public policy to the private business context. This shall be achieved by
 - combining the health damage cost approach with a private cost characterisation approach in order to carry out social CBAs of emission control measures at site level;
 - transparently evaluating the underlying approaches, their strengths and limitations.

- Derive methodological recommendations accounting for uncertainty in health damage costs. This shall be achieved by
 - varying the most influential parameters regarding modelling features and methodological choices in order to analyse the sensitivity of results with respect to key parameters;
 - discussing uncertainty on a quantitative and qualitative basis and describing the consequences of uncertainty for public and private decision-making.

1.3 Approach and related contents

Figure 1.1 summarises the global approach followed in this thesis as well as related contents.





Following the above depicted approach, this thesis includes the following contents, each chapter ending respectively with a short summary:

In chapter 2, the current economic and environmental challenges of fossil fuel-fired power generation in Europe are described. This notably includes environmental and health impacts caused by atmospheric emissions and the associated regulatory framework, providing the rationale for economic assessment of emission control measures.

In chapter 3 the key methods underlying social CBA are presented, split into techno-economic and environmental-economic approaches. After introducing the theory, a comprehensive literature overview serves to categorise existing scientific work dealing with the economic assessments of emission reduction measures at fossil fuel power plants. Crucially, this overview points out the limitations of currently available models with regard to variable power plant operation and therefore motivates the development of a new assessment framework within this thesis.

In chapter 4, an in-depth methodology analysis and review of implementations of the damage cost approach is carried out. The aim is twofold. On the one hand, the methodological evolution over the past 20 years is transparently described and illustrated, based on existing and newly derived damage cost estimates. On the other hand, quantitative analyses are carried out regarding the influence of key parameters and assumptions on health damage costs, providing information for sensitivity analysis.

Chapter 5 features an application of social CBA at an exemplary power plant using an existing assessment model. The analysis demonstrates the influence of the geographic emission source location on quantified damages amongst other factors. An in-depth gap analysis of the assessment model used reveals its limitations, e.g. the use of time- and space averaged atmospheric modelling results.

A new methodological framework for health damage cost assessment at site level is developed and implemented in order to tackle the key limitation of existing models, as presented throughout chapter 6. This framework is completed by developing the second key element needed for social CBA, i.e. a costing methodology for selected primary and secondary emission control measures.

In chapter 7, the newly developed damage cost assessment framework and the costing methodology are applied in several case studies that are classified into three categories, each serving to analyse specific aspects. The influence of emission patterns and atmospheric modelling features on health damage costs is analysed first, permitting a quantitative comparison of a highly resolved modelling to time- and space-averaged modelling.

This is followed by an analysis of methodological assumptions concerning health impact assessment. Finally, social CBAs of emission control measures at site level are carried out, discerning primary and secondary abatement measures. Sensitivity analyses test the influence of key parameter and methodological assumptions on the results obtained, providing new insights on uncertainty and the overall robustness of the developed framework.

Based on the results of the previous chapters, chapter 8 starts with a discussion of the case study results. Moreover, it includes a comprehensive analysis of uncertainty underlying health damage costs and a discussion of how uncertainty influences decision-making in the private and policy context. This is followed by a critical appraisal of the developed methodology, serving at the same time to identify future research opportunities. Methodological recommendations regarding the application of the social CBA method in the business context conclude the chapter.

In chapter 9 the contents of this thesis are summarised, followed by an outlook.

2 Background: fossil fuel-fired power generation under changing economic and environmental conditions

This chapter describes the larger context of fossil fuel-fired power generation in Europe, setting the scene for the scientific undertakings in this work. The overarching European climate and energy policy targets are presented, as they are important drivers for the future development of the energy system including the future of fossil fuel-based power generation (cf. section 2.1).

Traditional strengths but also weaknesses regarding fossil fuel-based power generation are discussed with special regard to the often-cited triangle of objectives: security of supply, competitiveness, and environmental soundness (cf. sections 2.2 to 2.4). Particular attention is given to the environmental impacts caused by fossil fuel power plants, as they are a key element in the methodological developments of this thesis.

Due to the transboundary nature of atmospheric substance dispersion and the predominant influence of European energy and environment policy, the current chapter takes a European perspective in the sense that most background information, statistics, and results will be presented at the aggregated European Union (EU) level. This is partly complemented by national data for France and Germany.

2.1 The European climate and energy policy framework

The short- and long-term development of energy markets is strongly defined by overarching political targets, especially with regard to greenhouse gas (GHG) emission reductions, the share of renewable energy, and energy efficiency objectives (Table 2.1). As indicated, the degree of commitment varies between these targets and between years, leading to more or less emphasis with which these targets are pursued in practice. Table 2.1:European (EU), German (DE) and French (FR) energy policy targets in 2020, 2030 and 2050
(Assemblée Nationale 2015, BMBU 2016, BMWi 2014, Bundesregierung 2010, European
Commission 2008a, 2011, 2014b, 2015b, 2016, European Parliament and Council of the European
Union 2009)

		2020	2030	2050		
GHG emission reduction (towards the base year 1990)		EU: 20 % ^b	EU: 40 % ^{b*}	EU: 80 - 95 % ^{n-b}		
		DE: 40 % ^b	DE: 55 % ^{nt}	DE: 80 - 95 % ^{nt}		
		FR: 31 % ^b	31 % ^b FR: 40 % ^{nt}			
Renewable energy share in gross final energy consumption		EU: 20 % ^b	EU: 20 % ^b EU: 27 % ^{b*}			
		DE: 18 % ^b	DE: 30 % ^{nt}	DE: 60 % ^{nt}		
		FR: 23 % ^b	FR: 32 % ^{nt}	FR: not available		
Energy efficiency	Primary energy consumption	EU: 20 % ^{n-b*} (base year 1990)	EU: 30 % ^{b*} (base year 2007)	EU: not available		
	reduction	DE: 20 % ^{n-b*}	DE: not available	DE: 50 % ^{nt}		
		(base year 2008)		(base year 2008)		
	Final energy consumption reduction	FR: 17.4 % ^{n-b*} (towards a 2020 baseline scenario)	FR: 20 % ^{nt} (base year 2012)	FR: 50 % ^{nt} (base year 2012)		

^b = binding target; ${}^{b^*}$ = proposed binding target; ${}^{n\cdot b}$ = non-binding target; ${}^{n\cdot b^*}$ = non-binding overall target but with binding obligations at national level; nt = national target

For the year 2020, binding GHG and renewable energy targets were first set at the EU level and then broken down into binding national commitments. The European Commission can therefore start infringement procedures in the case of non-respect of these targets. For the year 2030, the European Commission has decided to set an average minimum European greenhouse gas reduction target without prescribing concrete and binding national targets. Likewise a minimum share of renewable energy consumption at European level is defined without a national breakdown (European Commission 2014b). Recently proposed legislation would also imply a binding energy efficiency target at EU level for 2030 (European Commission 2016). Although France and Germany (like other member states) have defined their own national targets, for the time being these are neither binding towards the European Commission nor binding at national level as they may be adapted by future governments. In the long run, i.e. up to 2050, certain targets are already specified, without however having a binding status so far. While the targets for greenhouse gas reduction and renewable energy shares are clearly defined, the case is more complex for energy efficiency, where targets are either expressed in absolute or

relative terms and with regard to primary or final energy consumption. In addition, national targets depend on further factors, e.g. whether certain sectors such as air transport are included or whether a temperature normalisation of consumption data is carried out. This makes cross country comparisons difficult.

The influence of these policy targets (and further influencing factors) on the future perspectives of fossil fuel-fired power generation are discussed next.

2.2 Historic strengths of fossil fuel-fired power generation

Fossil fuel-fired power generation, largely based on burning coal (including lignite), oil or natural gas in large combustion plants, has played an important role in the European energy mix for many decades. In the past, its main advantages over other forms of electricity generation were availability, henceforth termed security of supply, and competitiveness.

2.2.1 Security of supply

"Security of supply", i.e. the availability of electric power in sufficient quantity and quality from traditionally large centralised power plants, is a main advantage of fossil fuel-fired power generation compared to other generation technologies. Fossil fuel-fired power plants are "dispatchable", meaning their operation profile can be adjusted to match the demand side, both in terms of electricity supply and auxiliary services, e.g. frequency regulation (Joskow 2011). Their degree of flexibility in terms of minimum load, minimum operating time or ramp up -and down rates is subject to technical constraints and differs between power plant types (Brauner et al. 2012).

Not only in the short-term but also in the long run, fossil fuel-fired power generation has the capacity to secure the supply of electricity, due to substantial worldwide fuel reserves. While emphasizing the uncertainties inherent in any kind of future projection, a comparison of global reserves (i.e. "proven volumes of energy resources economically exploitable at today's prices and using today's technology") and resources (i.e. "proven amounts of energy resources which cannot currently be exploited for technical and/or economic reasons, as well as unproven but geologically possible energy resources which may be exploitable in future") with projected consumption data up to 2040 (Figure 2.1) indicates that fossil energy carriers remain available in abundant quantity for several decades or even centuries to come (BGR 2015, 2016).



Figure 2.1: Supply and consumption of non-renewable energy resources; reprinted with permission from and based on BGR (2016)

2.2.2 Competitiveness

As competitiveness is closely linked to generation costs, the levelized cost of energy (LCOE), i.e. the discounted lifetime fixed and variable costs per unit of electricity produced, are considered in the first instance. Estimates on the LCOE for electricity generation technologies vary widely (DECC 2013, European Commission 2008b, IPCC 2014a, Schröder et al. 2013, U.S. Energy Information Administration 2015), mainly depending on assumptions regarding:

- investments, fixed and variable operating costs;
- financing conditions, described by the weighted average cost of capital (WACC);
- power plant lifetime, conversion efficiency, and utilization rate (full load hours);
- CO2 emissions and related costs;
- discount rate.

As of today, fossil-based electricity generation still qualifies as a relatively cheap option compared to nuclear or renewable energy-based generation, albeit with a strong dependency on full load hours (Figure 2.2) and also on assumptions regarding future technology and cost developments ("learning curves").



Figure 2.2: Levelized cost of electricity of different electricity generation technologies as a function of full load hours per year; reprinted with permission from and based on Schröder et al. (2013)

At the same time, using the LCOE is subject to criticism, especially when comparing intermittent renewable technologies with dispatchable conventional technologies. This is mainly because the LCOE neither reflects attributes such as security of supply (availability and flexibility), nor system integration costs or the market value of the electricity generated (Edenhofer et al. 2013, IEA 2014, Joskow 2011, Ueckerdt et al. 2013). When accounting for these factors, the relative advantage of fossil fuel-fired generation compared to intermittent renewable energies tends to increase. On the contrary, when considering environmental impacts, most of which are not reflected in the LCOE, the competitiveness of fossil fuel-fired generation is expected to decrease, as discussed in the next section.

Subsidies on extraction and use of fossil fuels

Although no immediate strength but rather representing another type of market distortion, public subsidies for fossil fuel extraction (e.g. for hard-coal mining in Germany) or consumption (e.g. tax reliefs for specific types of fossil fuel used in certain industries) contribute to the competitiveness of fossil fuel-fired power generation in many countries worldwide (OECD 2013).

2.3 Supply and demand, structural changes and a tendency of increasing flexibility

One cause for a deteriorating business context of fossil fuel-fired power generation in past years is a tendency of falling electricity demand, particularly in the industry sector. The dependency of electricity demand on macro-economic factors can be observed when looking at the economic crisis of the years 2008 and 2009, initially sparked by turbulences in financial markets. In a number of EU countries, the crisis led to a slowdown of industrial activity and hence a tendency of decreasing electricity generation (Figure 2.3).



Figure 2.3: Gross EU28 electricity generation by energy carrier; adapted from eurostat (2017), using information from European Environment Agency (2015) for data aggregation purposes

Although electrification within certain sectors, especially transport and heating, could reverse this trend to some degree, more and more stringent energy efficiency measures

across all sectors (Table 2.1) are expected to sustain a downward pressure on electricity demand in the future.

Another cause for the difficulties of conventional power generation, particularly evident in Germany, are structural changes within the energy sector itself, such as the fast rising share of renewable energy sources that is set to increase further (Table 2.1). Through priority market access and fixed feed-in tariffs, they have contributed to decreasing wholesale electricity market prices ("merit-order effect"), while at the same time requiring more flexibility in the system (European Commission 2015a, Sensfuß et al. 2008).

The implications of increasing flexibility, induced by variable renewable energies, have been analysed in various studies, for instance by modelling prospective energy scenarios for the German market (Brauner et al. 2012, Nitsch et al. 2012). For fossil fuel-fired power generation, these foresee a reduction in annual full load hours¹ as well as an increasing tendency of part-load operation and more frequent load shifts, generally referred to as cycling. This in turn reduces the overall energy conversion efficiency with the consequence of increasing generation costs and with different effects on atmospheric emissions (Lew et al. 2013), further discussed in section 8.4. Moreover, increased cycling bears the risk of material wear or damage, further deteriorating the power plant's competitiveness (EPRI 2001, Kumar et al. 2012).

Notwithstanding these tendencies, as long as alternative flexibility mechanisms such as storage or demand-side solutions remain under development, an important share of electric backup capacity will continue to be provided by fossil fuel power plants. Fossil fuel-fired power generation, responsible for 42.7% of gross EU electricity generation in 2015 (Figure 2.3), thus faces a dilemma. On the one hand, it is expected to remain a stabilising element in the European power generation mix to ensure security of supply. On the other hand, its profitability is under increasing pressure by the latest market developments. Capacity markets, i.e. markets that remunerate the availability of production capacity as a supplement to electricity generated, are considered as a possible solution to this dilemma and are currently being introduced in some countries, e.g. France, while other countries like Germany stick to the so-called energy only market, i.e. without dedicated capacity markets. Yet, Germany foresees alternative measures to secure supply, e.g. a strategic reserve of fossil fuel power plants (European Commission 2015a).

¹ i.e. operating hours at full nominal power plant capacity (or equivalents hereof)

2.4 Atmospheric emissions and environmental impacts

A less obvious, though increasingly important influencing factor on the competitiveness of fossil fuel power plants is their environmental performance and related regulatory constraints. Emissions into air, water or soil arise at different stages of the fuel cycle of large combustion plants, as illustrated in Table 2.2. These substances, either directly or after reacting with other chemical substances, are potential sources for environmental and human health impacts (cf. section 2.4.4).

Source release:		Substances											
Air (A) Water (W) Land (L)	Particulate matter (PM)	Oxides of sulphur (SO _x)	Oxides of nitrogen (NO _x)	Oxides of carbon	Organic compounds	Acids/alkalis/salts	Hydrogen chloride/fluoride	Volatile organic compounds	Metals and their salts	Chlorine (as hypochlorite)	Mercury and/or cadmium	PAHS	Dioxins
Fuel storage and handling	А				W			А					
Water treatment	W								W		W		
Exhaust gas	А	А	А	А	А		А	А	А		А	А	А
Exhaust gas treatment	W				W				WL		W		
Site drainage incl. rainwater	W				W								
Waste water treatment	W				W	W							
Cooling water blowdown	W				W				W	w	W		
Cooling tower exhaust								A					

 Table 2.2:
 Potential emissions related to fuel cycle stages enabling electricity generation in large combustion plants; adapted from European Commission (2006b)

Among these impacts, the most relevant in view of policy constraints are related to emissions of greenhouse gases, e. g. carbon dioxide (CO_2) or methane (CH_4) and so-called classical air pollutants, including nitrogen oxides (NO_x) , sulphur dioxide (SO_2) , particulate matter (PM), ammonia (NH_3) , and non-methane volatile organic compounds (NMVOC).

2.4.1 Greenhouse gas emissions: contribution of fossil fuel power plants and regulatory framework

Even though the methods developed in this thesis concern classical air pollutants, some statistical and legislative background on greenhouse gas emissions is provided due to their relevance for climate policies. Over the period from 2005 to 2014, "fuel combustion in energy industries" was causing over a quarter of reported anthropogenic greenhouse gas emission in the EU 28 (Figure 2.4). Within this category, the bulk of emissions is due to the combustion of fossil fuels, next to a minor contribution from biomass combustion.



Figure 2.4: Annual EU28 greenhouse gas emissions per sector; adapted from eurostat (2016a)

Anthropogenic greenhouse gas emissions contribute to global warming, implying largely unknown, yet potentially drastic societal and economic consequences (IPCC 2014b). Greenhouse gas emissions from electricity generation and other sectors are therefore regulated under the EU's Emission Trading Scheme (ETS; Directive 2003/87/EC) that established an EU-wide market for emission allowances, whilst limiting the overall amount

of emissions from the concerned sectors. Greenhouse gas emissions from non ETS sectors are regulated under the EU's so-called Effort Sharing Decision (Decision No 406/2009/EC).

Ambitious future greenhouse gas emission reduction targets set out in the European energy policy framework (Table 2.1) are expected to lead to increasing prices of emission allowances². This in turn shifts generation costs of fossil fuel power plants upwards, reducing their competitiveness. A possible emission mitigation measure for greenhouse gases at fossil fuel power plants is carbon capture and storage (CCS). Yet, the economic, environmental and societal perspectives of CCS remain doubtful, as more and more demonstration projects have been scaled back or cancelled (Scott et al. 2013).

2.4.2 Classical air pollutant emissions: contribution of fossil fuel power plants

In 2014, the sector "Public Electricity and Heat Production" contributed around 17%, 47% and 4% to the reported European NO_x, SO_x, and PM_{2.5} (particulate matter with an aerodynamic diameter below 2.5 μ m) emissions (Figure 2.5 to Figure 2.7)³. These substances cause adverse effects on human health and ecosystems (cf. section 2.4.4) and have therefore been regulated in the EU for several decades (cf. section 2.4.5).

² So far the EU ETS has not proved very effective in increasing the price of carbon emission allowances. This is notably due to an oversupply of emission allowances, caused by forecasting uncertainties amongst others.

³ The data, reported under the CLRTAP (cf. section 2.4.3), excludes international shipping, aviation and natural emissions (European Environment Agency 2014b). Note that there are slight differences in the sector classification as compared to Figure 2.4, however concerning only the category non-road transport.



Figure 2.5: Annual EU28 NO_x emissions derived from Nomenclature For Reporting (NFR) 2014 sector classification; based on European Environment Agency (2016)



Figure 2.6: Annual EU28 SO_x emissions derived from NFR 2014 sector classification; based on European Environment Agency (2016)



Figure 2.7: Annual EU28 PM_{2.5} emissions derived from NFR 2014 sector classification; based on European Environment Agency (2016)

2.4.3 Formation mechanisms of air pollutants

Atmospheric emissions of relevance in the current thesis are formed during combustion processes. During atmospheric dispersion, they partly transform into secondary pollutants that, like certain primary pollutants, cause human health impacts (cf. section 2.4.4). Therefore, the pollutant formation mechanisms are briefly described.

Sulphur dioxide (SO₂)

Sulphur dioxide is a product of the complete combustion of sulphurous fuels such as coal or heavy fuel oil. The chemical composition and concentration of the sulphur compounds are strongly dependent on the type of fuel. For the CBA in section 7.3.1 the sulphur content in heavy fuel oil is assumed to vary between 0.22% and 0.33%.

Nitrogen oxide (NO_x)

The substances nitrogen monoxide (NO) and nitrogen dioxide (NO₂) are summarised using the generic term nitrogen oxide (NO_x). During combustion processes mainly NO is formed

contained in air depends on the availability of oxygen and is favoured by high temperatures (above 1300°C). At temperatures below 1000°C the formation of NO depends on the amount of available oxygen and additionally on the nitrogen-content of the fuel (fuel NO). Prompt NO, which is less important, is formed in the oxygen-deficient part of the combustion chamber (Baumbach 1994, European Commission 2006b).

Primary and secondary particulate matter (PM)

Particulate matter is a general term referring to dust particles of different sizes. These dust particles can either be suspended in gas or in liquid. Dust particles surrounded by air are called aerosols.

A classification of PM can be performed based on the aerodynamic diameter of the particles. PM with a diameter of less than 10 μ m is often referred to as fine particles. Fine particles are subdivided into:

- PM₁₀: particles with an aerodynamic diameter of less than 10 μm (including PM_{2.5});
- $PM_{2.5}$: particles with an aerodynamic diameter of less than 2.5 μ m.

The fraction of particles with a diameter between 2.5 μ m and 10 μ m is called coarse fraction. Other possibilities to characterize PM are by its chemical composition or by particle numbers.

Atmospheric PM emerges from the operation of fossil-fired power plants either as primary PM or as secondary PM. Moreover, PM also comprises natural components, such as mineral dust or sea salt.

Primary PM is formed inside the power plant and results from mineral compounds in the fuel used. Coal used in power stations, for instance, consists of 5 to 40 % mineral compounds that cannot be burned and that are transformed into different kinds of particles in conversion processes. Metal compounds can also be found as part of PM.

Secondary PM is formed where substances like SO₂ and NO_x react with ammonia (NH₃) to build so-called secondary inorganic aerosols (SIA). The products formed in these oxidizing reactions are ammonium sulphate ((NH₄)₂SO₄) and ammonium nitrate (NH₄NO₃). NH₃ preferentially reacts with SO₂ to form sulphate and the composition of secondary particles depends thus on the availability of both SO₂ and NO_x (Lee et al. 2015). Another type of secondary PM are secondary organic aerosols (SOA) that are formed on the basis of non- methane volatile organic compounds (NMVOCs), but whose formation mechanisms are generally less well understood than those of SIA (Hallquist et al. 2009).

Ozone

Ozone (O₃) is not directly emitted by power plants but formed in a secondary reaction. The principal substances of anthropogenic origin that are involved in this photochemical process, i.e. depending on the availability of sunlight, are NO_x, and NMVOC. They are therefore termed precursor emissions. Both the dissociation of NO₂ through sunlight as well as the oxidation of NO by NMVOC contribute to ozone formation that is therefore influenced by the availability of both substances in a non-linear way. For instance, in a typical metropolitan area setting with limited VOC and high levels of NO_x, reductions in NO_x emissions may lead to an increase in ozone formation (Finlayson-Pitts and Pitts 1993).

Carbon oxides

During combustion processes, the carbon contained in fossil fuels reacts with oxygen from the surrounding air to build CO₂. To guarantee a complete combustion, a certain excess of air or oxygen is therefore needed. As an intermediate product and being an indicator of an incomplete combustion, carbon monoxide (CO) is formed (Baumbach 1994).

Other pollutants related to combustion processes

- Non methane volatile organic compounds (NMVOC): NMVOC are emitted from fossil fuel power plants in small amounts, resulting from incomplete combustion or from fugitive emissions;
- Trace elements: Trace elements, such as arsenic (As), chromium (Cr), copper (Cu), mercury (Hg) or lead (Pb) can either be attached to particles but also, as for instance Hg, appear in the gaseous phase as a part of the flue gas, mainly related to coal combustion (European Commission 2006b). These trace elements are regularly also referred to as heavy metals even though arsenic needs to be classified as a metalloid;
- Ammonia (NH₃): Small emissions of NH₃ from can result from processes linked to the condenser of steam turbine power plants (European Commission 2006b). NH₃ is also used for NO_x reduction in the selective catalytic reduction (SCR) process. As described above, NH₃ is involved in the formation of SIA and also contributes to acidification and eutrophication.

2.4.4 Environmental and health impacts caused by classical air pollutants

Atmospheric emissions of substances like SO₂, NO_x, and PM, either directly or through the formation of secondary pollutants, have the potential to cause effects on human health, building materials, agricultural crops, and biodiversity (including ecosystem services). In Table 2.3, those environment and health impacts are displayed that are quantifiable in monetary terms using methods and models developed in the ExternE (Externalities of Energy) project series (cf. section 3.5.3) and follow-up projects. For a comprehensive list of environmental and health impacts caused by fossil fuel power plants, see Table A.1 (Appendix).

Impact category	Precursor substances	Effects caused by primary and secondary pollutants
Human health	PM ₁₀ PM _{2.5} SO ₂ NO _x NH ₃ NMVOC	Mortality, i.e. life expectancy reduction due to short- and long-term exposure to primary and secondary PM, ozone, and NO ₂ Respiratory and cardiovascular diseases due to expo- sure to primary and secondary PM, ozone, and NO ₂
Building Materials	SO ₂ NO _x NH ₃	For a detailed list of health endpoints, cf. section 6.1.3 Damage to construction materials caused by acidifying substances (particularly SO_2)
Agriculture	SO ₂ NO _x NH ₃	Variation in crop yields (due to SO ₂ and ozone) Acid input to agricultural soils Eutrophication of agricultural soils, avoided fertiliser use (through N deposition)
Ecosystems	SO ₂ NO _x NH ₃	Loss of (plant) species due to eutrophication and acidi- fication

 Table 2.3:
 Main quantifiable environment and health impacts caused by classical air pollutant emissions (European Commission 2005a, Preiss and Klotz 2008, WHO 2013a)

2.4.5 The regulatory framework targeting classical air pollutant emissions

Transboundary effects of air pollution have been addressed as early as 1979 through the United Nations Economic Commission for Europe (UNECE) Convention on Long-Range Transboundary Air Pollution (CLRTAP). Although much progress has been made since to reduce emissions from anthropogenic sources, air quality is still high on the political agenda. The EU's 7th Environmental Action Programme (EAP; Decision No 1386/2013/EU) reiterates the intention to reduce air pollution to levels "that do not give rise to significant negative impacts on and risks to human health and the environment".

Due to its key contribution to anthropogenic emissions of classical air pollutants, fossil fuel power plants have been regulated for many years through different types of instruments at different levels (Table 2.4).

At international level, they fall under the scope of the UNECE CLRTAP that can be classified as a common strategic objective, defining general targets that are pursued through further legislation. The CLRTAP is implemented through several sub-protocols. At EU level, these sub-protocols have been transposed through specific pieces of legislation, e.g. the National Emission Ceilings Directive (NECD; 2016/2284/EU) that ensures the respect of the Gothenburg Protocol targets. Reflecting the subsidiarity principle, EU Directives need to be transposed into member state legislation within given delays. Several examples of national transpositions for the case of Germany are given in Table 2.4.

The more recent Thematic Strategy on Air Pollution (TSAP, European Commission 2005b) can be regarded as a common strategic objective at EU level. Its targets are to be achieved, inter alia, through the Ambient Air Quality Directive (AAQD; 2008/50/EC) and also the NECD. The NECD is a special case in that it sets overall national emission ceilings to be respected in given years, but without prescribing the instruments to achieve these targets. More specific regulations are therefore required in parallel, categorised into source-oriented legislation, fuel quality and air quality standards in Table 2.4.

Among the examples, emission control at fossil fuel power plants is largely addressed by the Industrial Emissions Directive (IED, Directive 2010/75/EU), effective since January 2013. For power plants with a rated thermal capacity above 50 MW, it sets out emission limit values for NO_x, SO₂, and dust (particulate matter), unless derogations are used (cf. section 2.4.6). Emissions from power plants between 1 and 50 MW thermal capacity are covered by the Medium Combustion Plants Directive (MCPD; Directive EU/2015/2193).

Focusing predominantly on local air quality, the Environmental Impact Assessment Directive (EIAD; 2014/52/EU) sets out requirements to be respected before an operation permit can be obtained. For instance, power plant emissions shall not lead to exceedances of predefined local concentration thresholds. This condition is typically verified using health risk assessment (cf. section 3.4.2).

	International	European Union	Member State level (Germany)	Local level
Common strategic objectives	UNECE CLRTAP, featuring 8 sub-protocols, targeting classi- cal air pollutants (e.g. through the Gothenburg Protocol), per- sistent organic pollutants, heavy metals, etc.	Thematic Strategy on Air Pollution (TSAP; COM(2005) 446 final)		
National emission ceilings	The 1999 Gothenburg Protocol (to Abate Acidification, Eu- trophication, and Ground-level Ozone)	National Emission Ceilings Directive (NECD; 2016/2284/EU)	e.g. 39. BImSchV	
Source- oriented legislation		Industrial Emissions Directive (IED; 2010/75/EU) Medium Combus- tion Plants Directive (EU/2015/2193)	e.g. 4. BImSchV, 13. BImSchV	
Fuel quality standards		Sulphur Content of Certain Liquid Fuels (1999/32/EC)	e.g. 10. BImSchV	
Air quality standards		Ambient Air Quality Directive (2008/50/EC)	e.g. 39. BImSchV	Air quality plans
		Environmental Im- pact Assessment Directive (EIAD; 2014/52/EU)	Gesetz über die Umwelt- verträglich- keitsprüfung	

 Table 2.4:
 Regulations targeting classical air pollutants and relevant to fossil fuel-fired power generation, differentiated by category and scope (own compilation)

2.4.6 Best available techniques, the industrial emissions directive and the concept of disproportionate costs

In the EU, emission abatement (beyond greenhouse gases) is largely achieved by requiring that industrial installations are equipped with best available techniques (BAT), a concept introduced in the frame of the Integrated Pollution Prevention and Control Directive (IPPC, Directive 2008/1/EC, cf. European Parliament and Council of the European Union (2008)). The IPPC Directive and six other Directives were replaced by the Industrial Emissions Directive (IED, Directive 2010/75/EU, cf. (European Parliament and Council of the European Union 2010)) effective since 2013. In its annexes, the IED sets out emission limit values (ELVs) for substance releases into air and water. It reinforces the use of BAT, defined as follows (IED, art. 3-10):

"'best available techniques' means the most effective and advanced stage in the development of activities and their methods of operation which indicates the practical suitability of particular techniques for providing the basis for emission limit values and other permit conditions designed to prevent and, where that is not practicable, to reduce emissions and the impact on the environment as a whole [...].

'available techniques' means [...] under economically and technically viable conditions, taking into consideration the costs and advantages, [...]"

The IED (art. 14-3) demands that BAT conclusions⁴ are used as future basis for granting operation permits of concerned installations. These BAT conclusions are defined in the updated BAT REFerence (BREF) document on large combustion plants that has been adopted in 2017 and that will become effective in the year 2021. For power plant operators this implies that more stringent ELVs than those currently given in the IED's annex will become applicable in future years. In turn, power plant operators would need to decide whether employing additional emission control measures makes economic sense.

However, beyond other flexibilities, the IED (article 15-4) foresees a possibility for derogation, "where an assessment shows that the achievement of emission levels associated with the best available techniques as described in BAT conclusions would lead to disproportionately higher costs compared to the environmental benefits".

⁴ BAT conclusions are synthetic descriptions of Best Available Techniques (BAT) including the most relevant information, such as abatement levels of different emission reduction techniques. In addition, they will serve as reference for setting the permit conditions of future and refurbished power plants.

The following criteria may apply for costs being considered disproportionate:

"(a) the geographical location or the local environmental conditions of the installation concerned; or (b) the technical characteristics of the installation concerned." (ibid.)

The notion of "disproportionality" is crucial in this context, as it allows considering private costs and environmental benefits jointly, which refers to social CBA, further described in section 3.2. Under the increasingly tense economic conditions for power plant operators described above, the IED thus allows assessing investments in emission reduction measures on a case by case basis in terms of their overall societal efficiency. Mainly for this reason, the methodological developments in the current thesis bear an interest for power plant operators.

2.5 Summary of chapter 2

Fossil fuel-fired power generation, as a traditional and transitional energy source, continues to play a central role in the European energy mix, especially with regard to security of supply. Yet, one of its historic strengths, i.e. competitiveness, has come under increasing pressure through changes in the electricity system ("merit-order effect") and associated flexibility requirements.

Another challenge for fossil fuel-fired power generation are environmental regulations that aim at reducing the environmental and health damages related to atmospheric emissions. Source-oriented regulations, such as the EU's Industrial Emissions Directive, prescribe emission limit values, potentially requiring to retrofit fossil fuel power plants with new emission control measures. Given the increasingly challenging business context, it is crucial for operators to verify that such investments do not bear disproportionate costs. One method to do so, also in view of the requirements of the EU's Industrial Emissions Directive, is social cost-benefit analysis, introduced in the next chapter.

3 Methods: economic assessment of emission control measures

Emission control measures can be evaluated on a techno-economic or environmentaleconomic basis (Figure 3.1). The focus henceforth is on environmental-economic assessment methods.



Figure 3.1: Economic methods for assessing emission control measures; derived from European Commission (2006a)

This chapter therefore starts by introducing the discipline of environmental economics, in particular the neoclassical approach to dealing with environmental problems, involving the notions of externalities and cost-benefit analysis (CBA; cf. sections 3.1 and 3.2). To seek an "optimal" pollution level, not only private cost characterisation methods are needed (cf. section 3.3), but also methods to assess the societal benefits of pollution control. The latter can be achieved using the damage cost approach (cf. section 3.4). In a state-of-the-art review (cf. section 3.5) it is shown how the presented methods have been put into scientific practice. Essentially, this review serves to identify the gaps and limitations of currently available methods that this thesis aims to overcome.

3.1 Environmental economic theory underlying the assessment of emission control measures

3.1.1 Defining environmental economics

Different definitions of environmental economics coexist (Walz 2009). Sticking closely to the wording, a straightforward definition is:

1) The application of economic theory with regard to environmental problems.

The neoclassical economic foundations underlying this definition are described in the following (cf. section 3.1.2).

An alternative definition of environmental economics, reflecting a more empirical perspective and corresponding with the methodological developments carried out in this thesis, is:

2) The assessment of environmental impacts using economic methods.

Even though the methods presented in sections 3.2 to 3.4 and applied throughout chapters 4 to 7 extend beyond the discipline of environmental economics, they serve to answer environmental-economic-related questions. Finally, environmental economics can be understood as:

3) The analysis and application of economic policy instruments with regard to environmental problems.

Although environmental policy and its impact on industrial activity are a driving factor behind the methodological developments of this thesis (cf. section 2.4.6), policy instruments themselves are only briefly mentioned in section 3.1.6. For a microeconomic-oriented discussion on environmental policy instruments, see for instance von Böventer (1995).

3.1.2 The neoclassical approach to dealing with environmental problems

Neoclassical economics is based on two central interrelated principles (Endres and Fraser 2011):

- Scarcity: Scarcity is a central pillar of microeconomics. Scarcity of environmental resources, in their role as a production factor but also for providing supporting or cultural services⁵, has stimulated the development of environmental economics as a scientific discipline. In a theoretic setting, scarcity drives individuals to consider utility gains or losses of their economic (trans)actions, triggering welfare optimisation and thereby enabling the functioning of markets as allocation mechanism;
- An efficient outcome of the market mechanism, i.e. a free market functioning without the need for government intervention.

In the particular case of environmental resources however, the market mechanism is regularly disturbed due to:

- A lack of clearly defined property rights: Environmental resources often constitute public goods, characterised by non-rivalry and non-excludability in their consumption (von Böventer 1995, Walz 2009). This makes them prone to overconsumption and partly explains another critical factor, i.e.
- A lack of market prices for environmental resources and the services they are offering to society.

As a result, environmental resources tend to be overused or degraded, implying a socially inefficient outcome of the market mechanism ("market failure"), and providing the rationale for policy intervention.

⁵ Supporting and cultural services refers to the concept of ecosystem services, i.e. the direct and indirect contributions of ecosystems to human well-being (TEEB 2010).

3.1.3 Rationale for employing environmental-economic methods in the context of industrial production activities

In the context of industrial production activities, environmental resources can be characterised as (Steven 1994):

- an input factor to the production process, e.g. as raw materials;
- an absorptive capacity for outputs from the production process, e.g. environmental media like air, water or soil that absorb releases arising from industrial activities.

In the first case, not using the environmental resource as an input factor implies an opportunity cost in the sense of foregone welfare (Walz 2009). In the second case, the unrestricted use of the environmental resource implies welfare losses through health or environmental impacts caused by released substances. If these impacts of economic activity on third parties remain unpriced, i.e. outside the market, externalities arise, as further described in sections 3.1.5 and 3.1.6. Taking a public welfare perspective, environmental economic methods are therefore instrumental to:

- put a price tag on the environment and its resources in order to enable their integration into the market mechanism;
- assist in defining the appropriate level of policy intervention, i.e. seeking a balance between all quantifiable societal (private as well as external) costs and benefits related to an industrial activity. To this end, social CBA is an appropriate method, cf. section 3.2.

3.1.4 Limitations of the neoclassical approach

Some general limitations of the neoclassical approach to solving environmental problems shall be discussed, whilst a more specific discussion in light of the results of this thesis can be found in section 8.4.

With regard to sustainable development, it is criticised that the neoclassical approach relates to the principle of so-called weak sustainability (Hohmeyer and Ottinger 1994, Neumayer 2010, Walz 2009). This concept dates back to the economist Solow, who worked on the substitutability between manmade human capital and natural capital (Solow 1974). Weak sustainability assumes a strong substitutability of these two forms of capital. It follows that, from an inter-generational perspective, it would be sufficient to maintain the total capital stock and not necessarily the natural capital as such. Apart from risking to extinguish natural capital and thereby losing critical life supporting functions, such a principle bears the risk of neglecting so-called tipping points or maximum environmental capacities, e.g. thresholds beyond which uncontrollable or irreversible damage occurs.

The assessment of environmental externalities and their use in social CBA is subject to several ethical limitations. Basically, these approaches take a utilitarian, anthropocentric viewpoint, i.e. defining well-being on the basis of human utility. It is criticised that, in many cases, this results in ignoring or underestimating the value of natural systems, for instance due to the fact that existence values and other intangible components are often difficult to consider (Ott 2014, Walz 2009). Further on, the neoclassical approach largely ignores distributional aspects across populations but also generations (Hohmeyer and Ottinger 1994). Those affected by environmental or health damage are often treated as a homogenous mass, whereas it could be worthwhile to analyse the people affected in further detail, e.g. in order to protect particularly vulnerable subpopulations. When transferring methods or parameters across space or time, methodological choices, e.g. on the discount rate to be used or on equity weighting, have implications that are often not sufficiently accounted for in decision-making. Moreover, according to Walz (2009), indirect impacts, occurring over longer time periods, are often not sufficiently accounted for when assessing environmental externalities ("monetary underestimation").

3.1.5 Definitions related to externalities theory

In economic terms, externalities are a market failure that leads to a divergence between equilibrium and social optimum, as exemplified below.

Just et al. (2004) define an **externality** "as the case where an action of one economic agent affects the utility or production possibilities of another in a way that is not reflected in the marketplace."

External costs are a monetary measure of negative externalities.

A notion often encountered for describing negative environmental externalities and related to the main method for measuring these (cf. section 3.4) is **damage costs**. In contrast to external costs, which, by definition, only refer to the non-internalised fraction of the total environmental damage costs (Rabl and Holland 2008), damage costs include internalised as well as non-internalised effects. Given that it is generally difficult to define the degree of internalisation of specific external effects, the term damage costs is used henceforth to describe the quantified monetised impacts due to air pollution.

Micro economic theory is primarily interested in defining **marginal effects**, meaning those effects attributable to the last (= marginal) unit of a good or service. In the given context, for instance, air pollution-related damage or abatement costs can be estimated per tonne of a substance emitted or removed, yielding marginal damage or abatement costs respectively.

3.1.6 Production externalities and internalization through policy instruments

Externalities theory is rooted in micro-economics and focuses on interactions between economic agents and related welfare losses (von Böventer 1995). Externalities linked to industrial production processes were introduced in the early 20th century by economist Arthur Pigou. After observing that the production of certain goods had unintended side-effects that are not accounted for in its market price, he distinguished private and external effects of production (Pigou 1912).

The influence of a negative production externality on welfare is illustrated in Figure 3.2, showing an equilibrium allocation of goods, described by the intersection of the supply and demand curve, in the absence (left) or presence of an externality (right).



Figure 3.2: Illustration of a negative production externality; left: quantity (q) and price (p) under a competitive equilibrium (E); right: new equilibrium (E*) when internalising a negative externality, also showing associated welfare losses (w) if internalisation did not take place; derived from Pigou (1932)

When accounting for the externality, the cost curve is shifted upwards, and marginal private costs become marginal social costs. This leads to a new, socially optimal, equilibrium in which the welfare loss, originally induced by the externality, has been internalised. In this example, internalisation of a negative environmental production externality induces a price increase, whilst at the same time reducing overall societal costs by reducing the demand (from q to q*).

In addition to introducing the concept of production externalities, Pigou is also known for the so-called Pigouvian tax, a policy instrument intended to correct for negative externalities (Pigou 1932). Yet, the need for government intervention was not unanimously accepted throughout economic history. Its most prominent challenger was Coase (1960), who is known for the following theorem. It states that, where property rights are clearly defined and transaction costs are low, markets will automatically arrive at optimal allocations of resources, without the need for (government) intervention beyond potentially establishing the property rights. Effectively this process would result in an automatic internalisation of externalities.

As already mentioned, clearly defined property rights for environmental resources are often lacking. In addition, in cases of environmental pollution, transaction costs tend to be high due to the large number of potentially impacted stakeholders. For these reasons, the need for policy intervention through environmental regulations (cf. section 2.4.5) is nowadays well accepted. Two broad groups of policy instruments are distinguished, their respective advantages and shortcomings being the subject of a lively and enduring debate among economists (Pearce 2002):

- Incentive-based, also called market-based instruments. These include pollution taxes, e.g. a carbon tax, as well as marketable permits, e.g. the EU ETS;
- Quality standards, also called command-and-control measures. These include technological or performance standards, e.g. defining emission limit values as in the case of the EU IED (cf. section 2.4.3).

3.2 Cost-benefit analysis of emission control measures

3.2.1 The socially "optimal" level of pollution

Social CBA is a method to approximate the socially optimal level of pollution, at which the marginal private costs of pollution abatement equal the associated marginal social benefits (Figure 3.3).



Figure 3.3: Illustration of the optimal level of pollution (O), associated marginal costs (c), marginal benefits (b) and level of emission abatement (a); derived from Samuelson and Nordhaus (2007)

The illustration represents a simplification in that the exact shape of both marginal curves depends on a variety of factors, e.g. the geographic location, the substance emitted, back-ground conditions as well as the information available for quantifying costs and benefits. In economic textbooks, both marginal curves are typically assumed to be linear, derived from convex underlying cost and benefit curves. Yet, non-linear representations of the marginal curves also exist (Pappin et al. 2015). When regarding a bundle of pollutants at an aggregated level, it is plausible to assume a decreasing marginal societal benefit from pollution abatement, e.g. due to concentration thresholds for pollutants like ozone or ni-

trogen dioxide, below which further impacts are less likely to occur. Likewise, the marginal private costs are assumed to increase at higher levels of emission abatement due to technological constraints and the resulting additional efforts needed to further reduce emissions. The above example shows that a socially optimal level of emission control typically consists neither in unrestricted emissions nor in a complete elimination of emissions.

3.2.2 The decision rule in social CBA

CBA relies on a comparison of social costs and benefits, i.e. including private as well as external effects. In order to separate it from a purely private CBA, the term social CBA is used. In social CBA, the discounted benefits of a policy, programme or project (in the following simply called measure) are compared to the associated discounted costs (Pearce et al. 2006). While both absolute (based on the net present value, cf. below) and relative (based on the ratio) comparisons of benefits and costs are common, the former is considered more meaningful as it conveys information on the magnitude of the result. In case of a positive net present value, a measure is said to be advantageous or acceptable from a societal perspective, as formalised by the decision rule in equation 3.1.

Net present value: $\{\sum_{i,t} B_{i,t} \times (1+s)^{-t} - \sum_{i,t} C_{i,t} \times (1+s)^{-t}\} > 0$ 3.1

where:

- *B* refers to the benefits. In the context of this thesis, these consist of avoided health damage costs (cf. sections 3.4 and 6.1);
- *C* refers to the costs. In the context of this thesis, these consist of investment and operating costs, as a function of the measure's life time (t) and operating hours per year (cf. sections 3.3 and 6.2);
- *s* refers to the discount rate that is used to estimate the present value of future costs and benefits (cf. section 3.2.3);
- *i* refers to the affected individuals. These are determined by the scope of the analysis, i.e. the geographic extent of the modelling domain (cf. section 6.1.1).

Ex-ante (before a measure is carried out) and ex-post (after a measure is carried out) CBAs are distinguished. In the regulatory context, ex-ante CBA is more commonly used.

3.2.3 Discounting in CBA

Ex-ante CBA relies on discounting future costs and benefits, requiring value trade-offs between the future and the present. In policy appraisals, discount rates larger than zero are regularly used. This implies that future costs and benefits are given less weight than today's costs or benefits (Pearce et al. 2006). Which discounting scheme (value of the discount rate, constant vs. declining) to use remains a contentious issue and depends strongly on the context. Social discount rates, used in public policy-oriented social CBA, can be determined in two ways (Boardman et al. 2006, Harrison 2010, Zhuang et al. 2007):

- using the rate of return on risk-free investments, reflecting the time value of money, e.g. 10 year USA government bonds⁶;
- using the utility-oriented "Ramsey formula", composed of
 - the pure rate of time preference, and
 - the consumption growth rate (gross domestic product (GDP) per capita) multiplied by the elasticity of marginal utility of consumption.

Discount rates for use in policy impact assessments are provided by government institutions. The European Commission (2009) recommends a real social discount rate of 4% for both costs and benefits, which is adopted in this thesis (cf. section 6.3). Real discount rates, as opposed to nominal discount rates, include effects related to price level changes (inflation). When using the Ramsey formula as an orientation, a 4% social discount rate implies a long-term average consumption growth rate of about 2% (Steinbach and Staniaszek 2015). In France, a 4.5% real social discount rate is recommended by Quinet et al. (2013), which is slightly higher than the 3% real social discount rate recommended in Germany by the Umweltbundesamt (2012) for time periods up to 20 years into the future.

Using a social discount rate in the CBA is complementary to using a private interest rate (e.g. the weighted average cost of capital, cf. section 3.3) for the conversion of investment costs into annuities. Private interest rates differ from case to case, as they reflect the opportunity cost of capital and risks related to investment decisions (Klingelhöfer 2006). For this reason, the private interest rate related to emission control investments is varied in the applied part of this thesis, demonstrating the sensitivity of CBA results with regard

⁶ Given the current period of relatively low interest rates (see <u>http://www.bloomberg.com/markets/rates-bonds/government-bonds/us</u>, last accessed: 2017-05-18), this would lead to a lower discount rate than recommended by most government agencies.
to this parameter (cf. section 7.3.3). Private interest rates are typically higher than social discount rates given the higher risk and opportunity costs of private sector investments as opposed to public policy investments. The combined use of a private interest rate and social discount rate, as in chapters 5 and 7 of this thesis, is called the two-stage discount-ing procedure, adopted from Kolb and Scheraga (1990).

3.3 Techno-economic assessment of emission control measures

Techno-economic assessments are commonly used in production economics, serving for instance to:

- Identifying the most cost-effective emission control measure, e.g. assessing how a given emission reduction target can be reached in the least costly way;
- Estimating the marginal costs of emission reduction (also called marginal abatement costs), e.g. expressed in € per tonne of emission reduced. This includes a private cost characterisation of emission control measures. Under certain conditions, emission abatement costs can also be used as a proxy for environmental damage costs (cf. section 3.4);
- Optimising the functioning and performance of specific emission control measures, e.g. using simulation or optimization modelling and other engineering approaches.

Basic guidance on cost-effectiveness analysis in the context of European air quality legislation is available in the "Integrated Pollution Prevention and Control Reference Document on Economics and Cross-Media Effects" (European Commission 2006a), whose revision is pending. One central recommendation is to use the annuity method, which is presented below. On a more detailed basis, Klingelhöfer (2006) describes methods for the private financial assessment of environmental protection measures whilst distinguishing different contexts defined by different environmental policy instruments.

3.3.1 Cost characterization of emission control measures using the annuity method

In this thesis, an exemplary cost characterisation of different emission control measures at a large combustion plant is carried out (cf. section 6.2.2), based on the annuity method. For this purpose, the total costs of emission control measures are decomposed into:

- Capital Expenditures (CAPEX), also called investment, including expenditures for pollution control equipment and its installation, as well as contingency;
- Operational Expenditures (OPEX), typically composed of a fixed and a variable component. The fixed component covers costs for maintenance, insurance and wages. The variable component covers
 - electricity consumption,
 - products used for operation (e.g. reagents),
 - waste or by product-related costs (or benefits for valorised residues),
 - other equipment-related costs (e.g. replaced components).

Following the principles of the annuity method (equation 3.2), capital expenditures are decomposed into equivalent annual components that implicitly account for the underlying cost of capital, defined through the interest rate and equipment lifetime:

$$C_{ann,CAPEX} = CAPEX \times \frac{(1+i)^t \times i}{(1+i)^t - 1}$$
3.2

where:

- Cann, CAPEX refers to the annualised capital costs;
- *i* refers to the interest rate (weighted average cost of capital);
- *t* refers to the equipment lifetime.

The right term of equation 3.2 is also known as capital recovery factor. Total annual costs are obtained by summing the annualised capital costs and annual operational costs.

3.4 Environmental-economic assessment of emission control measures

Two principal approaches for the assessment of environmental-economic impacts related to atmospheric emissions have emerged:

- (Marginal) damage cost approach: Damage costs are assessed in a bottom-up way, starting at the emission source, tracing relevant substances throughout the environment and finishing by assessing their impacts at receptor level, e.g. human health effects. To this end, a step-wise procedure called impact pathway approach is used, further detailed below. When relating results to the last (i.e. marginal) activity unit, the term marginal damage costs is used.
- (Marginal) abatement cost approach, also termed standard-price approach: As a second-best option, marginal abatement costs serve to approximate damage costs under equilibrium conditions, i.e. where the marginal costs to prevent an extra unit of emission are assumed to equal the damage costs caused by this extra unit of emission (Baumol and Oates 1988). The approach therefore seeks to define the cost of the last (i.e. marginal) measure implemented to comply with a politically set goal, assuming that it corresponds to the welfare economic optimum. A typical field of application is greenhouse gas-related external costs, where the high level of uncertainty surrounding future impacts often prevents the application of a damage cost approach (Pearce et al. 1996, UBA 2008). A draw-back of the marginal abatement cost approach is that it cannot be used in CBA of related policies (i.e. greenhouse gas-related measures in the example given above). Since benefits are assumed to at least equal costs, the conditions for passing the CBA decision rule will implicitly be fulfilled.

Beyond these rather sophisticated approaches, benefit transfer (also called value transfer), i.e. the transfer of data or values from an original study site to a policy application site, represents another option, especially in situations where original studies cannot be carried out due to lacking resources (Navrud and Ready 2007). Given the potentially large uncertainties involved, the use of benefit transfer and in particular the compatibility between study and policy site, should be well justified.

3.4.1 Estimating marginal damage costs using the impact pathway approach

Where feasible, external costs are most precisely calculated in a site- and time-dependent way following the so-called impact pathway approach (European Commission 2005a), distinguishing steps similar to the Driver-Pressure-State-Impact-Response (DPSIR) scheme (Smeets and Weterings 1999), whilst neglecting the "Response" part. The impact pathway approach is an implementation of the marginal damage cost approach, featuring the following steps when analysing impacts of atmospheric emissions (Figure 3.4):

- Emission scenario characterisation and air quality modelling, i.e. defining emission source specifications and associated atmospheric emissions; using atmospheric dispersion and chemical conversion modelling (or other adapted techniques) to define ambient concentration or deposition changes at receptor level;
- Exposure assessment, i.e. estimating population (or other receptors') exposure to concentration or deposition changes. Different routes of exposure exist, e.g. inhalation or ingestion;
- Impact assessment, i.e. assessing physical impacts using baseline data and risk functions (also called exposure-response, concentration-response or doseresponse functions), derived from epidemiological or toxicological studies⁷;
- 4. Monetary valuation of impacts, i.e. transferring physical into monetary units, based on the loss of wellbeing (or utility) of concerned individuals. In some cases, this loss can be valued using market prices (e.g. for agricultural crops or treatment of illnesses). In the case of non-market goods, such as the quality of human health, valuation is based on individual preferences, either observed (revealed preferences) or directly expressed (stated preferences) using environmental-economic methods (Markandya and Ortiz 2011, Pearce et al. 2006), cf. also section 6.1.4.

⁷ Epidemiological studies provide evidence on the relations between concentration levels and risk changes through observations of population cohorts under natural exposure conditions, usually over longer time periods. Toxicological studies rely on controlled experiments to analyse the effects of substances on organisms (Savitz 1988).



Figure 3.4: Impact pathway approach for assessing health-related damage costs following the DPSIR scheme (oval shapes = input data; rectangles = output data; rounded rectangles = assessment steps); derived from European Commission (2005a)

In principle, the impact pathway approach allows assessing the effects of emissions on various receptors, e.g. humans, agricultural crops, building materials or ecosystems. Due to their predominant weight in quantifiable damage costs (e.g. Holland 2014a), however, the current thesis concentrates on human health-related impacts. In this particular case, the last three steps of the impact pathway approach can be conceptualised by equation 3.3 that allows expressing impacts first in physical and then monetary units:

$$C_i = \sum_r \underbrace{\Delta c_r \times p_r}_{E} \times f_{i,r} \times s_{CR_i} \times t \times m_i$$
3.3

where:

- Ci represents the damage costs related to health impact i, given in €base year;
- Δc_r is the concentration change of a given pollutant in a given sub-region, expressed in $\left[\frac{\mu g}{Nm^3}\right]$ for inhalation-related exposure;

- p_r is the number of affected individuals (persons); combined with Δcr , the total population exposure E is obtained, given in $\left[\frac{\mu g}{Nm^3} \times person\right]$, calculated by summing up over all sub-regions r of the assessment domain;
- *f_{i, r}* is the share of the population in sub-region r affected by health impact i [fraction], comprising information on the corresponding age and risk groups;
- s_{CR_i} is the slope of the impact function of health impact i, for inhalation given in $\left[\frac{additional\ cases}{Nm^2 \times person \times year}\right]$, merging information on the risk increase (mainly presented as relative risk and also called concentration-response function, cf. Table 6.7; here assumed to be linear with respect to concentration changes) and baseline rate of a given health impact i (IOM 2011);
- *t* is a factor to account for different assumptions on particle toxicity, by default and following latest recommendations (WHO 2013b) having a value of one, while values for secondary PM may deviate in sensitivity scenarios (cf. section 5.4.4.2);
- m_i is the monetary value per occurrence of health impact i, given in $\left[\frac{\epsilon_{\text{base year}}}{case}\right]$.

3.4.2 Distinguishing two policy-relevant types of health assessment

As briefly mentioned in the context of air quality-related regulations, different types of health assessment are commonly used for policy support (Table 3.1):

• Health impact assessments, often serving as an input to social CBA or for accountability studies⁸. These usually include a quantification of physical impacts at larger domains and often rely on advanced atmospheric dispersion modelling, e.g. using Euler models⁹;

⁸ Accountability studies aim at assessing the effectiveness of environmental or health policies, e.g. by estimating the public health benefits from respecting stricter ambient air quality standards (HEI 2003).

⁹ Eulerian models rely on solving differential equations to trace the flow of fluids in the atmosphere, e.g. driven by advection, diffusion, and chemical reactions (cf. section 6.1.1.2). This makes them suitable for modelling chemically reactive substances.

 Health risk assessments, typically used in the frame of permitting procedures. These often rely on Gaussian dispersion modelling¹⁰ and consider a limited, local spatial scale. Their main objective is to verify that toxicologically-defined risk thresholds are respected for a defined area.

Although partly relying on the same data basis, health impact assessment and health risk assessment thus differ in terms of application, scope, and results.

Following a definition of the World Health Organization (WHO 1999), "Health Impact Assessment is a combination of procedures, methods and tools by which a policy, program or project may be judged as to its potential effects on the health of a population, and the distribution of those effects within the population." The term thus encompasses a variety of quantitative and qualitative methods, differing in practical details, such as considered impacts, stakeholder involvement or timing (Mindell et al. 2003). In the given context, health impact assessment, as third step of the impact pathway approach, is used as a basis for estimating damage costs. Details on the approach implemented in the current work are provided in section 6.1.3.

Environmental impact assessment, relevant in permitting procedures, encompasses a variety of physical, biological or socio-economic effects, typically relying on health risk assessment when evaluating health effects. However, the use of health impact assessment is not excluded (CDC 2015).

¹⁰ Gaussian plume models assume that the vertical and horizontal concentration profiles of substances can be estimated using two independent normal distributions. Driven mainly by wind and atmospheric turbulence, the vertical and horizontal dispersion of a plume is calculated using simplified equations (Rabl et al. 2014). Gaussian models are therefore more suitable for modelling non-reactive substances.

 Table 3.1:
 Comparison of the main features of health impact assessment and health risk assessment applied to emission sources; based on CDC (2015)

	Health impact assessment	Health risk assessment
Approach	Assessing the potential impacts of a measure (characterised by so- cial, economic or environmental changes) on a population's health, including distributional effects	Assessing biophysical health risks from exposure to hazard- ous substances, e.g. chemicals
Application	Decision-making in voluntary and regulatory processes; using a ra- ther broad definition of health, in- cluding physical and psychological health and general well-being	Mainly regulatory decision con- text, e.g. permitting of industrial facilities under the EU Environ- mental Impact Assessment (EIA) Directive (2014/52/EU)
Result	Physical health endpoints, e.g. cases of illnesses; potentially us- ing aggregate metrics, such as dis- ability adjusted life years (DALYs) or monetary valuation	Risk quotients or number of ex- ceedances, e.g. testing whether predefined concentration thresholds or standards are re- spected
Scope	Local to hemispheric effects caused by one or several sources; exposure zone depending on the study design, but typically larger areas, e.g. country-level; one or several assessment years	Local effects caused by a specific source; exposure zone typically several kilometers around the source; typically several assess- ment years
Air quality modelling	Different kinds of atmospheric models used, depending on the study scale; regularly including at- mospheric chemistry processes, e.g. the formation of secondary pollutants in Eulerian models	Typically using local-scale Gauss- ian modelling, regarding the dis- persion of primary pollutants and with limited consideration of atmospheric chemistry

3.4.3 Uncertainty underlying the environmental-economic assessment of health damage costs

Health damage costs are inherently uncertain, as their assessment involves modelling, measurements, and expert judgements. Given their use in decision-making, particularly within social CBA, it is important to identify the nature and sources of uncertainty and discuss the implications for decision-making. This section serves to present relevant terms and definitions, uncertainty sources as well as uncertainty quantification approaches related to health damage cost assessments. A quantitative and qualitative discussion of uncertainty underlying the results of this thesis can be found in section 8.2.

Terms and definitions

Distinctly defining the nature and sources of uncertainty is not straightforward given that different definitions coexist, depending on the use context. A key question is whether variability should be considered as a type of uncertainty (the viewpoint adopted in this thesis) or rather as a distinct and separate concept, e.g. in the context of chemical risk or exposure assessment (U.S. EPA 2011b, WHO 2008). While uncertainty is precisely defined in certain domains, e.g. in the evaluation of measurement data (IPCC 1996, JCGM 2008), the multidisciplinary nature of the impact pathway approach requires a broader classification of uncertainty. Here, uncertainty definitions used in health risk assessment and greenhouse gas emission monitoring and assessment are adopted (Institute of Medicine 2013, IPCC 1996, Swart et al. 2009). A classification is proposed, covering different types of uncertainty based on their nature and the main underlying sources (Table 3.2).

Nature/type of uncertainty	Source
Aleatory (or exogenous) uncertainty: statistical variability and heterogeneity	Natural variations in the data or system being as- sessed
Epistemic uncertainty: model and parameter uncertainty	Lack of knowledge about the values, causal relation- ships or functional forms underlying models or data
Deep uncertainty	Uncertainty/ignorance about fundamental pro- cesses or assumptions underlying an assessment

Table 3.2:Uncertainty classification including the nature/type of uncertainty and associated sources;
adapted from (Institute of Medicine 2013, IPCC 1996, Swart et al. 2009)

Another relevant type of uncertainty in the given context is decision-rule uncertainty due to expert choices, cf. the literature cited in Bachmann (2006). Depending on the degree of knowledge available, it may be either related to epistemic uncertainty or deep uncertainty (in cases of serious disagreement).

Uncertainties underlying the impact pathway approach

When assessing health damage costs based on the impact pathway approach, variability, as well as model and parameter uncertainty are overlapping and it is not always possible to clearly separate them. The main types and sources of uncertainty at the different stages of the impact pathway approach are (cf. the discussion in section 8.2 and (Bachmann 2006, Kim et al. 2015, Mansfield et al. 2009)):

- Air quality modelling
 - Variability, e.g. temporal and spatial variability of input data according to the respective modelling resolution;
 - Model uncertainty, e.g. due to the structure of the mathematical models and the numerical schemes used for simulating complex atmospheric chemistry processes;
 - Parameter uncertainty, e.g. related to input data regarding meteorology, emission inventory, land use data etc.;
 - Decision-rule uncertainty, e.g. choices of the numerical approximations regarding grid resolution, vertical layers, and time resolution.
- Exposure assessment
 - Variability in population exposure patterns (temporally and spatially);
 - Model uncertainty related to the spatial and temporal resolution used;
 - Parameter uncertainty, e.g. related to population statistics.
- Health impact assessment
 - Model uncertainty, e.g. due to the functional form of concentrationresponse curves;

- Parameter uncertainty, e.g. related to the central estimate of the concentration-response function used;
- Decision-rule uncertainty, e.g. due to choosing the mortality impact assessment approach.
- Monetary valuation
 - Parameter uncertainty, e.g. related to the valuation of intangible goods;
 - Decision-rule uncertainty, e.g. related to the components (tangible versus intangible) to be considered for valuation.

As a key difference, advancements in knowledge allow to reduce epistemic uncertainty, whereas variability (aleatory uncertainty), as an inherent feature of the system being modelled, persists. Characterising variability, e.g. through information about the magnitude of the spatial or temporal variability, is useful when it comes to decision-making. Increasing the modelling resolution generally enables to better capture variability of input data. In particular it allows estimating the potential errors induced by the use of time- or space-averaged assessment approaches (cf. section 8.2).

Common approaches for uncertainty assessment of health damage costs

Uncertainty characterization methods can be broadly classified into screening methods, qualitative approaches, deterministic approaches, and probabilistic uncertainty analyses (WHO 2008).

A typical example for a qualitative uncertainty characterisation is the so-called review of unquantified biases, describing the potentially important, but unquantified influencing factors, as well as in what way they would impact the results (Holland 2014b, U.S. EPA 2012c). Such a review is carried out in section 8.2.3.

Commonly used quantitative uncertainty characterisation methods related to the assessment of health damage costs are (Bachmann 2006, Holland 2014b, Holland et al. 2005b, Spadaro and Rabl 2008):

• Scenario or sensitivity analysis. It is used to (deterministically) assess the effect of variations in model inputs or modelling choices on variations in model outputs (cf. chapter 7 of this thesis);

• Probabilistic uncertainty assessment. A typical and widely-used example of a probability parameter uncertainty assessment is Monte Carlo analysis. Essentially, it consists of two steps. First, probability distributions around central value estimates of key parameters used for health damage cost assessment are defined; second, a sufficiently large number of modelling runs is conducted, during which input parameters are simultaneously and randomly varied according to their predefined characteristics. The results give a broader overview on uncertainties than a single sensitivity analysis and also permit to identify the most influential input parameters. Yet, the quality of the results strongly depends on the quality of the input data, as also discussed in section 8.2.

3.5 State-of-the-art review including gap analysis: economic assessment of emission control measures

This section aims to review existing approaches for the techno-economic, environmentaleconomic, and CBA-based assessment of emission control measures (Table 3.3). The overview is not intended to be exhaustive, but includes those publications considered most relevant for the context of this thesis. Specific models for an environmental-economic damage cost assessment are presented in section 3.5.7. The following filter criteria were used in order to obtain relevant results from scientific databases and related literature:

- assessment of emission control measures for classical air pollutants released from point emission sources in the energy sector in a Western European (or North American) setting;
- applied research, i.e. involving some kind of modelling and results;
- research with a strong link to environmental policy-making.

Moreover, guidelines for deriving private costs of emission control measures were considered. Table 3.3:Overview on exemplary studies assessing emission control measures of classical air pollutants in
the energy sector, including information on their scope and models used (own compilation)

	Aggregated level (e.g. sector or region)	Site-level	Site-level and accounting for variability
Techno-	EU	EU	EU
economic assess-	Economic assessment of BAT (Schultmann et al. 2001)	Industrial emission control measures at site level (Rentz 1979)	Site-specific emission control costs (TETEL 2015)
ment	Reference installation approach (Geldermann and Rentz 2004)	USA	USA
	BREF large combustion plants (European Commission 2006b)	PM control strategies, US context (Atkinson and	Site-specific emission control costs
	EU-wide emission control strate- gies, GAINS model (Amann et al. 2014)	Lewis 1974)	(U.S. EPA 2002)
	National level		
	Sector-specific emission control scenarios, Germany (Breun et al. 2012)		
	Sector-specific emission control scenarios, France (MEEM 2016)		
Environ-	EU	EU	EU
mental- economic assess- ment	Electricity sector damage costs in EU countries, ExternE project se- ries, (European Commission 1999b, 2004) ELL and German damage cost as-	Damage costs EU, Risk- Poll/Uniform World Model, cf. section 3.5.7 (Curtiss and Rabl 1996, Rabl and Spadaro 2000)	Damage costs of Dan- ish facilities, EVA model, cf. section 3.5.7 (Andersen et al. 2006)
	sessment, EcoSense model, e.g. (Krewitt et al. 1999)	Damage costs Poland, EcoSenseWeb model, cf.	Health damage costs of power plants with
	Sectorial damage costs in four EU countries (Droste-Franke et al. 2005)	section 3.5.7 (Czarnowska and Frangopoulos 2012) Damage costs of European	variable operation profiles in Europe, cf. section 6.1
	EU and Danish damage cost assess- ment, EVA model, cf. section 3.5.7 (Brandt et al. 2013)	industrial facilities (European Environment Agency 2011, 2014a)	
	National level	USA	
	Social costs of energy consumption, Germany (Hohmeyer 1988)	US Damage cost assess- ment (Mendelsohn 1980,	
	Sectorial damage costs in Germany (Schwermer et al. 2014), (van der Kamp et al. forthcoming)	2007)	

	Aggregated level (e.g. sector or region)	Site-level	Site-level and accounting for variability
Social	EU	EU	EU
cost- benefit analysis	EU air quality policy (Holland 2014a, Watkiss et al. 2005) USA US air quality policy (U.S. EPA 2011a, 2012c) National level French air quality policy (MEEM 2016)	Case study of a coal-fired power plant, cf. chapter 5 (Bachmann and van der Kamp 2014) UK IED derogation cost- benefit analysis tool (Ricardo AEA 2017)	Social CBA methodol- ogy, cf. chapter 6

3.5.1 Early pioneering work

In one of the first works identified, Atkinson and Lewis (1974) used source-receptor transfer coefficients based on Gaussian air quality modelling to respectively define the effect of reducing PM emissions in a source region on concentration changes in receptor regions in the USA. In parallel, they also estimated emission control costs, however without proceeding towards a CBA, as no health impact assessment was carried out. The study's main objective was to define cost-minimal reduction strategies that would enable regions to respect given air quality standards. As a limitation in scope, the Gaussian modelling only considered primary PM dispersion at a restricted geographic scale.

Extending the former work, Mendelsohn (1980) coupled a techno-economic assessment with an environmental model that allows incorporating environmental damage costs. However, this work also steps short of a classical CBA, mainly due to the author's lack of confidence in central monetary values that could be assigned to health impacts. Using Gaussian air quality modelling based on simplified chemistry as well as dose-response functions that were qualified as relatively uncertain, health impacts were quantified primarily in physical units. In a sensitivity analysis, the author sought to define what levels of health costs would be needed in order to break even with the abatement costs of distinct abatement measures. Almost 30 years later on the same author co-published an article in which a damage cost assessment methodology is applied in the US context (Muller and Mendelsohn 2007).

As one of the pioneers in Europe, Rentz (1979) carried out extensive work on the technoeconomic assessment of industrial emission control measures. Notably, he pointed out the non-feasibility of CBA, mainly due to lacking capabilities for a regionalised, site-specific exposure assessment that accounts for atmospheric chemistry of a "pollutant mix". In addition, lacking scientific capabilities for a health impact assessment were mentioned. A notable recommendation given by Rentz (and followed in the current thesis) is to proceed towards the "disaggregated level" for an assessment of emission control measures.

Regarding the environmental-economic assessment of energy conversion technologies, Hohmeyer (1988) can be counted among the pioneers in Europe, estimating social costs of energy, including health damage costs. As a limitation, he relied on damage cost factors from literature studies, without carrying out an original modelling for the German context. Social costs of energy were used to point out the competitive disadvantage of renewable energy sources compared to conventional energy sources in the electricity market, brought about by the lacking consideration of the external costs. The study thus made an early call for renewable energy support schemes.

3.5.2 Costing methodologies

Reviewing costing methodologies is hindered by the fact that emission control costs are site- and context-dependent. For this reason, this review focuses on methodological background documents and spreadsheets that permit the evaluation of power plant-specific emission control costs. In parallel, studies describing reference installations or technologies and respective abatement costs for use at a more aggregated level are also considered.

A comprehensive guidance for defining air pollution control costs of technical measures at fossil fuel power plants is given by U.S. EPA (2002). Parts of the same methodology have also been implemented in spreadsheets by the Task Force on Techno-Economic Issues (TFTEI 2015)¹¹. Although the version of the TFTEI spreadsheet that is used in the frame of the current work (cf. section 6.2) enables the user to consider part-load operation to some extent, the necessary data could not be obtained and therefore efficiency losses due to part-load operation were not considered.

¹¹ Formerly known as Expert Group on Techno Economic Issues (EGTEI), working under the United Nations Economic Commission for Europe (UN-ECE) on emission control costs amongst other.

On a more aggregated and technical level, the BAT REFerence (BREF) document for large combustion plants (European Commission 2006b) provides information on emission control measures, including typical ranges of abatement costs and emission levels (cf. section 2.4.6). Dedicated scientific methods supported the preparation of the BREF document, e.g. the reference installation approach (Geldermann and Rentz 2004, Nunge 2001). It seeks to define representative installations and associated abatement costs for use at a more aggregated level, e.g. to be used in integrated assessment models. Another work dealing with the economic assessment of BAT by Schultmann et al. (2001) provides a systematic approach for the assessment of investments and operating costs. It aims at providing a consistent methodological basis for the economic assessment of BAT from different sectors.

3.5.3 The Externalities of Energy project series

The most widespread methodology to estimate external costs related to atmospheric emissions in the energy sector has been developed in a joint US-European research initiative (European Commission 1995, Oak Ridge National Laboratory and Resources for the Future 1992). On the European side, this project became known as the Externalities of Energy (ExternE) project series (European Commission 1999a, 2005a). The impact pathway approach has been developed and promoted within ExternE as a bottom-up method to estimate environmental and health damage costs related to atmospheric emissions. Owing to advances in atmospheric modelling and in health impact assessment, site-specific assessments of impacts caused by energy conversion technologies were enabled, implemented in different tools (cf. section 3.5.7). The development of the health impact assessment methodology during more than 20 years is scrutinised in chapter 4. Besides assessing external costs of individual power plants in Europe, the ExternE methodology also provided the scientific basis for a more aggregated assessment of air quality scenarios at European level.

3.5.4 The GAINS model and its role in European policy-making

The Greenhouse Gas and Air Pollution Interactions and Synergies (GAINS) integrated assessment model has played an important role in air quality-related policy proposals at the European level. The model (or modelling framework, as it relies on inputs from other models), is used to determine cost-effective emission control strategies per sector and country that enable achieving predefined emission reduction targets, e.g. related to certain health objectives, at lowest costs (Amann 2012). As such it takes into account both the emission and exposure level, the latter being used to quantify health impacts.

For data input, GAINS relies on sector-level activity data from fundamental models, e.g. PRIMES for the energy sector (Capros et al. 2013), in certain cases complemented by national activity projections delivered by national governments. Besides, it incorporates emission factors, emission control measures and related costs. Its main output are optimised emission scenarios per country and sector, including reduction technique deployment, underlying private abatement costs as well as associated health benefits. For the latter, the impact pathway approach is used. In order to minimise emission control costs while assuring the respect of European-wide health targets, European Monitoring and Evaluation Programme (EMEP) source-receptor matrices¹² are used. These source-receptor matrices are derived from a 15% emission reduction per grid cell and are therefore suitable to define impacts at sectorial level, whereas individual emission sources are generally not well represented by this approach (cf. also discussion below). Source-receptor matrices with similar characteristics are underlying the EcoSenseWeb model (cf. sections 3.5.7 and 5.2), used amongst other to estimate sectoral reference damage costs for Germany (Schwermer et al. 2014, van der Kamp et al. forthcoming).

National research activities often aim at refining components used in integrated assessment models at larger scales. For the case of Germany, Breun et al. (2012) used simulation modelling to determine detailed future emissions from the industrial, residential, and traffic sectors. The underlying objective was to study the influence of policy instruments on these emission levels. In France, a concerted effort at national level was undertaken to develop an assessment methodology comparable to that of GAINS and including a CBA of sectorial emission control measures (cf. next paragraph).

3.5.5 Use of CBA for policy support and at power plant level

CBA has been extensively used for decision support in EU air quality policy-making, e.g. in the frame of the Clean Air For Europe (CAFE) Programme (Holland et al. 2005d) that led to the Thematic Strategy on Air Pollution (TSAP: European Commission 2005b) and also for the more recent EU Clean Air Policy Package, released in 2013 (European Commission 2014a, Holland 2014a). In both cases, the CBA builds upon the ExternE methodology and

¹² According to Tarrasón (2009), "source-receptor (SR) matrices give the change in various pollutant indicators in a receptor country, sub-area or grid square, resulting from a change in anthropogenic emissions from a specific emitter grid area, sub-region or country."

on the GAINS model. At French national level, CBA results (amongst other indicators) served to analyse the impact of future sectorial emission control measures (MEEM 2016). Moreover, CBA is routinely used in the USA to estimate the impact of air quality-oriented policy proposals, e.g. related to the Clean Air Act (U.S. EPA 2011a). While the scientific assessments underlying these policy proposals aim at defining future sectorial emission reductions at an aggregated level, they are not concerned with assessing individual industrial emission sources.

3.5.6 Social CBA in the presence of variable operation profiles

Most available models for the site-specific impact assessment consider local background conditions in terms of population densities, meteorology or emission inventories. The level of underlying detail, however, differs.

For instance, the damage cost assessments by the EEA (European Environment Agency 2011, 2014a) combine site-specific emission data of large industrial facilities with national average damage cost factors per tonne of emission, partly adjusted for sector particularities, e.g. emission heights. In this and other cases, e.g. the EcoSenseWeb model, EMEP source-receptor matrices are the basis for deriving region- or country-specific damage costs. Accordingly, the precision of the results depends on the quantity of emissions and the modelling resolution, as emissions are assumed to arise homogenously within a source cell. Moreover, the precision of damage estimates depends on the emissions' time profile throughout the year. Source-receptor matrices are usually developed assuming a decrease in emissions among all sources within a given source region, e.g. by 15% in the case of the EMEP matrices (Tarrasón 2009). Due to this static nature, they are not suitable for assessing point sources with variable operation profiles.

A dedicated air quality modelling, e.g. as proposed in the current work or in the case of the EVA model (Andersen et al. 2006), allows to consider space as well as time-variable factors in the damage cost assessment and therefore qualifies to assess impacts from sources with variable operation profiles. At the same time, such a dedicated modelling is costly in terms of time and computer resources and applications remain therefore often constrained to the aggregated level so far. The latter is also true for the case for the EVA model and its available results.

No studies could be identified that provide a methodology for social CBA of emission control measures at point emission sources with variable operation profiles. The current thesis intends to fill this gap by first developing a site-specific damage cost assessment approach, as described in section 6.1. In a next step, the damage cost assessment is integrated with an economic assessment within a framework for social cost-benefit analysis. The costing methodology takes into account the operation hours per year as well as the equipment lifetime. This way, variable operation profiles are covered to a certain degree, cf. section 6.2.

3.5.7 Models for health damage cost assessment related to atmospheric emissions from point sources

While noting that a number of guidance documents and spreadsheets exist that contain (national average or sector-specific) damage costs per tonne of pollutant emitted (DEFRA 2015, European Commission 2006a, Preiss et al. 2008, Schwermer 2007, Schwermer et al. 2014, van der Kamp et al. forthcoming), the focus here is on models that allow estimating damage costs of individual point emission sources.

Table 3.4 summarises the features of selected assessment tools. These have been classified according to the following criteria:

- ability to assess individual point emission sources or multiple emission sources;
- air quality modelling capabilities;
- health impact assessment and monetary valuation in terms of up-to-dateness of parameters and geographic context for which parameters are provided;
- spatial and temporal resolution, e.g. possibility to consider variable emission patterns, the size and the resolution of the modelled zone;
- input data needs;
- access rights, i.e. stating whether the tools are freely accessible.

EcoSenseWeb, Risk Poll, and the EVA model enable a site-specific assessment of air pollution impacts in Europe, partly including the local domain with a limited spatial extension (Andersen et al. 2006, Preiss and Klotz 2008, Spadaro 2004). Some models, such as EcoSenseWeb or RiskPoll, allow assessing damage costs of additional impact categories beyond human health, e.g. crops or materials, which are not further considered here.

The overview reveals that the available tools are not adequate for the purpose of the current thesis. Among the freely accessible tools, only EcoSenseWeb and Risk Poll qualify for an application in the European context, envisaged here. A major limitation of Risk Poll

is its simple air quality modelling that does not allow for a proper consideration of time and space dynamics. In addition, the input parameters for health assessment and monetary valuation included in Risk Poll are not up-to-date. Even though the feasibility of using EcoSenseWeb for a CBA of pollution abatement is demonstrated in this thesis (chapter 5), a major shortcoming regards the assessment of impacts from point sources with variable operation profiles. In the parameterised air quality model underlying EcoSenseWeb, atmospheric emissions are assumed to occur evenly throughout the year, contrary to the characteristics of a flexible operation mode. Moreover, the local assessment model in EcoSenseWeb only considers effects due to primary particles and uses a relatively coarse spatial modelling resolution, making it unsuitable to capture local effects in sufficient detail (cf. the weak point analysis in section 5.5.2). Consequently, an extended health damage cost assessment framework is developed and implemented in this thesis, briefly characterised at the bottom of Table 3.4 and more thoroughly described in section 6.1.

	Emission source	Air quality modelling	Health impact assessment	Monetary valuation	Spatial resolu- tion (km grid)	Temporal resolution	Input data needs	Access rights
EcoSenseWeb 1.3 (Preiss and Klotz 2008), cf. section 5.2	Point source	Parameterised Eulerian model (Eu- rope); Gaussian model (local), no specific model runs	EU average (year 2009); not directly modifiable	EU average (year 2009); not directly modifiable	Northern hemi- sphere (100), Europe (50), local (10)	1 year (cur- rent/future); constant emission rate	Low (cf. Table A.2)	License fee
<i>Risk Poll</i> 1.052, re- lease 2012 (Rabl et al. 2014, Spadaro 2004)	Point source	Integrated simpli- fied dispersion model, no specific model runs	EU average (year 2005); modifiable	EU average (year 2005); modifiable	Europe; other according to data availability	1 year; constant emission rate	Low / high (depend- ing on precision level)	Free
<i>EVA model</i> (Andersen et al. 2006, Brandt et al. 2013)	Point/ multiple sources	Integrated Eulerian model, specific model runs possible	Adapted EU average (year 2011); modifiable	Adapted EU average (year 2011); modifiable	Flexible; North- ern hemi- sphere (150), Europe (50), Denmark (16.7)	1 year; variable emission rate	Data integrated (high oth- erwise)	Re- stricted
BenMAP CE 1.0.8, release 2014 (U.S. EPA 2012a)	Point/ multiple sources	Not integrated; external monitoring or modelling data needed	US/China aver- age (year 2014); modifiable	US/China (year 2014); modifiable	Flexible (depending on air quality modelling)	1 year; depending on air quality model	Data integrated (high oth- erwise)	Free
Current thesis, cf. section 6.1	Point/ multiple sources	Not integrated; Eulerian model; optional local (Gaussian) 'plume- in-grid' modelling	EU average/ France (year 2015); modifiable	Adapted EU average (year 2015); modifiable	Flexible; three geographic domains with different reso- lutions	1 year; variable emission rate	Data integrated (high oth- erwise)	Re- stricted

 Table 3.4:
 Comparative overview on selected assessment tools to quantify human health-related damage costs from point emission sources (own compilation)

3.5.8 Overview: transferring a decision-making framework from the public policy to the business domain

Table 3.5 shows a proposal regarding the main methodological aspects to consider when transferring the social CBA methodology for the assessment of emission control measures from the public policy to the business domain. The implementation of these elements into a methodological framework is described throughout chapter 6. Further information related to uncertainty is provided in section 8.2. Final recommendations regarding the methodology transfer are provided in section 8.5.

	Public policy	<u>\</u>	Business
	EU/sector/country		Specific site
Objective of social	СВА		
	Impact assessments related to policy-making at Euro- pean or national scale		Verifying whether costs of emission control measures are disproportionate
Techno-economic c	ost assessment		
General approach	Reference technology ap- proach, annuity approach		Site-specific assessment, an- nuity approach
Environmental-eco	nomic benefit assessment		
General approach	Damage cost assessment: unspecific average cost fac- tors or parameterised mod- elling results, considering sectoral particularities such as height of release		Damage cost assessment: site-specific approach
Modelling resolution	Low to medium for aggre- gated assessments		High, reflecting the specific operating site and condi- tions

Table 3.5:Proposal of elements to consider when transferring a social CBA methodology for the assessment
of emission control measures from the public policy to the business domain (own compilation)

3.6 Summary of chapter 3

Externalities are the unpriced side effects of (trans)actions between economic agents, e.g. negative environmental impacts caused by industrial production processes. When expressed in monetary terms, they are called external costs or damage costs. To correct for this market failure, governments seek to internalise externalities through policy measures up to the point where marginal damages equal marginal abatement costs.

A prerequisite to support the internalisation process are techno-economic and environmental-economic assessment approaches for emission control measures. Social cost-benefit analysis combines both types of approaches, seeking to define the net present value of control measures by considering private and external effects. Private costs of emission control measures are most appropriately estimated using the annuity approach. Health damage costs related to atmospheric emissions are typically estimated using the impact pathway approach. The impact pathway approach has been implemented through various tools that account, to some degree, for the specific site of an emission source, however with limited temporal and spatial resolution. This is particularly critical for assessment at power plant level with variable operation patterns. New methodological advancements are therefore proposed in this thesis. By use of a highly time- and spaceresolved modelling, the limitations of currently available assessment tools shall be overcome (cf. chapter 6). Specific case studies are used to conclude on the added value of such a dedicated modelling approach compared to alternative approaches (cf. chapter 7). Feeding into a social cost-benefit analysis framework for the assessment of emission control measures at site level, the methodology enables public administrations and industry alike to verify that scarce resources are spent efficiently to enhance overall societal welfare.

4 Health damage cost assessment: in-depth methodology review and adaptation

This chapter features an in-depth health damage cost methodology review and adaptation, covering a timespan from the beginnings of the ExternE project series to the EU's recently released 2013 Clean Air Policy Package. It serves to identify the main methodological influencing factors on health damage costs and as such it supports informed choices of assessment parameters (cf. section 6.1) and of potential levers for sensitivity analyses (cf. section 7.3.3). The contents are largely based on van der Kamp and Bachmann (2015).

4.1 Context and objectives

In the context of preparing the EU's 2013 Clean Air Policy Package (European Commission 2014a), the World Health Organization carried out a meta-analysis that led to the publication of a new set of recommended concentration-response functions (WHO 2013a) for use in cost-benefit analysis (CBA) of air pollution policies. This can be seen in a tradition of changing modelling components and assessment parameters throughout different EU projects over time, resulting in differences in published damage costs (Droste-Franke 2005, European Commission 2003, Krewitt et al. 1999, Krewitt et al. 1997, Rabl et al. 2011, Spadaro and Rabl 2002). A few studies analysed the link between methodological assumptions and quantified damage costs, covering the years 1995 to 2005 at most (Krewitt 2002, Krewitt and Schlomann 2006). Given the manifold scientific developments over the past 20 years and the continued relevance of air-pollution-induced health costs for policymaking both in and outside Europe, the aim of this chapter is to deepen these analyses on the impacts of methodological developments on the magnitude of quantified damage costs and extend these to the present. One important contribution to a better scientific understanding is to disentangle the influencing factors that are otherwise hidden in aggregated damage cost estimates. This is also of relevance with regard to identifying the most influential parameters to be used in sensitivity analyses of damage costs.

To illustrate developments in terms of exposure modelling, risk assessment and monetary valuation, a coal-fired power plant unit located in Western Europe is used as a case. Since quantitative results exist only for projects up to the year 2009, a methodological adaptation is carried out in order to also produce damage cost estimates based on newest European recommendations. This analysis focuses on the variability in damage cost estimates arising from heterogeneous methodological choices. Methodological influences are assessed independently of operational or geographical variations, addressed in chapter 5.

4.2 Approach

Four implementations of the impact pathway approach (termed IPA implementations henceforth) and a sensitivity scenario for the most recent implementation are compared: ExternE1998, New Elements for the Assessment of External Costs from Energy Technologies (NewExt2004), New Energy Externalities Developments for Sustainability (NEEDS2009) and Year2013/2013*.

These are applied to an exemplary emission point source for which damage costs had already been quantified before, i.e. a 600 MW_{electric} pulverised coal combustion unit, located in Western France. To ensure comparability, technical emission source specifications were kept constant. For ExternE1998 and NewExt2004, published damage costs were used (European Commission 1999a, b, 2004). The NEEDS2009 results were estimated using the tool EcoSenseWeb (cf. section 5.2) and subsequently updated to obtain damage costs for Year2013 and Year2013* (cf. section 4.3).

While the data provided in section 4.3 helps to explain most of the observed changes in damage costs, disentangling the influence of single parameters is not straightforward. To quantitatively estimate their influence on damage costs, the following elements are successively analysed.

First and regarding population exposure (step 1), the question is: which damage costs result if the population exposure modelling of the most recent tool is used instead of the original exposure modelling? Aggregated population exposure figures per pollutant from the NEEDS2009 assessment were derived and then respectively combined with damage costs per pollution increment from the original 1998 and 2004 IPA implementations (cf. section 4.3.4).

In addition to an updated exposure modelling, the next question concerns particle toxicity (step 2): what is the impact of using the current recommendation of equal particle toxicity? To this end, results from step 1 were adapted in terms of particle toxicity. Factors applicable to secondary PM were defined by using the toxicity coefficients as stated in section 4.3.3 and assuming a mass ratio of nitrates and sulphates of 2:1, derived from dedicated EcoSenseWeb calculations.

Among all health endpoints, long-term mortality accounts for the largest share in total quantified health-related damage costs (cf. section 4.4.1). The influence of updating the related impact assessment parameters (risk function as well as monetary valuation) according to NEEDS2009 is assessed on top of the outcome of step 2 (updated exposure modelling and updated particle toxicity).

4.3 Models and data used

4.3.1 The tool EcoSenseWeb, its case study application and updates

Part of the damage costs were calculated with the web-based software tool EcoSenseWeb 1.3 (cf. section 5.2), developed in the NEEDS project (Preiss and Klotz 2008). Choosing this model allows comparison with the ExternE1998 and NewExt2004 results, calculated with its predecessor, the EcoSense desktop tool (IER 2004). EcoSenseWeb is applicable to stationary point emission sources in Europe. Data needed for a site-dependent externality assessment are provided, i.e. receptor data, impact function slopes, and monetary values. Only health impacts caused by classical air pollutants are considered here.

As a novel feature, EcoSenseWeb results are manually adapted in terms of impact functions and monetary values to reflect the latest expert recommendations at European level. Two parameter sets were defined, representing a base case and a sensitivity case with differing mortality valuation, denoted by Year2013 and Year2013*, respectively (cf. sections 4.3.4 and 4.3.5).

4.3.2 Exposure modelling

Two types of regional (European-wide) air quality models were used (Table 4.1), i.e. a Lagrangian model and a (parameterised version of a) Eulerian model that mainly differ in the mathematical treatment of air parcels and associated chemical interactions

(U.S. EPA 2009). Moreover, modelling resolutions as well as emission, meteorological and population data varied. To ensure comparability, only regional, i.e. European-scale, air quality models are considered. For availability reasons, the Year2013 assessments do not include the updated EMEP (European Monitoring and Evaluation Programme) source-receptor matrices (Schulz et al. 2013), having been used to assess the EU Clean Air Policy Package in 2013.

	ExternE1998	NewExt2004	NEEDS2009, Year2013/2013*
Air quality model	Windrose Trajectory (Lagrangian) Model	Windrose Trajectory (Lagrangian) Model	EMEP/MSC-West Eulerian dispersion model, parameterised
Emission inven- tory	1990	1998	2010 (projected in 2006)
Meteorology	1990	1998	Average of 1996, 1997, 1998 and 2000
Modelling resolu- tion	Eurogrid (100km x 100km)	EMEP50 (50km x 50km)	EMEP50 (50km x 50km)
Population data	EUROSTAT REGIO	EUROSTAT REGIO	SEDAC 2007 and NEEDS

Table 4.1:Air quality and exposure modelling characteristics for classical air pollutants except ozone for the
considered IPA implementations (European Commission 1999b, 2004b, Preiss and Klotz 2008a)

Beyond the changes displayed in Table 4.1, ozone damages were modelled by a generic factor in 1998 and by a site-dependent approach since 2004 (European Commission 2004). The ozone exposure metric was changed from ExternE1998 and NewExt2004, relying on 6-hour average values, to NEEDS2009, being based on the SOMO35 (Sum Of (maximum daily 8h) Means Over 35 parts per billion, ppb) metric, disregarding ozone effects below 35 ppb (European Commission 2005a).

4.3.3 Particle toxicity

Assumptions about the toxicity of different PM compounds relative to primary particles also varied over time: In ExternE1998, sulphates were estimated to be 1.67 times more toxic than nitrates and primary PM_{10} particles because of their typically smaller size and hence larger damage potential (European Commission 1999a, 2005a). In NewExt2004,

due to lacking evidence for its effects, nitrates were assumed to be half as toxic as sulphates and primary particles (European Commission 2004). In NEEDS2009 and Year2013/Year2013*, all types of particles are assumed to be equally toxic (WHO 2007, 2013b), although the topic remains debated (cf. section 8.4).

4.3.4 Impact functions

The four IPA implementations use health-related impact functions that are pollutant (i.e. PM₁₀, PM_{2.5} and ozone), risk group and age group-specific (Table 4.2). Driven by new or re-evaluated scientific evidence, changes concerned the impact functions, the affected population fraction or the particle size to which impacts are associated. For comparability reasons, effects that could not be quantified with EcoSenseWeb were disregarded, i.e. SO₂-related endpoints in ExternE1998 and NewExt2004, and NO₂-related endpoints in Year2013/Year2013*. While the latter risks underestimating damage costs, potential double counting with PM impacts and causality issues are debated (WHO 2013b). Since NEEDS2009 and even more so in the Year2013 implementation, endpoints are more frequently related to PM_{2.5} than to PM₁₀. Some endpoints were dropped for Year2013, e.g. bronchodilator usage, and others have been newly introduced, e.g. asthma symptom days among children due to PM₁₀.

Drawing on recommendations by the WHO (2013a), only impact functions with a sufficiently high level of confidence were considered in the 2013 implementations, i.e. categories A* and B* according to the HRAPIE (Health Risks of Air Pollution In Europe) project. For mortality assessment and following HRAPIE, only natural causes are considered, i.e. no deaths due to accidents or other external causes. Two assessment approaches co-exist. While the YOLL (Years Of Life Lost) approach, based on life table calculations (Miller et al. 2011), was generally advocated in the ExternE project series (European Commission 2005a), the U.S. EPA estimates cases of deaths (U.S. EPA 2011a). To account for differing expert opinions, both approaches were used in parallel in the CAFE (Clean Air For Europe) Programme (Hurley et al. 2005) and the impact assessment underlying the EU Clean Air Policy Package (Holland 2014a). For the IPA implementations studied, mortality risks in adults are expressed in YOLL, while for infants cases of death are estimated. The risk coefficient for cases of premature death due to long-term PM_{2.5} exposure as provided by the HRAPIE project needs to be converted into EU average YOLL, consistent with earlier projects. For Year2013 and Year2013*, following an expert recommendation (Hurley et al. 2005), the YOLL-based impact function from NEEDS2009 is scaled linearly using the quotient of the relative risk provided by the HRAPIE ($0.062/10 \ (\mu g \ PM_{2.5}/m^3)$ and the NEEDS project (0.06/10 (µg PM_{2.5}/m³)).

Table 4.2: Health-related impact function slopes (s_{CR}) and corresponding risk/age group, expressed as 'effect increase per (10 µg/m³ x person x year)' and grouped by pollutant (European Commission 2004, Torfs et al. 2007, WHO 2013a)

Health endpoint		ExternE	NewExt	NEEDS	Year2013
Risk group, age group	Unit	1998	2004	2009	/2013*
Ozone ^a					
All-cause, natural mortality (acute) ^b	YOLL				
All, all ages		0.00004	0.00004	0.00002	0.00003
Asthma attack	days				
Asthmatics, all ages		0.4930	0.4930		
Bronchodilator usage (summertime)	cases				
Asthmatics, 20+				0.7300	
Cardiovascular hospital admission (excl. stroke)	cases				
All, 65+					0.0005
Cough day	days				
All, children 5-14				0.9300	
Lower respiratory symptoms (excl. cough)	days				
All, children 5-14				0.1600	
Minor restricted activity day (MRAD) ^c	days				
All, 18-64				0.1154	
All, all ages					0.1201
All, adults 18+		0.0976	0.0976		
Respiratory hospital admission	cases				
All, 65+				0.0001	0.0001
All, all ages		0.00004	0.00004		
Symptom day	days				
All, all ages		0.3307	0.3307		
PM ₁₀ (primary and secondary)					
All-cause infant mortality	cases				
All, infants 0-1				0.0001	0.0001

Health endpoint <i>Risk group, age group</i>	Unit	ExternE 1998	NewExt 2004	NEEDS 2009	Year2013 /2013*
All-cause, natural mortality	YOLL				
All, 30+		0.0072	0.0039		
All-cause, natural mortality (acute) ^b	YOLL				
All, all ages		0.00003	0.00003		
Asthma symptom day	days				
Asthmatic children, children 5-19					1.7374
Bronchitis prevalence					
All, children 6-12	cases				0.0149
Bronchodilator usage	cases				
Asthmatics, 20+				0.9125	
Asthmatics, adults 18+		1.6290	1.6290		
Asthmatics, children 0-18		0.7790	0.7790		
PEACE criteria, children 5-14				0.1825	
Cardiac hospital admission	cases				
All, all ages				0.00004	
Cerebrovascular hospital admission	cases				
All, all ages		0.0001	0.0001		
Chronic bronchitis case	cases				
All, 18+					0.0005
All, 27+				0.0003	
All, adults 18+		0.0005	0.0005		
Chronic bronchitis episode	cases				
All, children 0-18		0.0161			
Chronic cough	cases				
All, children 0-18		0.0207	0.0207		
Congestive heart failure	cases				
All, 65+		0.0002	0.0002		

4 Health damage cost assessment: in-depth methodology review and adaptation

Health endpoint	Unit	ExternE	NewExt	NEEDS	Year2013
Risk group, age group	Ome	1998	2004	2009	/2013*
Cough day	days				
Asthmatics, adults 18+		1.6760	3.3520		
Asthmatics, children 0-18		1.3350	2.6700		
Lower respiratory symptoms (wheeze)	days				
Asthmatics, adults 18+		0.6060	0.6060		
Asthmatics, children 0-18		1.0290	1.0290		
Lower respiratory symptoms (incl. cough)	days				
All, children 5-14				1.8600	
Respiratory symptoms, adults 15+				1.3000	
Respiratory hospital admission	cases				
All, all ages		0.00002	0.00002	0.00007	
Restricted activity day (RAD); (RAD — net) ^d	days	T			
All, adults 18+		0.2499; (0.2472)	0.2499; (0.2472)		
PM _{2.5} (primary and secondary)					
All-cause, natural mortality	YOLL				
All, 30+					0.0107
All, all ages				0.0065	
Cardiovascular hospital admission	cases				
All, all ages					0.0002
Minor restricted activity day (MRAD)	days				
All, 18-64				0.5772	
Respiratory hospital admission	cases				
All, all ages					0.0003
Restricted activity day (RAD); (RAD – net) ^d	days				
All, all ages				0.6061; (0.0957)	0.8930; (0.5853)

Health endpoint Risk group, age group	Unit	ExternE 1998	NewExt 2004	NEEDS 2009	Year2013 /2013*
Work loss day (WLD)	days				
All, 15-64				0.2070	
All, 20-65					0.6385

^a Seasonal 6 hour-averages for 1998 and 2004; SOMO35 for 2009 and 2013

^b Assuming 0.75 YOLL per case for ExternE1998, NewExt2004 and NEEDS2009 (Preiss and Klotz 2008) and 1 YOLL per case for Year2013/2013*(Holland 2014b)

^c MRAD should be calculated net of asthma attacks due to ozone (European Commission 2004)

^d To avoid double counting, net effects are obtained by correcting for work loss days, hospital admission days, minor restricted activity days and symptom days due to ozone or PM (Preiss and Klotz 2008, WHO 2013a)

4.3.5 Monetary valuation

Monetary values of health endpoints varied over the years (Table 4.3). Depending on the mortality risk metric, different monetary values are used: YOLL are valued by a so-called Value Of a Life Year (VOLY; previously also abbreviated by VLYL, Value of a Life Year Lost), while cases of death are valued by a Value of a Statistical Life (VSL,) cf. OECD (2012). For the Year2013 implementation, the monetary parameters from the CBA of the EU Clean Air Policy Package were adopted (Holland 2014a). This implies that newer evidence for the adult mortality-related VOLY from the NEEDS2009 project is disregarded. Instead, the parameters recommended in the CAFE Programme in 2005, based on NewExt2004, are used (Holland et al. 2005a). To account for differing expert judgments on this crucial parameter, alternative values for the Year2013 (Holland 2014a) and Year2013* (Desaigues et al. 2011) implementation were used, the latter equalling the NEEDS2009 valuation. Further changes of the Year2013/2013* implementations towards the NEEDS2009 implementation concern:

- All-cause infant mortality: the median VSL based on CAFE implies a decrease of 52% relative to NEEDS;
- Chronic bronchitis cases: an updated value from the Health and Environment Integrated Methodology and Toolbox for Scenario Assessment (HEIMTSA) project (Hunt et al. 2013) is used, amounting only to about 25% of the NEEDS valuation;
- Restricted activity days: the CAFE valuation is used, being 36% smaller than in NEEDS;

• Work loss days: these are valued according to data from the Confederation of British Industry, reducing the NEEDS value by 60%.

For ExternE1998, monetary parameters were converted from ECU₁₉₉₅ into ξ_{2000} using a factor of 1:1 and an average annual inflation rate of 1.5% for the relevant period. For the Year2013/2013* implementations, values were converted from ξ_{2005} to ξ_{2000} using an average annual inflation rate of 2.1% (eurostat 2013).

Health endpoint	ExternE 1998	NewExt 2004	NEEDS 2009	Year2013/ 2013*
all-cause infant mortality (VSL)	-	-	3 000 000	1 442 086 ^b
all-cause natural mortality (VOLY, acute exposure)	166 979	75 000	60 000ª	52 005 ^b / 60 000ª
all-cause natural mortality (VOLY, long-term exposure)	90 847	50 000	40 000ª	52 005 ^b / 40 000 ^a
asthma attack	81	75	-	-
asthma symptom day	-	-	-	38
bronchitis prevalence	242	-	-	530
bronchodilator usage	40	40	1	-
cardiac hospital admission	-	-	2 000	-
cardiovascular hospital admission	-	-	-	2001
cerebrovascular hospital admission	8 478	16 730	-	-
chronic bronchitis case	113 115	169 330	200 000	48 310
chronic cough	242	240	-	-
congestive heart failure	8 478	3 260	-	-
cough day	8	45	-	-
lower respiratory symptoms (wheeze)	9	8	38	-
lower respiratory symptoms	-	-	38	-
minor restricted activity day (MRAD)	48	45	38	38

Table 4.3: Monetary values (m) expressed in €₂₀₀₀ for health endpoints (European Commission 1999a, 2004, Holland 2014a, Preiss and Klotz 2008)

Health endpoint	ExternE 1998	NewExt 2004	NEEDS 2009	Year2013/ 2013*
respiratory hospital admission	8 478	4 320	2 000	2 001
(net) restricted activity day (RAD)	81	110	130	83
symptom day	48	45	-	-
work loss day	-	-	295	117

^a Expressed in €₂₀₀₀ (Preiss and Klotz 2008); by contrast, the original study (Desaigues et al. 2011, Desaigues et al. 2007) presumably uses €₂₀₀₅, which would reduce the indicated values

^b median parameter estimate

4.4 Results

4.4.1 Development of health damage costs over time

The coal-fired power plant's health-related damage costs vary between 1.77 (NewExt2004) and 5.21 (ExternE1998) €-cent₂₀₀₀ per kWh of electricity produced (Figure 4.1). Damage costs decreased by 66% from 1998 to 2004, mainly due to reduced mortality impacts, and increased by 57% from 2004 to 2009. Depending on assumptions about mortality impact valuation, estimated damage costs either increase by 15% (Year2013) or decrease by 5% (Year2013*) from 2009 to 2013.

For data availability reasons, only the self-calculated results can be further decomposed. With 66%/77%/70% (NEEDS2009/Year2013/Year2013*) of all quantified damage costs, mortality due to long-term PM_{2.5} exposure is by far the most important single health endpoint, consistent with previous observations (Holland et al. 2005b, Krewitt 2002, Spadaro and Rabl 2008). This is followed by work days lost (10%/7%/10%), net restricted activity days (7%/11%/14%), and chronic bronchitis cases (10%/4%/5%).



Figure 4.1: Marginal damage costs of atmospheric emissions from a coal-fired power plant unit, estimated with different IPA implementations

4.4.2 Quantitative analysis on the influence of single elements in the assessment chain

In the following, the influence of individual elements of the assessment chain is studied (Figure 4.2), referring to section 3.4.1, equation 3.3.

Using NEEDS2009 exposure modelling instead of the original exposure modelling reduces the quantified health-related damage costs by 21% and 9.4% for ExternE1998 and New-Ext2004, respectively. This implies that human exposure to classical air pollutants was estimated to be higher in 1998 than in 2004 and 2009 for the studied case. Additionally assuming equal instead of differential particle toxicity reduces damage costs by another 17% for ExternE1998 and increases them by 43% for NewExt2004, underlining the importance of particle toxicity in equation 3.3. When finally using the NEEDS2009 impact function and monetary valuation for long-term exposure mortality, damage costs are reduced by 29% for ExternE1998 and increased by 15% for NewExt2004, bringing both values considerably close to the NEEDS2009 result.


Figure 4.2: Influence of exposure modelling, particle toxicity and long-term mortality assessment on healthrelated damage costs of atmospheric emissions from a coal-fired power plant unit

This comparison is limited in that the NEEDS2009-based exposure modelling includes neither ozone-related impacts below the 35 ppb concentration threshold nor direct impacts from SO₂. Both types of impacts are included in the original damage costs from ExternE1998 and NewExt2004, explaining part of the residual differences in Figure 4.2.

Given its importance, the development of long-term exposure mortality impact assessment deserves more scrutiny. The original impact function used in 1998 was directly derived from a US study (European Commission 2004). Accounting for differences between Europe and the USA slightly decreased the impact function in 2004. Simultaneously, the associated monetary value was substantially reduced due to a more refined valuation method. From 2004 to 2009, the monetary value decreased further, reflecting updated evidence (Desaigues et al. 2011). This change, however, did not reduce damage costs because of the increased toxicity coefficient for secondary particles. From NEEDS2009 to Year2013, the monetary valuation of long-term exposure mortality impacts increased, while the risk slope remained almost constant. Effectively, an increase of damage costs per concentration increment resulted. In the sensitivity case (from NEEDS2009 to Year2013*), the monetary value remained constant while the risk slope slightly increased, thereby rising mortality-related damage costs.

4.5 Summary of chapter 4

Energy-related health damage cost estimates of four impact pathway approach implementations are compared, corresponding to different stages of the ExternE project series and follow-up activities. Referring to recent European recommendations, two variants of an up to date assessment are proposed, differing only in mortality impact valuation. Disentangling influencing factors that are otherwise hidden in estimates of human healthrelated damage costs is a central achievement of this chapter, realised through literature research and guantitative analyses. Case study results reveal that exposure modelling accounts for differences between damage costs of up to 21%. Among health endpoints, mortality impacts due to long-term PM_{2.5} exposure are by far the most influential single endpoint (e.g. 76% of the health-related Year2013 damage costs). Particle toxicity assumptions remarkably influence damage costs as well, affecting all particulate matter (PM)-related endpoints simultaneously. PM-related mortality impacts should thus be preferentially studied in sensitivity analyses of health damage costs (cf. section 7.2.2). As a limitation, the quantitative findings cannot be generalised because exposure modelling results and corresponding damage costs depend on the emission pattern and the location of the source. Due to data and model accessibility constraints, the influence of individual exposure modelling components (e.g. spatial modelling resolution) could not be further analysed. However, the analysis of modelling features in section 7.1 provides further insight on this topic.

5 Exemplary application of social cost-benefit analysis using an existing model and weak point analysis

In this chapter, a social cost-benefit analysis (CBA) of emission control at an exemplary coal-fired power plant is carried out. Using and adapting existing assessment models, the feasibility of social CBA at plant level is demonstrated, whilst also allowing to assess the influence of geographic and further settings on damage costs through a sensitivity analysis. The contents are largely based on Bachmann and van der Kamp (2014).

5.1 Introduction

As a result of several European research initiatives in the context of energy externalities, the EcoSenseWeb model enables quantifying environmental impacts in economic terms with a focus on emission reductions related to classical air pollutants. Using the EcoSenseWeb model, a social CBA of an exemplary emission control measure is carried out here, testing the criterion of "disproportionality" in the sense of the EU industrial emissions directive (IED) (cf. section 2.4.6). As was shown for instance by Krewitt et al. (2001), Bachmann (2006), and more recently by Czarnowska and Frangopoulos (2012), the magnitude of external effects varies due to environmental settings. Accordingly, the focus is put on the first criterion for applying for derogations from BAT, i.e. the geographical location or the local environmental conditions. Without loss of generality, the case is made for a typical coal-fired power plant at different Western European locations. Nevertheless, the general approach is applicable also to other kinds of industrial installations as well as to other policy contexts.

5.2 Model: damage cost assessment using EcoSenseWeb

For classical air pollutants, damage costs are estimated on the basis of the impact pathway approach (cf. section 3.4.1), as implemented in the EcoSenseWeb assessment model.

EcoSenseWeb 1.3 (Preiss and Klotz 2008) is used to calculate the environmental benefits resulting from the installation of emission control equipment at an exemplary fossil-fired power plant (cf. section 5.4). The model allows assessing impacts resulting from various emissions into air, comprising classical air pollutants (such as NO_x, SO₂, and primary particles), trace pollutants (such as heavy metals, arsenic, and dioxins) and greenhouse gases. Radionuclide releases and impacts on biodiversity due to land use changes can also be assessed but are disregarded here given the scope of the IED (articles 2 (1), 3 (1a) and 3 (2)). Data and methods necessary to quantify impacts according to the impact pathway approach for releases in Europe are provided. Most of the data is based on results from the EU research project NEEDS (cf. section 4.3 for more details on the models and parameters used). Merely source-specific data need to be entered by the user. Table A.2 (Appendix) gives an overview on all kinds of data that a user can potentially enter. Fewer data may be specified, depending on the scope of the damage costs to be assessed (e.g. regional and/or local; human health, climate change and/or biodiversity). Below, the approaches implemented in EcoSenseWeb for the guantification of impacts from different kinds of air pollutant groups are briefly presented, mainly based on the information provided by Preiss and Klotz (2008).

5.2.1 Classical air pollutants

For classical air pollutants, following the impact pathway approach, impacts on different receptors are considered, i.e. human health, crops, building materials, and biodiversity.

Different dispersion models are available in EcoSenseWeb to assess classical air pollutants in a nested way, covering local, regional (European) and Northern hemispheric impacts. At the local scale, i.e. 100 x 100 km around a point source, a Gaussian plume model calculates dispersion of primary PM only. At the European and Northern hemispheric scale, dispersion as well as chemical conversion is assessed with help of source-receptor matrices derived from the EMEP Eulerian model, involving non-linear relationships. The European modelling domain covers the EU28 plus 11 non EU countries as well as some sea regions. Some larger countries (e.g. France and Germany) are subdivided into sub-regions to allow for a more precise assessment of emissions from these countries.

Different impacts are covered such as mortality and morbidity for human health due to exposure via inhalation, crop yield losses due to airborne pollutants, acidification of soils or fertilising effects, damage to materials due to SO₂ exposure or acid rain, and potential disappearance of plant target species due to acidification and eutrophication.

Monetary values have been obtained through contingent valuation studies (eliciting the Value of a Life Year, VOLY, and monetary values of several respiratory diseases) as well as by relying on market data for crops, treatment costs (e.g. bronchodilator uses), wages (work loss days), and restoration costs (impacts on biodiversity and on building materials). Some impacts occur in the long run. No explicit information on the discount rates used is given in the consulted literature.

In general, receptor data, impact functions and monetary values were determined in different years. Damage costs in EcoSenseWeb are expressed in a common base unit (i.e. ϵ_{2000}). Impact functions and monetary values (cf. the NEEDS2009 parameters in Table 4.2 and Table 4.3) stem from different regions, i.e. from Europe and North America.

5.2.2 Trace pollutants

Trace pollutants are assessed to lead to increased human health risks only. Neither dispersion, nor exposure nor impact models for trace elements are implemented directly in EcoSenseWeb. Rather, external unit costs pre-calculated with different models in previous studies are used for trace pollutants. These rely on population data and monetary values that vary from those used in the assessment of classical air pollutants.

5.2.3 Greenhouse gases

Greenhouse gas (GHG) emissions are valued either by a default value, a user-specified constant value, a marginal abatement cost (MAC) or a marginal damage cost (MDC) approach.

When following the MAC approach, monetary values for GHGs are available for emissions occurring between 2000 and 2050. These were derived from a meta-analysis of different models and scenarios by Kuik (2007). The values are valid for a CO_2 -eq. concentration target of between 535 and 710 ppm or a CO_2 concentration target of between 440 and 570 ppm. The values in ξ_{2005}/t CO₂ are: 19 until 2020, 23 in 2025 and 61 in 2050 with linear interpolation between these reference years.

EcoSenseWeb allows assessing GHGs up to the year 2100 when using the other approaches, i.e. a default value of $19 \in 2000/t \text{ CO}_2$ (derived based on MAC principles, cf. European Commission (2005a)), a user-specified constant value (of whatever kind), or the

MDC approach. The MDC approach relies on the FUND model¹³, developed partly in research projects funded by the European Commission with several publications in the peer-reviewed literature. The MDC values implemented in EcoSenseWeb strongly depend on whether or not the optional equity weighting is used. Equity weighting accounts for the fact that a damage of $1 \in$ is less severe in a highly developed country than in a developing country. Without equity weighting, the values are below $10 \notin_{2000}/t$ CO₂ while they are two to five times higher than the default value of $19 \notin_{2000}/t$ CO₂ when considering equity weighting. As a result, the user's choice concerning monetisation of GHG emissions has a pronounced impact on the finally obtained results.

5.2.4 Settings for calculations

Beyond the facility-specific input parameters (Table A.2), EcoSenseWeb offers a few assessment options (cf. Table 5.1). These concern whether or not air quality is assessed at different scales; for regional (i.e. Europe-wide) air quality, the user can choose between two background emission scenarios and between two meteorological conditions; three variants of human health impact functions are provided; impacts on biodiversity are optionally assessed; GHGs and micro pollutants can be included or excluded; for GHG emissions, different monetary values can be specified. For the calculations presented in section 5.4 default settings in EcoSenseWeb were used (Table 5.1). For the sensitivity results, human health impact functions as well as the assumptions regarding air quality modelling have been varied (cf. section 5.4.4).

¹³ Cf. <u>http://www.fund-model.org/publications,</u> last accessed: 2017-05-18

Setting concerning	Choice	Remarks
Classical air pollutants – dispersio	on modelling	
Local air quality model	Included	-
Regional air quality model	Included	-
Background emission scenario	2010 ^d /2020 ^s	Influencing the chemistry of the non-local atmospheric dispersion modelling (non-linearities), antici- pated emissions
Meteorological data	Default year ^d / Future year ^s	Influencing the regional atmos- pheric dispersion modelling. The de- fault setting is an average of the years 1996, 1997, 1998 and 2000; alternatively, a future setting can be chosen, corresponding to the year 2003
Hemispheric air quality model	Not included	Given the European scope of the IED, hemispheric results are disre- garded
Classical air pollutants – impact a	ssessment	
Health risk functions of classical air pollutants	SIA_E_PPM ^d / SIA_D_PPM_Core ^s	Default risk functions, assuming equal toxicities of different primary and secondary particles; alterna- tively assuming differing toxicities of primary and secondary particles
Biodiversity losses due to acidifi- cation and eutrophication	Included	-
Assessment of other pollutants		
Micro pollutants	Not included	For data availability reasons
GHG valuation	Default value	19 € ₂₀₀₀ /tonne CO₂-eq. (for all op- tions, cf. section 5.2.3)

Table 5.1:Settings of EcoSenseWeb as used in the analysis of the current chapter;
based on Preiss and Klotz (2008)

^d default assessment; ^s sensitivity analysis

5.3 Analysis of the influence of the geographic location on the damage costs

In the following, a specific investment into an emission control technology is presented that shall be compared to the related environmental benefits, depending on the environmental setting as well as on sensitivities towards specific assumptions. Through a CBA, the overall societal efficiency of the investment at a specific site is tested. While private costs are taken from a publicly available dataset, the EcoSenseWeb model is used to define environmental benefits resulting from the investment.

5.3.1 Case study: retrofitting a coal-fired power plant with a DeNOx system

Installing and operating a selective catalytic reduction (SCR) for reducing NO_x emissions (DeNOx) at the investigated coal-fired power plant changes its technical characteristics (Table 5.2).

	Without SCR	With SCR
General data		
Plant type	Existing plant, pulver- ised hard coal, steam turbine	Existing plant, pulver- ised hard coal, steam turbine
Thermal input [MW]	1714	1714
Net Electricity sent out [MW]	600	597
Full load hours per year [h/a]	4500	4523
Annual Net electricity generation [GWh/a]	2700	2700
Abatement techniques		
Flue Gas Desulphurisation	installed	Installed
Electrostatic Precipitator	installed	Installed

 Table 5.2:
 Data used as input into EcoSenseWeb to define the damage costs of a coal-fired power plant unit without and with a selective catalytic reduction (SCR) and primary abatement NOx measures installed; adapted from European Commission (2004)

	Without SCR	With SCR		
Flue-gas denitrification	n/a	Installed (primary + secondary)		
Data relevant for atmospheric modelling				
Stack height [m]	220	220		
Stack diameter [m]	5.1	5.1		
Flue gas volume stream (full load) [Nm³/h]	1960000	1960000		
Flue gas temperature [K]	363	363		
Emissions - classical air pollutants				
SO ₂ [mg/Nm ³]	252	252		
NO _x [mg/Nm³]	820	200		
Particulates (PM ₁₀) [mg/Nm ³]	2.0	2.0		
Particulates (PM _{2.5}) [mg/Nm ³] ^a	1.7	1.7		
NH₃ [mg/Nm³]	0.0	0.0		
NMVOC [mg/Nm³]	4.0	4.0		
Emissions – GHG				
CO ₂ [tons/a]	2187000	2197935		
CH₄ [tons/a]	27.0	27.1		
N ₂ O [tons/a]	59.4	59.7		

^a As the share of PM_{2.5} in PM₁₀ is not known, a value of 86 % is used by default for hard coal-fired power plants operated at full load (Dreiseidler et al. 2000)

The overall efficiency of the plant is reduced by 0.5% due to the SCR's electricity consumption (European Commission 2006b). To produce the same net amount of electricity, the power plant with SCR is assumed to operate some additional hours per year, increasing overall air emissions. The SCR is characterised as follows:

 The SCR, in combination with primary measures, allows to respect the minimum regulatory limit set by the IED (Annex V) for existing coal-fired combustion plants (>300 MW_{thermal}) of 200 mg NOx per Nm³ flue gas emitted;

- The flue gas first passes the SCR followed by an electrostatic precipitator for PM reduction and then by a wet flue gas desulphurisation device (high dust configuration), not causing additional NH3 emissions since ammonia is assumed to be absorbed in the electrostatic precipitator used for PM abatement (VGB and EURELECTRIC 2010);
- No additional oxidation module exists at the SCR for reducing non-methane volatile organic compound (NMVOC) emissions;
- Catalyst degradation over time is not considered.

When calculating damage costs, the same power plant unit is assumed to be situated either in Brussels (Belgium), Cartagena (Spain) or Helsinki (Finland, cf. Table 5.3), varying in terms of population density, background emissions and background meteorology. As a result, the parameter values for the two scenarios of Table 5.2 are inserted into EcoSenseWeb three times, i.e. for each of the sites displayed in Table 5.3.

Location	Brussels	Cartagena	Helsinki
Latitude (°N)	51.12	37.59	60.13
Longitude (°E)	4.30	-0.10	24.69

Table 5.3: Geographic coordinates of the three investigated power plant locations

5.3.2 Scope and discounting

Given that CBA theory has been presented in previous chapters of this work (in particular section 3.2), only the most relevant assumptions and data used are presented here.

The decision criterion for the CBA is the net present value. As regards the issue of "who counts" and spatial scope, the individuals considered in the current analysis are those living in the EU as well as some neighbouring countries (i.e. countries for which results are available in the EcoSenseWeb regional assessment). A scope larger than the EU is considered appropriate for two reasons. First, the IED is relevant for the European Economic Area. Second, impacts outside the EU are also of concern to EU policy makers (European Commission 2009). The time horizon for the investment is 20 years according

to an assumed remaining power plant lifetime during which the emissions and technical parameters are assumed to be constant.

For determining private costs, the two-stage discounting procedure by Kolb and Scheraga (1990) is used: capital investments are annualised over a given time horizon using an assumed (real) interest rate and then discounted at a (real) social discount rate. Given the Europe-wide scope of the IED, a time-constant social discount rate of 4% is used, recommended in the European Impact Assessment Guidelines (European Commission 2009). While analysing the sensitivity of the chosen discount scheme is generally advisable (Pearce et al. 2006), its choice will not affect the overall result of the current study because the CBA does not compare alternative projects with different cost structures over time.

5.3.3 Determining private costs

Private costs are assessed with an MS Excel spread sheet provided by EGTEI (Expert Group on Techno-Economic Issues (2012))¹⁴, adapted as regards interest rate, equipment lifetime, emission factor without SCR and primary measures as indicated in Table 5.2 and catalyst cost (15 000 \in per m³, cf. (European Commission 2006b)). Following the twostage discounting procedure described above, capital investments related to the SCR retrofit are annualised over 20 years using an assumed (real) interest rate of 7% and then discounted at the (real) social rate of 4% mentioned above. Results are expressed in ξ_{2000} per year.

5.3.4 Determining damage costs

The quantified environmental benefits are due to avoided damage costs of air pollution. These are calculated by means of EcoSenseWeb version 1.3 (cf. section 5.2), according to the methodological assumptions specified in section 5.2.4 and the input data presented in Table 5.2 and Table 5.3. EcoSenseWeb results are expressed in \pounds -cent₂₀₀₀ per kWh. These are then multiplied by the annual amount of electricity generated in order to yield absolute damage costs in \pounds -cost in the section cost is the section cost is the section cost in the section cost is the section cost in the section cost is the section cost is the section cost in the section cost is the section cost in the section cost is the section cost is the section cost is the section cost is the section cost in the section cost is the section cost in the section cost is the section cost is the section cost in the section cost is the section cost in the section cost is the section cost is the section cost is the section cost is the section cost

¹⁴ An international expert group on the costs and technical parameters of power plant technologies; nowadays called TFTEI (Task Force on Techno Economic Issues), see <u>http://tftei.citepa.org/work-in-pro-</u><u>gress/costs-of-reduction-techniques-for-lcp</u> for further information; last accessed: 2017-05-18. The spreadsheet used here is an earlier version of the cost assessment tool adapted for CBA in chapter 7.

5.4 Results and discussion

In this section, the standard CBA and selected sensitivity analysis results are presented, followed by a discussion on limitations (weak point analysis of the EcoSenseWeb model).

5.4.1 Costs

The annualised capital costs as well as operating costs as shown in Table 5.4 enter the CBA as private costs.

Table 5.4:Private costs of installing a SCR and primary abatement measures at a coal-fired power station,
according to EGTEI (2012)

Hard Coal, (grade 2) ^a : existing plant, retrofitted with primary measures and SCR				
Investment costs without catalyst $[\mathbf{\varepsilon}_{2000}]$	25 056 842			
Costs for catalyst [€ ₂₀₀₀]	11 842 105			
Annualised capital cost [€2000 per year]	3 483 000			
Annual operating costs [\mathcal{E}_{2000} per year]	3 361 012			

^a Characterised by lower heating value of 24.9 GJ/t and sulphur content of 0.8%, according to EGTEI (2006)

5.4.2 Benefits (avoided marginal damage costs)

According to the default EcoSenseWeb results, the marginal damage costs per kWh produced are reduced through the installation of a DeNOx by 33%, 22%, and 17% at Brussels, Cartagena and Helsinki, respectively (Figure 5.1).



Figure 5.1: Damage costs of a coal-fired power plant located in Brussels (Belgium), Cartagena (Spain), or Helsinki (Finland) without (wo) and with (w) DeNOx, calculated by means of EcoSenseWeb v1.3 (default assessment)

These results translate into recurring annual benefits as the difference between both scenarios, entering the CBA, as shown in Table 5.5.

 Table 5.5:
 Annual damage costs without (wo) and with (w) DeNOx and resulting annual benefits (default assessment)

	Damage costs wo DeNOx [€ ₂₀₀₀ /a]	Damage costs w DeNOx [€ ₂₀₀₀ /a]	Annual benefit (wo – w) [€ ₂₀₀₀ /a]
Brussels	112 036 500	74 626 700	37 409 800
Cartagena	75 793 900	59 232 500	16 561 400
Helsinki	62 218 800	51 930 500	10 288 300

5.4.3 CBA results

Entering the annual costs (Table 5.4) and the annual benefits (Table 5.5) into the CBA decision rule (cf. section 3.2) shows that for the standard case, the SCR retrofit can be

said to be proportionate for all locations investigated (Table 5.6). The magnitude of the net present value clearly reflects the differences in population density in the affected geographic areas. Reducing emissions at a site that is surrounded by densely populated regions (Brussels) is clearly more beneficial than the same reduction in a region characterised by a low population density. Differences in meteorology also play a role, though the impact on the results is more complex and could not be assessed here.

Table 5.6:Discounted benefits, discounted costs and net present values of the CBA of a SCR installation at a
coal-fired power plant at three different locations (default assessment)

	Discounted benefits [€ ₂₀₀₀]	Discounted costs [€ ₂₀₀₀]	Net present value [€ ₂₀₀₀]
Brussels	528 691 310	96 732 845	431 958 466
Cartagena	234 007 154	96 732 845	137 274 310
Helsinki	145 395 835	96 732 845	48 662 990

5.4.4 CBA results – sensitivity analysis

For the sites with the highest (Brussels) and lowest results (Helsinki), the influence of one technical (full load hours per year; section 5.4.4.1) and three methodological assumptions (particle toxicity, meteorological conditions, and background emissions, section 5.4.4.2) are investigated. Even though EcoSenseWeb allows further settings to be changed, excluding certain models (e.g. on local air quality) or including others (e.g. on hemispheric air quality or micro pollutants) is not deemed appropriate. The sensitivity of GHG emission valuation will not be analysed either despite being a contentious issue for methodological and uncertainty reasons. This study being concerned with NO_x emissions abatement, GHG emissions are hardly affected (cf. Figure 5.1). The impact of changing GHG valuation is thus expected to be limited.

5.4.4.1 Variation in full load hours (operation time)

Changes in European energy markets have led to less base- and mid-load generation of coal-fired power plants (cf. section 2.3). When reducing the annual operating hours from 4500 to 2500, lower benefits from less emission reduction and lower (operating) costs of the SCR result at both sites (Table 5.7).

The CBA results suggest different conclusions for the two investigated sites. At reduced full load hours, a negative net present value is obtained for Helsinki. This result suggests that a short operation time combined with a less densely populated impacted region may make the SCR investment disproportionate. Still, the decrease to 2500 full load hours does not alter the CBA result for Brussels.

 Table 5.7:
 Discounted benefits, discounted costs and net present values for 2500 varying full load hours per year (otherwise default settings)

	Discounted benefits [€ ₂₀₀₀]	Discounted costs [€ ₂₀₀₀]	Net present value [€ ₂₀₀₀]
Brussels	293 748 803	82 359 007	211 389 795
Helsinki	80 785 671	82 359 007	-1 573 335

5.4.4.2 Varying assumptions on particle toxicity and regional air quality

With a considerable influence on quantified human health costs, the toxicity attributed to different types of particles is scientifically controversial. In the ExternE project series, different particle toxicities have been used (European Commission 1999a, 2004). At EU level and up to now, the WHO recommends considering particles of different sizes, of different composition and from different sources as equally toxic (WHO 2007). Recent evidence suggests that secondary particles originating from the power sector are less toxic than, for instance, primary particle emissions from road traffic (Grahame and Schlesinger 2007). Related to the revision of EU air quality policies in 2013, the WHO (2013b) revisited the question of particle toxicity, pointing out the differing evidence, however without changing its recommendation. Yet, to reflect this uncertainty, the CBA results based on equal particle toxicity are confronted with those based on differential particle toxicity, according to which health effects caused by primary particles (PM₁₀ and PM_{2.5}), by nitrates (secondary particles formed from NO_x) and by sulphates (secondary particles formed from SO₂) are multiplied by a factor of 1.3, 0.5, and 0.7, respectively.

The discounted benefits decrease substantially under the scenario "different particle toxicity" (Table 5.8). While the DeNOx investment can be considered proportionate for both sites, little changes such as a slight reduction in full load hours at the Helsinki plant, for instance, would alter the CBA outcome. Under future air quality model settings, however, the environmental benefits increase, making disproportionate costs less likely in the scenarios analysed. Note that there are further changes, in particular concerning demography that cannot be considered here.

	Particle toxicity	Air quality model setting	Discounted benefits [€ ₂₀₀₀]	Discounted costs [€ ₂₀₀₀]	Net present value [€ ₂₀₀₀]
	Equal	default	528 691 310	96 732 845	431 958 466
Brussels	Equal	future	723 811 757	96 732 845	627 078 912
	Different	default	262 094 118	96 732 845	165 361 273
		future	434 584 715	96 732 845	337 851 870
Helsinki	Equal	default	145 395 835	96 732 845	48 662 990
	Lquai	future	225 382 624	96 732 845	128 649 780
	Different	default	98 685 991	96 732 845	1 953 146
	Dijjerent	future	135 817 264	96 732 845	39 084 419

 Table 5.8:
 Discounted benefits, discounted costs and net present values as a function of default and sensitivity assumptions regarding particle toxicity and air quality modelling

5.5 Limitations and weak point analysis

Despite the effort made in selecting representative data and assumptions for the environmental CBA carried out in this chapter, practical and methodological limitations exist, discussed in the following.

5.5.1 Practical considerations regarding the application of the "disproportionate" cost criterion related to industrial emission control technologies

When transposed into Member State legislation the Industrial Emission Directive allows for different implementations of the "disproportionate costs" criterion. Although considering disproportionate costs helps to ensure that societal welfare is enhanced through air quality policies, some countries may decide not to allow for derogations on these grounds. According to the preamble of the IED and without providing more details, "welldefined criteria" shall be taken into account when granting derogations because of disproportionate costs. It will presumably be up to authorities at different levels to define standardised methods. European-wide harmonised standards would allow assessing sector activities across member states in a consistent, non-distortive way. At the time of writing (i.e. end of 2016), consultations with stakeholders, particularly concerned industry sectors, are ongoing at the level of the European Commission, aiming to develop European-wide and possibly sectorial reference methodologies for the assessment of costs and benefits from emission control. Examples of standardisation initiatives already exist by the German Federal Environment Agency (Umweltbundesamt 2012)¹⁵ at national level, or the Clean Air For Europe (CAFE) CBA methodology (Holland et al. 2005a) at European level. However, these recommendations are partly out of date, concerning most notably the monetary value used for assessing the risk of premature mortality (cf. chapter 4). This highlights the need to regularly revise methodological standards according to new scientific findings.

5.5.2 Weak point analysis of the EcoSenseWeb model

Whether the degree of spatial resolution in EcoSenseWeb is sufficient to be applied in the frame of IED derogations regarding "the geographical location or the local environmental conditions of the installation concerned" is a critical and crucial question. The spatial resolution of population data used in the regional model underlying EcoSenseWeb is rather heterogeneous. Likewise, classical air pollutants are not assessed to be emitted at a specific site for the regional scale modelling. Rather these emissions are assumed to occur somewhere in a rather large region (country level or coarse subdivisions of larger countries such as France or Germany). The 50 km grid cell resolution of the regional dispersion model that feeds into EcoSenseWeb does not allow for a proper consideration of effects at the local scale. At the same time, the local dispersion model of EcoSenseWeb, covering an area of 100 km around the emission source, is said to provide "only rough estimates on a coarse resolution", being "more reliable for flat … terrain" (Preiss and Klotz 2008). Moreover, the local model only considers health effects induced by primary PM, thus lacking health impacts due to ozone or secondary particles. Especially in urban areas,

¹⁵ An update of this work with central contributions by this thesis' author regarding the assessment of impacts from classical air pollutants has recently been concluded (van der Kamp et al. forthcoming).

additional approaches are therefore employed in order to better account for higher population densities in exposure modelling when using EcoSenseWeb results, e.g. (Umweltbundesamt 2012, van der Kamp et al. forthcoming).

EcoSenseWeb offers a limited number of assessment options (cf. section 5.2.4). Increasing the degree of freedom through more variable parameter setting options is desirable from a scientific point of view. In particular, in order to integrate scientific advancements, impact functions, and monetary valuation parameters should be directly modifiable¹⁶. From an application point of view, however, it would be at the expense of user friendliness and also in contrast to the potential strive for a harmonised tool including default settings to be used by authorities across Europe.

EcoSenseWeb does not provide specific information on the estimates' uncertainties. General uncertainty indications related to the health-related damage costs of classical air pollutants are mentioned in the user manual, referring to an assessment of a different air quality model that relies on similar impact functions and monetary values: a typical geometric standard deviation of 3 is suggested for health impacts via inhalation (Spadaro and Rabl 2008), cf. also section 8.2.1. For other impacts assessed above, no uncertainty information is provided. Uncertainty indications being important in decision-making contexts, this lack of detailed uncertainty information in EcoSenseWeb needs remediation.

An important limitation not only of EcoSenseWeb, but also of other external cost assessment frameworks currently available (e.g. the CAFE CBA) is the assessment of biodiversity and ecosystem services-related impacts. In EcoSenseWeb, the potential disappearance of a part of a target plant species for Dutch conditions is used and extrapolated to the rest of Europe (cf. (Ott et al. 2006). At the same time, the current state-of-the-art of biodiversity-related impact assessments concerns the evaluation of changes in ecosystem services (TEEB 2010). In addition, EcoSenseWeb relies on restoration costs for valuation. These can only reliably be used under specific conditions that are rarely met (Bockstael et al. 2000, OECD 2002).

Beyond the limited assessment options and lacking uncertainty consideration, several inconsistencies underlying the methodology used in EcoSenseWeb shall be mentioned. These concern, for instance, different reference years (e.g. meteorological data, receptor

¹⁶ The indirect approach for modification of EcoSenseWeb input parameters employed in chapter 4 is clearly more time and resource consuming than adapting parameters directly within the tool.

data or studies on monetary values), different degrees of geographical validity (e.g. monetary values or risk functions), and different models for assessing concentrations of SO₂ and other classical air pollutants.

Most critically in view of the changing conditions in the European energy sector, the EcoSenseWeb model does not allow distinguishing between different operation schedules. Instead, due to the underlying regional dispersion modelling, constant emissions throughout a given year are assumed. As a result, power plants with relatively low operating hours per year, such as peak-load plants, cannot be dealt with appropriately. Moreover, seasonal peaks of given power plants, such as in the colder winter months, are not well represented. The impacts of peak-load plants with a variable operation pattern throughout the year may thus differ substantially from those assessed by EcoSenseWeb, particularly if effect thresholds are involved (e.g. for ozone). To address this limitation and analyse its importance, a dedicated case study is carried out in section 7.1.3. Besides, the results obtained by using the advanced assessment approach developed in this thesis and those obtained by EcoSenseWeb are compared in section 7.3.2.

5.6 Summary of chapter 5

Using an exemplary case from the power sector, a social cost-benefit analysis is carried out, integrating damage costs from an integrated assessment model, i.e. EcoSenseWeb. The guiding question is whether an emission control measure at a fossil-fired power plant leads to disproportionate costs according to the EU's Industrial Emissions Directive. The chosen example, a DeNOx investment, proves to be socially efficient at all assessed geographic sites. Costs tend to be disproportionate under certain conditions, e.g. lower operating hours per year and remote geographic locations, justifying case-specific derogations. As a result, the techno-economic characteristics of the emission control measures in question and the conditions under which these are installed and operated influence the result of the disproportionate cost evaluation. This shows the need for competent authorities to rely on rather detailed assessments in terms of site and operation conditions when granting derogations due to variable environmental conditions. It is questionable whether the EcoSenseWeb model fulfils this condition when it comes to evaluating impacts from power plants with variable operation profiles, as discussed in the weak point analysis.

6 Development of a cost and benefit assessment methodology for emission control measures at point sources

This chapter, as the methodological core of the thesis, describes how economic assessment approaches are enhanced in order to carry out social cost-benefit analysis (CBA) at the power plant level whilst accounting for variable operation profiles.

6.1 Development of a health damage cost assessment framework

In view of the central objective of the current work, i.e. accounting for the site- and timedependence of health damage costs, the following requirements are formulated:

- Air quality modelling: considering the precise location of the emission source and the fate of environmental emissions. This includes a highly time- and space-resolved dispersion modelling, accounting for meteorology, chemistry, topography, background emission levels, etc.;
- Exposure assessment: considering population distribution, numbers, and age statistics, preferably in a spatial resolution similar to that of the air quality modelling;
- Health impact assessment: following latest recommendations and using representative input data;
- Monetary valuation: following latest recommendations and using representative input data.

The assessment framework has been developed with these requirements in mind. Following the steps of the impact pathway approach, a marginal damage cost approach is implemented (Figure 6.1).



Figure 6.1: Schematic overview of input data, assessment steps and output of the health damage cost assessment framework; own conception based on the impact pathway approach

Atmospheric dispersion modelling has been carried out by CEREA¹⁷ using the Polyphemus platform. The resulting ambient concentration statistics have been integrated into a geographic information system (GIS) for exposure assessment, before proceeding towards a physical and monetary impact assessment. Underlying data processing steps and data sources are summarised in Table 6.1 and specified in the following sections.

¹⁷ Centre d'Enseignement et de Recherche en Environnement Atmosphérique, a French common research laboratory between EDF Research and Development (R&D) and École des Ponts, ParisTech

	Data input	Data processing steps
Emission (source) scenario definition	Power plant operation scenarios based on own as- sumptions; cf. Table 6.2	Scenario construction: defining (hourly) emission rates of air pollu- tants/pollutant groups (SO ₂ , NO _x , and PM) during one year
Atmospheric modelling	Power plant emission scenar- ios, background emission in- ventory and meteorology; cf. section 6.1.1, Table 6.4	Chemical transport modelling: carried out by CEREA using the Polyphemus platform (Mallet et al. 2007)
Exposure assessment	Hourly concentration fields of PM ₁₀ , PM _{2.5} , and ozone (O ₃) (Polyphemus output data) Population counts and ad- ministrative boundaries; cf. section 6.1.2, Table 6.5	Data extraction, i.e. extracting concen- tration fields from binary files and im- porting them into geodatabases: defin- ing headers (including information on the geographic extensions, the grid cell resolution and number format amongst others) and geographic pro- jections (possibly using geographic transformations) to generate raster data; Zonal statistics: using "resampling" (generating a very fine grid cell resolu- tion, depending on the size of the cor- responding administrative units) and "zonal statistics as table" functions in ArcGIS® 10.3 in order to define aver- age concentration levels per adminis- trative unit (France: commune level; Europe: NUTS3 level); storage of gen- erated data in table format; Exposure calculation: merging infor- mation on spatially and temporally re- solved concentration changes with spatially resolved population counts.
Health impact assessment	Age group data, risk group data, health baseline data, and risk functions; cf. section 6.1.3, Table 6.6 and Table 6.7	Merging data on population exposure with health impact functions; specific approach for assessing long-term ex- posure mortality risks in adults using life table methods

Table 6.1: Data input and processing steps used in the health damage cost assessment (own compilation)

	Data input	Data processing steps
Monetary valuation	EU-average parameters adapted from Holland (2014a); cf. section 6.1.4, Table 6.8	Merging information from health im- pact assessment with monetary values

6.1.1 Air quality modelling

6.1.1.1 Definition of emission scenarios

Different operation scenarios and corresponding emission levels at power plant level were defined (Table 6.2) according to the research objectives underlying this work, i.e. advancing the assessment methodology of emission control measures at fossil fuel power plants with variable operation profiles.

These scenarios enable the analysis of methodological aspects (e.g. influence of spatial and temporal modelling resolution) as well as the application of CBA at site-level, accompanied by sensitivity analyses (cf. chapter 7). Note that only the operation phase of the power plant is considered, while related up- and downstream activities are not deemed relevant here. In order to illustrate the annual operation pattern of the power plant units in the scenarios S1, S2, and S3, the daily SO₂ emissions of scenario S2 are displayed (Figure 6.2). Further relevant characteristics of the power plant are as follows: heavy fuel oil-fired power plant, boiler, consisting of 4 units and located in Western France. The stack height is assumed to be 220 m and the stack diameter 5.37 m.

 Table 6.2:
 Power plant operation scenarios and related emission levels, valid for units 1-4 of the power

 plant respectively; parameter changes from one scenario to the next in bold; own assumptions

Sce- nario	Full load hours (eq.)	Elec. capac- ity per unit	Flue gas vol- ume stream	SO₂ concen- tration	SO ₂ emit- ted	NO _x concen- tration	NO _x emit- ted	PM concen- tration	PM emit- ted
S0	Units 1-4	4 of the po	wer plant do	not operate	(baseline	scenario)			
S1	500 h/a	600 MW _{el}	1800000 Nm³/h	Fuel with 0.33% sulphur content; 600 mg/Nm ³	540 t/a	Initial state: 600 mg/Nm³	540 t/a	Initial state: 40 mg/Nm ³	36 t/a
S2	500 h/a	600 MW _{el}	1800000 Nm³/h	Fuel with 0.22% sulphur content: 400 mg/Nm ³	360 t/a	Low-NOx burner; 400 mg/Nm ³	360 t/a	Initial state: 40 mg/Nm ³	36 t/a
S2b	8760 h/a	34 MW _{el}	1800000 Nm³/h	Fuel with 0.22% sulphur content: 400 mg/Nm ³	360 t/a	Low-NOx burner; 400 mg/Nm³	360 t/a	Initial state: 40 mg/Nm ³	36 t/a
S2c	8760 h/a	600 MW _{el}	1800000 Nm³/h	Fuel with 0.22% sulphur content: 400 mg/Nm ³	6307 t/a	Low-NOx burner; 400 mg/Nm³	6307 t/a	Initial state: 40 mg/Nm³	630.7 t/a
S3	500 h/a	600 MW _{el}	1800000 Nm³/h	Fuel with 0.22% sulphur content: 400 mg/Nm ³	360 t/a	Low-NOx burner; 400 mg/Nm ³	360 t/a	Electro static precipi- tator: 20 mg/Nm ³	18 t/a



Figure 6.2: Daily SO₂ emissions per unit in scenario S2, reflecting the operation profile of the oil-fired power plant throughout the year; own compilation

6.1.1.2 Atmospheric modelling

Table 6.3 summarises elements of the scope definition related to atmospheric dispersion modelling and illustrates the scope of the methodology developed in this thesis.

Table 6.3:	Elements to be considered in the scope definition of atmospheric modelling and specific choices
	made in the current thesis (grey shaded); own compilation

Emission source	point source		line source	5	area	a source
Considered pollutants	primary			secondary		
Spatial extension	local	nati	onal	regional		global
Spatial resolution	high		medium		low	
Time extension	day	mor	nth	season		year
Time resolution	hour	day		month		year
Air quality modelling	simple box mode	el	Gaussian		Eule	erian
Modelling approach	averaged		parameter	ised	spe	cific

As regards the spatial scope, ambient air pollution concentration data related to the different emission scenarios (Table 6.2) at three scales were obtained:

- 1) Local domain, i.e. *Île-de-France* region and parts of surrounding regions (bottom-left of Figure 6.3);
- 2) National domain, i.e. mainland of *France* and Corsica (top-right of Figure 6.3);
- 3) European domain, i.e. large parts of *Europe*¹⁸ (Figure 6.4).



Figure 6.3: Modelling domain "France", displaying the "canton"¹⁹ level, with enlarged section of "Île-de-France" modelling domain, displaying "communes" per (colour-differentiated) NUTS3 zone; emission source located at white dot; maps based on IGN (2013)

¹⁸ Although this domain includes parts of Bosnia and Herzegovina (mapped without background colour fill), these countries are excluded from the analysis due to lacking administrative data. The same applies to Andorra and San Marino.

¹⁹ Displaying the "commune" level at national scale is not helpful here - the figure would become unreadable due to the over 36 000 communes on the territory.



Figure 6.4: Modelling domain "Europe", displaying NUTS3 administrative zones per (colour-differentiated) country; map based on eurostat (2015a), © EuroGeographics for the administrative boundaries

The background emission and meteorological data of the year 2009 were used in the air quality modelling, as its meteorological conditions²⁰ are considered representative for current day analyses, i.e. marked by increasing average temperatures when compared to the historic long-term average rates.

In preparing the WRF meteorological data, a so-called 1-way nesting was used, meaning that meteorological boundary conditions from a higher-level scale are passed on towards the lower-level in order to obtain more accurate simulation results at the borders of the respective domains.

²⁰ Cf. <u>http://www.meteofrance.fr/climat-passe-et-futur/bilans-climatiques/autres-annees/bilan-de-lannee-2009</u>, last accessed: 2017-05-18

The atmospheric dispersion modelling was carried out using the software package Polyphemus (Mallet et al. 2007). Detailed specifications and input data are summarised in Table 6.4.

	Île-de-France	France	Europe
Air quality models used (modules within Polyphemus platform)	Polair3D (Eulerian model), "plume-in-grid" modelling (Gaussian)	Polair3D (Eulerian model), "plume-in-grid" modelling (Gaussian)	Polair3D (Eulerian model), "plume-in-grid" modelling (Gaussian)
Geographic coordinates of domain (decimal degrees)	1.2°E - 3.4°E/ 48.2°N - 49.69°N	5°W - 9.84°E/ 41.2°N - 51.32°N	10.2°W - 19.03°E/ 36°N - 58.49°N
Horizontal and vertical grid cell resolution at surface	0.045° (~3 km x ~5 km)	0.2249° (~17 km x ~25 km)	0.4497° (~35 km x ~50 km)
Input: emission source characteristics	Oil-fired power plar Figure 6.2	nt, 4 units; operation p	profiles accord. to
Input: emission inventory - year	European Monitorii 2009/EMEP - 2020	ng and Evaluation Prog (projected scenario)	gramme (EMEP) -
Input: meteorology - year	WRF 3.6.1 (Weathe 2009	r Research and Foreca	isting Model 2015) -
Input: land cover data	Global Land Cover 2	2000 database (Hartle	y et al. 2006)
Output: concentration fields	Substances/substar PM _{2.5/10} (primary/se	ice groups available: N condary (organic/inor	10 , NO_2 , SO_2 , O_3 , ganic) fractions)
Output: time coverage	365 days, 1-hour tir	ne steps	

Table 6.4: Characteristics and data sources of the atmospheric modelling; based on Legorgeu (2016)

Polair3D (Boutahar et al. 2004), the main module used, is an Eulerian chemistry-transport model that accounts for advection (transport of air by wind), turbulent diffusion (random and chaotic time-dependent motions of air, mainly vertical) and chemical transformations, involving gaseous and aqueous chemistry (Figure 6.5).



Figure 6.5: Atmospheric processes considered in Eulerian chemistry-transport models; reprinted with permission from and based on Sportisse (2007)

Differential equations within 3-dimensional grid cells are used to estimate the fate of aerosols and other substances in the atmosphere. It builds on the SIREAM (SIze REsolved Aerosol Mode) model for defining size distributions (Debry et al. 2007), distinguishing 17 aerosol species and treating effects such as condensation, evaporation, coagulation, and nucleation. In a post-processing step and using the models ISORROPIA (Nenes et al. 1998) and H²O (Couvidat et al. 2012) respectively for secondary inorganic aerosols and secondary organic aerosols, Polair3D allows to consider the formation of secondary PM species. For the current application, a modelling interval of 600 seconds is used, but only every 6th time-step, equivalent to one hour, is extracted as output concentration field in order to limit the overall data amount. Eight vertical layers are considered in the modelling used here (Legorgeu 2016), of which only the lowest layer (i.e. ambient air) is relevant for health impact assessment.

In a complementary analysis and for a more robust dispersion modelling around the point emission source, a so-called "plume-in-grid" modelling is applied (cf. section 7.1.4), further described by Kim et al. (2014) and Korsakissok and Mallet (2010). This means that a Gaussian model is embedded within the Eulerian model domain, simulating the dispersion of the emission plume near the source at higher temporal and spatial resolution. As a consequence, the diffusion and dispersion of the pollutants are modelled with a higher precision before eventually injecting the concentration field data into the Eulerian modelling grid. The following specifications for the "plume-in-grid" modelling were used:

- Modelling time frequency: 200 seconds;
- Criterion used for injecting puff data from Gaussian into Eulerian model: either as soon as the horizontal size of the puff reaches the size of the Eulerian grid cell size or, if the first condition is not met, after a duration of 1 hour (puff travel time);
- Use at the metropolitan and national modelling domain, given that the effect at the European scale is expected to be limited.

Apart from verifying the consistency of the modelling outputs, a validation of modelled concentrations with real-world monitoring air quality data was not carried out in this thesis, given that this subject has already been addressed elsewhere, e.g. by Tombette and Sportisse (2007), Zhang et al. (2013), Lecœur and Seigneur (2013) or Kim et al. (2014). According to these, the Polyphemus models predict concentrations of ozone and of overall PM_{2.5} rather well, whereas the results for secondary PM in terms of nitrates and sulphates are less robust (cf. also section 8.4.1).

6.1.2 Exposure assessment

Exposure assessment-relevant data and sources are summarised in Table 6.5.

	Île-de-France	France	Europe
Population counts: level of detail and data source	Commune, year 2009 (INSEE 2009b)	Commune, year 2009 (INSEE 2009b)	NUTS3, year 2009 (eurostat 2015c)
Administrative bound- aries: level of detail and data source	Commune, year 2011 (IGN 2013)	Commune, year 2011 (IGN 2013)	NUTS3, year 2010 (eurostat 2015a) © EuroGeographics

Table 6.5: Data used for exposure assessment (own compilation)

Administrative boundaries are classified according to their NUTS (Nomenclature of Territorial Units for Statistics) level, a classification of geographic units established by the European Union²¹. NUTS3 refers to small regions, e.g. the "département" level in France. For surrounding countries, the sizes of the NUTS3 regions differ (cf. Figure 6.4), depending mainly on population numbers. Below NUTS3, local administrative units (LAU) exist, notably LAU2 which corresponds to the "communes", i.e. commune level in France (Figure 6.3).

A key element during exposure modelling is the spatial distribution of the population and the density per modelling grid cell. Below, the spatial distribution in France per commune is displayed (Figure 6.6).



Figure 6.6: Population number per commune in France in 2009 and emission source location (white dot); based on IGN (2013)

In order to combine the concentration raster data coming from the atmospheric modelling with population data that is available per administrative unit, the "zonal statistics" function, as implemented in ArcGIS version 10.3, is used. Its principle is displayed in Figure 6.7. In a first preparatory step, the raster data is resampled in order to obtain a higher resolution raster. The zonal statistics function then calculates the average value

²¹ <u>http://eur-lex.europa.eu/legal-content/EN/TXT/?qid=1424956678004&uri=URISERV:g24218</u>, last accessed: 2017-05-18

per administrative unit based on the raster data that is overlapping the respective administrative units. In case of several raster cells overlapping one administrative unit, the weighted average of the raster data is calculated.



Figure 6.7: Data processing during exposure assessment, featuring the zonal statistics function; exemplary scheme for the European domain (own compilation)

6.1.3 Health impact assessment

Health impact assessment aims at estimating physical health endpoints, using risk parameters, baseline rates and the affected population fractions. Unlike atmospheric modelling and exposure assessment that are carried out in a highly site-specific way, health impact assessment partly relies on EU-average data for both availability and consistency reasons (Table 6.6).

	Île-de-France	France	Europe
Age group data	Ile-de-France data, year 2009 (INSEE 2009c)	French average data, year 2009 (INSEE 2009c)	EU-average data, year 2009 (WHO 2016)
Health baseline data	French average data, year 2009 (WHO 2016)	French average data, year 2009 (WHO 2016)	EU-average data, year 2009 (WHO 2016), except for long-term exposure mortality assessment
Health risk functions	EU average data accordi	ng to WHO (2013a), cf. Ta	able 6.7

Table 6.6: Data used for health impact assessment at the three assessment domains (own compilation)

6.1.3.1 Parameter selection for health impact assessment

Numerous studies have studied the links between exposure to ambient air pollutants and changes in health status of the concerned population. This has led to a large variety of available health risk functions for many different endpoints.

For European assessments, reference datasets have been regularly compiled, e.g. in research projects related to the ExternE project series or by the World Health Organization (WHO), cf. also chapter 4. These respectively reflect the best available scientific evidence of their time. The last concerted effort at European level was the HRAPIE (Health Risks of Air Pollution In Europe) project by the WHO that recommends health impact functions for use in regulatory cost-benefit analysis (WHO 2013a) based on an extensive meta-analysis. Since these recommendations reflect the latest and most robust scientific evidence for European boundary conditions, it was decided to use them as basis for health impact assessment. Table 6.7 summarises health endpoints and associated data used by default in the damage cost assessment. It includes a quality metric (Q), differentiating between two groups of endpoints (WHO 2013a):

- A*: pollutant–outcome pairs for which enough data are available to enable reliable quantification of effects;
- B*: pollutant–outcome pairs for which there is more uncertainty about the precision of the data used for quantification of effects.

The asterisk (*) denotes that these endpoints are additive, i.e. they can be aggregated without risking double-counting of effects. For the default impact assessment, only effects from groups A* and B* are considered, noting that further, less robust, endpoints exist.

A double counting risk remains regarding NO₂ endpoints and corresponding endpoints related to PM_{2.5} or PM₁₀; therefore NO₂-related effects are currently not considered by default (indicated through grey colour in Table 6.7). However, a sensitivity analysis that includes NO₂-related effects is carried out, cf. section 7.2.1.

During data retrieval and compilation, recommendations from Holland (2014b), related to the WHO's HRAPIE project, were largely followed, while noting that:

- Net restricted activity days are defined as follows (Torfs et al. 2007, WHO 2013a): NetRAD = RAD – WLD – (RHA and CHA)*10 days - (asthma symptom days*(5/3));
- Work loss day statistics: EU average baselines were used since no up-to-date data was available for France in the WHO Health For All Database (HFA DB); employment rate taken from eurostat (2015b).

As regards the concerned population fraction, age group data is displayed in Table A.3 (Appendix) and for the only risk group fraction distinguished, i.e. asthmatics among children, the average prevalence provided by the WHO for Western Europe (4.9 %) is used throughout the European modelling domain. Even though WHO recommends a different rate in Northern and Eastern Europe, this has not been considered here for practical reasons and the relatively minor importance of this endpoint. Baseline rates for France and Europe are given in Table A.4 (Appendix).

ď	Pollu- tant	Metric	Health endpoint	Risk group	Age group	СRF [1/(10µg/m³)]	Baseline person (l	(BL) per EU average)	Source of baseline	s _{cR} (=CRF* [1/(10μg/ person*aj	BL) m³)*
*A	PM _{2.5}	Annual mean	All-cause natural mortality	All	Adults30+	0.0620	0.0146	Deaths/year	WHO DMDB	0.00980ª	YOLL
*A	$PM_{2.5}$	Daily mean	Cardiovascular hospital admission	All	AII	0.0091	0.0240	Cases/year	WHO HMDB	0.00022	Cases
*A	$PM_{2.5}$	Daily mean	Respiratory hospital admission	All	AII	0.0190	0.0139	Cases/year	WHO HMDB	0.00026	Cases
*	$PM_{2.5}$	Annual mean	Net restricted activity day (netRAD)	All	AII	0.0318	19	Days/year	(Ostro and Rothschild 1989)	0.88818	Days
*8	$PM_{2.5}$	Annual mean	Work loss day (WLD)	AII	Adults20 to64	0.0460	9.8739	Days/year	WHO HFA DB, (eurostat 2015b)	0.45420	Days
*	PM_{10}	Daily mean	Asthma symptom day	Asthm. children	Children5 to19	0.0280	0.1700	Daily inci- dence	(WHO 2013a), ISAAC study	1.73740	Days
*	PM_{10}	Annual mean	All-cause infant mortality	All	Infants	0.0400	0.0014	Deaths/year	WHO HFA DB	0.00006	Deaths
*8	PM_{10}	Annual mean	Bronchitis prevalence	All	Children6 to12	0.0800	0.1860	Annual preva- lence	(WHO 2013a), PATY study	0.01488	Cases
*8	PM_{10}	Annual mean	Chronic bronchitis case	All	Adults18+	0.1170	0.0039	Cases/year	(WHO 2013a), SAPALDIA study	0.00046	Cases
*A	03	SOM035	All-cause natural mortality (acute)	All	AII	0.0029	0.0108	Deaths/year	WHO MDB	0.00003 ^b	JIOY
¥*	0 ³	SOM035	Cardiovascular hospital admission (excl. stroke)	All	Adults65+	0.0089	0.0578	Cases/year	WHO HMDB	0.00051	Cases
*A	03	SOM035	Respiratory hospital admission	AII	Adults65+	0.0044	0.0297	Cases/year	WHO HMDB	0.00013	Cases
* 8	°°	SOM035	Minor restricted activity day (MRAD)	All	AII	0.0154	7.8	Days/year	(Ostro and Rothschild 1989)	0.12012	Days
A*	NO_2	Daily 1-h max	All-cause natural mortality (acute)	All	All	0.0027	0.0108	Deaths/year	WHO MDB	0.00003	Deaths
A*	NO_2	Daily mean	Respiratory hospital admission	All	All	0.0180	0.0139	Cases/year	WHO HMDB	0.00025	Cases
*	NO_2	Annual mean, > 20 μg/m³	All-cause, natural mortality	All	Adults30+	0.0550	0.0146	Deaths/year	WHO DMDB	0.00980ª	YOLL
*	NO2	Annual mean	Bronchitis symptom days	Asthm. children	Children5 to14	0.0210	0.2990	Annual preva- lence	(WHO 2013a)	2.29184	Days
a B d	ased on	simplified YOL	L approach, described a	bove; at	the French	level, a detaile	d life table	e approach is u	ised, equally des	cribed abo	ve

HFA DB (Health For All Database); MDB (Mortality Database); HMDB (Hospital Morbidity Database); SOMO35 (Sum Of Means Over 35 ppb) Abbreviations used: Q: quality metric, as explained above; CRF: concentration-response function; DMDB (Detailed Mortality Database);

 Table 6.7:
 Health endpoints and data used in the assessment framework developed in this thesis;

 NO2-related endpoints (grey) for sensitivity analysis; based on WHO (2013a)
6.1.3.2 Assessing long-term exposure mortality impacts

Long-term exposure mortality impacts due to PM_{2.5} (and NO₂, used in a sensitivity analysis) merit special attention, given that this endpoint regularly represents the highest share among quantified health-related damage costs (Holland 2014a). Two assessment approaches can be distinguished (WHO Regional Office for Europe and OECD 2015):

- Quantification is based on years of life lost (YOLL), which are subsequently valued by a so-called Value of a Life Year (VOLY);
- Quantification is based on the number of premature deaths, which are subsequently valued by a so-called Value of a Statistical Life (VSL).

While infant mortality is quantified using the number of premature deaths, mortality in adults due to O₃, PM_{2.5}, and NO₂ is expressed as YOLL, requiring additional assumptions and data as explained in the following. A quantification of premature deaths as an alternative to YOLL is presented in section 7.2.2.

Life table method approach to estimate mortality impacts due to long-term $PM_{2.5}$ and NO_2 exposure

The life table (also called survival curve) method aims at estimating the number of life years lost (gained) related to changes in ambient air pollution levels by accounting for changes in population numbers and mortality rates over time. It is based on national life expectancy data and mortality baseline rates per age group, therefore considering the age of death in the assessment. By relying on age group-specific data, it considers the susceptibility, e.g. of elderly people, to die prematurely due to air pollution risks, a factor that cannot be easily considered when quantifying cases of death. At the same time, the fact of implicitly assigning fewer life years lost to the death of elderly people is subject to ethical objections, as this could lead to age discrimination (OECD 2016). Notwithstanding these objections, the EU has chosen the YOLL-based approach as standard when estimating air pollution-related impacts in adults (Hurley et al. 2005), whereas the U.S. EPA by default bases their impact assessments on cases of death without considering the associated lifetime lost (U.S. EPA 2011a).

For the mortality risk assessment of the current thesis, a life table approach based on Miller (2013) was implemented, using life table templates combined with French age group data and mortality statistics (INSEE 2009a, c). For the rest of Europe, an alternative approach is used, explained below.

- Approach and assumptions for life table calculation: The impact of a 1-year pulse reduction of 1 μ g/m³ ambient PM2.5 (or NO₂) concentration on the all-cause mortality rate of adults aged 30+ in France is studied, using the concentration-response functions currently recommended by the WHO (cf. Table 6.7) and a discount rate of 0%, consistent with Hurley et al. (2005). It is assumed that effects occur with a given time lag, i.e. 30% of the mortality rate reduction in year 1, 50% during years 2 to 5 and the remaining 20% during years 6 to 20, consistent with a lag scheme by the U.S. EPA (2011a).
- Results: By cumulating the years of life gained for the entire cohort over a period of more than 100 years (starting in year 2009) the life years gained across the French population aged 30+ are obtained. For PM_{2.5}, this number amounts to 41 407 life years gained per 1 μ g/m³ ambient PM_{2.5} concentration reduction, whereas for NO₂, due to the somewhat lower risk coefficient, it amounts to 36 856 life years gained per 1 μ g/m³ ambient NO₂ concentration reduction. Considering the relevant population statistics, this translates into 64.08 (101.79) YOLL per 100 000 people all ages (aged 30+) per 1 μ g/m³ ambient PM_{2.5} concentration increase or 57.04 (90.6) YOLL per 100 000 people all ages (aged 30+) per 1 μ g/m³ ambient NO₂ concentration increase during 1 year.

These figures, used hereafter as reference for the mortality risk assessment at French national level, are close to the European average of 65.2 YOLL per 100 000 people (all ages) per $1 \mu g/m^3$ ambient PM_{2.5} concentration increase during 1 year, recommended by the CAFE and NEEDS projects (Hurley et al. 2005).

For illustrative purposes, an alternative scenario has been calculated, assuming that a $1 \ \mu g/m^3$ reduction in PM_{2.5} concentration is sustained over 20 consecutive years (equivalent to the assumed power plant lifetime in the CBA case study, cf. section 7.3.1). This scenario leads to 1 338 YOLL avoided per 100 000 (all ages) for a $1 \ \mu g/m^3$ ambient PM_{2.5} concentration decrease during 20 consecutive years and exceeds the result of the 1 year pulse change multiplied by 20 (20*64.08 = 1280). This is because the 1 year pulse approach does not account for changes in the population at risk over the years (i.e. a gradual increase in the population at risk through reduced mortality and hence a greater benefit from a risk reduction), whereas the full life table method does account for this fact, as explained by Miller (2013).

Where life table methods are not applicable, e.g. due to data constraints, a simplified approach can be used instead (Miller et al. 2011). It uses a correlation based on national

life expectancy to approximate YOLL due to changes in ambient $PM_{2.5}$ concentration levels, cf. equation 6.1.

YOLL per 100 000 people (aged 30+) = exp (8.161 – (0.04478 × life expectancy)) 6.1

This simplified approach, also recommended by Holland (2014b), is used to derive a European average YOLL factor, applied to all countries except France in the European assessment domain (cf. Table A.5). It relies on life expectancy data for 2009 provided by the WHO (2016). No differentiation is made between males and females, having, however only a minor influence on the YOLL estimates.

As an approximation and due to the lack of alternative available quantification approaches, the same European average YOLL factor is equally applied to estimate long-term exposure NO₂ mortality risks in a sensitivity analysis (cf. section 7.2.1).

6.1.4 Monetary valuation

Human health effects are valued by considering the following components, each requiring dedicated valuation methods (Markandya and Ortiz 2011):

- resource costs, e.g. treatment costs;
- opportunity costs, e.g. foregone income;
- disutility costs, e.g. related to suffering as a consequence of an illness.

Resource costs are typically valued using prices of marketable goods or services. Opportunity costs, e.g. concerning lost productivity, are estimated by relying on (labour) market data. For the valuation of non-market goods, e.g. in the case of disutility costs, specific methods are used. These use surrogate markets to elicit individual preferences either directly (stated preferences) or indirectly (revealed preferences).

All three components should be valued for each health effect when following the theoretical concept of total economic value (Boardman et al. 2006), whilst leaving aside non-use values. In practice, this can be achieved, for instance, by using specifically designed stated preference methods, such as choice modelling or the contingent valuation method (CVM). A complementary, more pragmatic approach is to capture these different components through different endpoints, e.g. separating resource, opportunity, and disutility costs. This is what has been regularly done in European impact assessments.

Table 6.8 displays the monetary values per health endpoint, largely following latest European recommendations (Holland 2014a), that are used within the health damage cost assessment framework. As can be seen from the table, the three valuation components are covered through different endpoints and double counting is largely²² avoided through using separate endpoints for related effects.

6.1.4.1 Monetary valuation of mortality risks

The monetary valuation of mortality risks has a predominant influence on quantifiable health damage costs and, given its intangible nature, involves higher uncertainties than the valuation of morbidity. In order to confront ethical concerns, it is important to stress that related assessments value a risk reduction of dying prematurely due to air pollution and not human life as such.

The reference VOLYs proposed in Table 6.8 and used in chapter 7 differ from the core values that have been used to support the impact assessment of recently proposed European air quality policies (Amann et al. 2014, Holland 2014a) and that in turn rely on a survey carried out in three countries during a previous European research project (European Commission 2004, Hurley et al. 2005). Newer evidence, however, is available from the NEEDS project (Desaigues et al. 2011), i.e. a study conducted in nine European countries and that focuses specifically on air pollution-related mortality risks. It uses improved stated preference techniques and is therefore preferred for the given setting (cf. also the parameter choices in chapter 4).

Mortality valuation remains a debated issue and alternative valuation approaches exist, e.g. using the value of a statistical life (VSL) approach, such as preferred in the US context (U.S. EPA 2012c) or other international institutions (OECD 2012), but also used for the valuation of infant mortality hereafter. Using a VSL-based approach is therefore tested in a sensitivity analysis (cf. section 7.2.2).

6.1.4.2 Adjustment of monetary valuation parameters across space and time

When using existing reference values for the assessment of environmental or health impacts in a different setting, i.e. concerning space or time, several adjustments should be considered (Pearce et al. 2006):

²² One exception can be noted: In valuing net restricted activity days (RAD), it is not entirely consistent to include a component that refers to lost productivity, which relates to work loss days that are quantified separately. However, for consistency reasons and due to the minor importance of net RAD in the overall results, the valuation as suggested by Holland (2014a) will nevertheless be used.

- Time value of money, i.e. the use of discounting, cf. section 6.3;
- Price level changes (inflation): Given that the values in Table 6.8 were originally provided for the monetary base year 2005, an annual average inflation rate of 1.87% within the Euro zone of for the period 2005 to 2015 has been used to adapt the values to the monetary base year 2015 (eurostat 2016b). On this basis, an inflation adaption factor of 1.2 is obtained;
- Individual income changes and impact on willingness-to-pay: When transferring health-related values (especially intangible values derived by willingness-to-pay surveys) over time, it is common practice to adapt these for changes in individual income of the concerned population. Following OECD (2012) recommendations, the assumptions below are used to derive an adaptation factor:
 - Growth in individual income: real Gross Domestic Product (GDP) per capita in the EU28: 0.64% per year on average between 2005 and 2015 (eurostat 2016c);
 - Elasticity of willingness-to-pay with regard to personal income: assumed to be 0.8, following OECD (2012).

On this basis, an upscaling factor for the period from 2005 to 2015 of about 5.2% is obtained (((1+0.0064)^0.8)^10=1.0524).²³ This upscaling factor is applied in all calculations presented in chapter 7.

²³ As a simplification, it is assumed that all original health valuation factors used here were initially representing the willingness-to-pay of the year 2005.

Endpoint	Value [€ ₂₀₁₅] ^a	Reference	Resource costs	Opportunity costs	Disutility costs
all-cause infant mortality (VSL)	1925675	Adapted from European Commission (2004), Hurley et al. (2005)		Via CVM	Via CVM
all-cause natu- ral mortality (VOLY, long- term exposure)	48142	(Desaigues et al. 2011)		Via CVM	Via CVM
all-cause natu- ral mortality (VOLY, acute exposure)	72213	Based on Desaigues et al. (2011) ^b		Via CVM	Via CVM
asthma symp- tom day	51	CVM based on Ready et al. (2004)			Via CVM
bronchitis prevalence/ symptom days	708	(Holland 2014a)		Based on asthma symp- tom day (14 days à 47 €)	Based on asthma symptom day
chronic bronchitis case	64510	(Holland 2014a), based on Hunt et al. (2011) and CVM by Maca et al. (2013) ^c	Via CVM	Via CVM	Via CVM
hospital admission (cardiovascular/ respiratory)	2672	(Hurley et al. 2005), CVM based on Ready et al. (2004)	Hospitali- sation costs (45%)	Lost produc- tivity (33%)	Via CVM (22%)
minor restricted activity day (MRAD)	51	CVM based on Ready et al. (2004)			Via CVM
(net) restricted activity day (RAD)	111	(Hurley et al. 2005), based on CBI 1998 study		Lost produc- tivity (direct costs)	Via CVM

Table 6.8: Default monetary valuation factors (net of income growth) and included components used in the assessment framework developed in this thesis (own compilation)

Endpoint	Value [€ ₂₀₁₅]ª	Reference	Resource costs	Opportunity costs	Disutility costs
work loss day (WLD)	156	(Holland 2014a), based on CBI 2013 study		Lost produc- tivity (direct costs)	

^a Parameters in €₂₀₀₀ or €₂₀₀₅ have been converted to €₂₀₁₅ using the above mentioned assumptions on price level changes; an adjustment for income growth has not (yet) been carried out for the numbers presented here

^b Assuming that the original value given by Desaigues et al. (2011) was implicitly discounted, the valuation of acute effects is up-scaled by a factor of 1/0.67, following European Commission (2005a)

^c Presumably, this value is a weighted average of the valuation of chronic bronchitis, mild chronic obstructive pulmonary disease (COPD) and severe COPD

6.2 Costing methodology

This section describes how the private costs for emission control techniques are characterised and provides specific values for use in the case study of chapter 7.3.1. Typical reference values and power plant data are used to derive annual costs over the power plant's life time and to display marginal abatement costs (per tonne of emission reduced) over a range of full-load hours per year.

6.2.1 Approach taken for techno-economic cost assessment

The approach used here is based on the Task Force on Techno Economic Issues²⁴ (TFTEI 2015), working under the mandate of the United Nations Economic Commission for Europe (UNECE) Convention on Long-Range Transboundary Air Pollution (CLRTAP). The TFTEI methodology builds on U.S. EPA (2002), while adapting elements and data for use in the European context and referring also to the EU's Reference Document on Economics and Cross Media Effects (European Commission 2006a). Among publicly available data, it is therefore considered a robust choice for cost estimation at plant level, striking a good balance between comprehensiveness and a manageable level of technical detail.

²⁴ Previously called Expert Group on Techno-Economic Issues (EGTEI); The spreadsheet is available on the TFTEI website: <u>http://tftei.citepa.org/work-in-progress/costs-of-reduction-techniques-for-lcp</u>, last accessed: 2017-05-18

To obtain the total annual costs, operational expenditures are added to the annualised capital costs (cf. section 3.3). Three specific, typical emission control measures are analysed in the following, i.e.:

- A dry low-NO_x burner (LNB) for NO_x emission reduction,
- switching to heavy fuel oil with reduced sulphur contents for SO₂ reduction, and
- an electrostatic precipitator (ESP) for PM reduction.

Reduction measures can be generally divided into primary and secondary measures. Primary measures aim at directly reducing the environmental burden at its source, e.g. during the combustion process. Secondary measures are also called end-of-pipe measures since they are designed to reduce emissions after their formation.

Note that even though the TFTEI spreadsheet allows considering part-load operation, this feature requires that efficiency losses and related emissions are determined by the user. Given that this data has not been available for the case studied here, the cost characterisation is based on full-load operation equivalents. Real reduction costs are therefore expected to be somewhat higher than those presented below.

6.2.2 Cost characterisation of potential emission control measures at a heavy fuel oil-fired power plant

The costs of retrofitting specific emission control measures at existing heavy fuel oil-fired power plant units are estimated using a set of general parameter assumptions (Table 6.9) and the default settings of the TFTEI spreadsheets. A simple script was developed to generate cost estimates over a range of full load hours that are displayed below.

Operating devices such as LNB and ESP not only leads to additional electricity consumption, reflected in the operating costs, but also to slightly increasing atmospheric emissions per unit of electricity produced. Due to the minor importance, this effect on emissions has been disregarded in the cost characterisation.

Parameter	Value
Monetary base year	Euro ₂₀₁₅
Thermal capacity per power plant unit	1715 MW _{th}
Electric efficiency	35%
Heavy fuel oil lower heating value at 0.33% sulphur	40.55 MJ/kg (based on detailed fuel compo- sition)
Plant lifetime (= equipment lifetime)	20 years
Interest rate (weighted average cost of capital)	6% (standard scenario)
Electricity costs	40 € per MWh _{el}
Fixed operation & maintenance (O&M) costs	3% of investment
Full load hours	500 full load hour equivalents per year, cor- responding to a capacity factor of 5.7% (standard scenario)

Table 6.9:General parameters used for cost characterisation of emission control measures
(own assumptions)

6.2.2.1 Cost characterisation of low NO_x burner (LNB)

Installing a LNB is a primary measure (i.e. modification of the combustion process) to reduce NO_x emissions. The injection of fuel and the air/oxygen mix into the combustion zone is altered, enabling to reduce the formation of both fuel NO_x and thermal NO_x (European Commission 2006b).

For the case studied here, the following specific assumptions are taken:

- NO_x concentration reduced from 600 to 400 mg/Nm³ through installing a LNB;
- specific equipment investment of 5.7 €2015/kWth.

Per power plant unit, these assumptions yield (TFTEI 2015):

- capital costs of 9 775 500 €2015 (= 5.7 €2015/kWth * 1715000 kWth);
- annual operating and maintenance costs of 293 265 €2015 (3% of 9 775 500 €2015).

Applying the annuity method and using equation 6.1 this translates into annualised costs of 1 145 538 \in_{2015} , or 7 261 \in_{2015} per tonne of NO_x reduced (at 500 full load hour equivalents).

Given the fixed percentage of operation and maintenance costs as well as a specific investment that depends only on the power plant capacity, the annualised costs are independent from actual full load hours. This represents a simplification in that a more intensive use potentially leads to an earlier replacement of technical components, not being considered here.

Annualised costs per tonne of NO_x reduced are dependent on full load hours, illustrated by the graph below (Figure 6.8). It shows a remarkable cost per tonne increase at low utilisation rates (i.e. low operating hours), mainly driven by annualised capital costs.



Figure 6.8: Cost curve (quasi-marginal abatement costs) for installing a low NO_x burner at an oil-fired power plant unit; dotted line at 500 h/a; based on own calculations

6.2.2.2 Fuel switching to very low/ultra-low sulphur heavy fuel oil

A relatively simple and low-cost primary measure for SO_2 emission reductions is fuel switching towards heavy fuel oil with lower sulphur contents. A potential constraint is the

availability of heavy fuel oil with ultra-low sulphur contents, but new European regulations requiring lower sulphur contents in the shipping sector (i.e. Directive 2012/33/EU) are likely to increase the supply of such fuels.

For the case studied here, it is assumed that the use of heavy fuel oil with 0.33% sulphur content allows to achieve a SO_2 flue gas concentration of 600 mg/Nm³ and reducing to 0.22% sulphur allows to respect 400 mg/Nm³.

The price differential for moving from a fuel with 0.5% sulphur content to 0.1% sulphur content is around 199-229 \in_{2009} /tonne. This value is based on Kalli et al. (2009) and applies to shipping fuels. Since a literature search for data applying to combustion fuels did not deliver any results, this value is adopted for the case studied here. Assuming an average Euro zone inflation rate of 1.47% between 2009 and 2015 (eurostat 2016b), the lower bound translates into 54.30 \in_{2015} per tonne per 0.1% sulphur content change. The specific reduction costs per tonne of emission reduced are constant, as they increase linearly with full load hours and thus with emission quantities. For these parameters choices, the abatement costs for SO₂ when switching from fuel with 0.33% sulphur to 0.22% sulphur amount to approximately 26 146 \in_{2015} per tonne of SO₂ removed (TFTEI 2015).

In order to estimate annual costs for this control measure, the total amount of reduced SO₂ per year and dependent on operating hours needs to be defined. Assuming a concentration decrease from 600 mg/Nm³ to 400 mg/Nm³, a flue gas volume stream of 1 800 000 Nm³/h (Table 6.2) and 500 full load hours per year, this yields an annual reduction of 180 tonnes²⁵ of SO₂. Multiplying these with the marginal cost of 26 146 \in_{2015} per tonne of SO₂ removed therefore leads to annual costs of 4 706 280 \notin_{2015} for the fuel switching per power plant unit and the given operating time.

6.2.2.3 Cost characterisation of an electrostatic precipitator (ESP)

For dust emission reduction, electrostatic precipitators are the most commonly applied secondary reduction technique in large combustion plants. The functional principle is to apply an electric charge to gaseous dust particles within the flue gas stream that are subsequently absorbed at oppositely charged collector plates (European Commission 2006b).

²⁵ (600-400) mg/Nm³ * 1 800 000 Nm³/h * 500 h/a * (1 t / 10^9 mg)

For the case studied here, the following specific assumptions are taken:

- installing an ESP allows achieving a dust concentration of 20 mg/Nm³;
- disposal of waste (fly ash);
- ESP material is carbon steel.

Combining the above with further standard assumptions (e.g. 500 full load hour equivalents), the total annualised costs amount to 471 664 \notin_{2015} per unit or 32 142 \notin_{2015} per tonne of PM reduced (TFTEI 2015).

The following graph (Figure 6.9) displays the specific costs per tonne of PM reduced over a range from 100 to 1500 full load hours per year, illustrating, once again, a steep cost increase at low utilisation rates.



Figure 6.9: Cost curve (quasi-marginal abatement costs) for installing an electrostatic precipitator at an oilfired power plant unit; dotted line at 500 h/a; based on own calculations

6.3 Discount rates used in case studies

In accordance with the scientific literature on discounting in CBA (e.g. Boardman et al. 2006, Harrison 2010, Pearce et al. 2006, Rabl et al. 2014) and the principles described in section 3.2.3, the following discount rates are adopted:

- Future annual costs and benefits are discounted at an equal social discount rate of 4% (European Commission 2009) given the European scope of the assessment.
- Using a social discount rate in the CBA is complementary to using a private interest rate for the private cost characterisation. Here, taking the perspective of a publicly owned utility, the weighted average cost of capital is assumed to be 6% by default. Due to the fact that this is rather at the low end of typical private interest rates in the energy sector (Capros et al. 2016), alternative assumptions are tested in a sensitivity analysis in section 7.3.3. The combination of a private interest rate and social discount rate is called the two-stage discounting procedure, further described by Kolb and Scheraga (1990).

6.4 Summary of chapter 6

The methodological framework for the assessment of health damages related to atmospheric emissions as well as the cost characterisation of emission control measures are described. The health damage cost assessment follows the stages of the impact pathway approach, i.e. definition of emission scenarios, exposure assessment, health impact assessment and monetary valuation. The complete modelling chain, its technical implementation, input data and related assumptions are transparently and comprehensively documented. Moreover, the emission scenarios and specific emission control costs used for the subsequent case studies are presented.

7 Application of the extended methodology framework and results

The newly developed methodology framework for the economic assessment of emission control measures at power plants with variable operation profiles is applied to several case studies. These are grouped according to their global objectives (Table 7.1).

The single cases are addressed through specific objectives that are respectively detailed using the following scheme: *What* (is the aim?), *Where* (i.e. at what scale are impacts assessed?), *For* (which emission scenarios?), and *Using* (which general and specific methodological assumptions?).

Research objectives and corresponding case studies		
Exploring the influence of emission patterns and atmospheric modelling features		
Emission intensity	7.1.1	
Background emission inventory	7.1.2	
Temporal modelling resolution	7.1.3	
Spatial modelling resolution	7.1.4	
Exploring the influence of methodological choices in health impact assessment		
Inclusion of NO ₂ -related health endpoints	7.2.1	
Alternative long-term exposure mortality impact assessment	7.2.2	
Lowering the concentration threshold for ozone-related health impacts	7.2.3	
Social CBA of emission control measures		
Primary and secondary emission control measures	7.3.1	
Comparison with simplified approaches for benefit assessment	7.3.2	
Sensitivity analysis regarding the social CBA	7.3.3	

Table 7.1: Overview on research objectives and corresponding case studies treated in the current chapter

To enable a consistent comparison of different power plant operation scenarios, the resulting health damage costs are expressed in \in per MWh of electricity generated.

In order to assess the effects that can be attributed to a specific power plant operation scenario, the respective health impacts are expressed relative to a so-called baseline scenario, i.e. without atmospheric emissions from the power plant. This baseline scenario is labelled 'S0' (cf. also Table 6.2).

7.1 Exploring the influence of emission patterns and atmospheric modelling features

This section serves to analyse the influence of different emission patterns on health damage costs, i.e. in terms of emission intensity and annual distribution. At the same time, by varying the spatial modelling resolution and by simulating a different temporal modelling resolution (constant versus variable emission pattern), it also permits to study the influence of atmospheric modelling features on health damage costs. It should be noted that nesting, i.e. integrating results from a sub modelling domain (e.g. France) into the parent modelling domain (e.g. Europe), was not carried out for the results presented hereafter. Adding the results from different scales would thus lead to an overestimation of the total health damage costs.

7.1.1 Emission intensity

Table 7.2 summarises the objective and approach to analyse the influence of emission intensity, i.e. the quantity of a pollutant emitted per time step. As shown in Table 6.2, there is a large difference in the annual emission quantities per pollutant between both scenarios compared here (> factor of 17).

Objective and research approach to assess the influence of emission intensity at source level on

h	health damage costs	
Objective	What is the influence on health damage costs of varying the emission intensity (emission quantity released per hour) at the source?	
Approach		
What	Assessing the influence of emission intensity on the health damage costs related to $PM_{2.5}$, PM_{10} and O_3 by comparing a constant operation scenario with normal and high emission intensity respectively	
Where	Île-de-France (IDF), France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4	
For	Scenarios S2b (normal emission intensity) and S2c (high emission intensity), com- pared to S0 (baseline), cf. Table 6.2	
Using	Standard modelling framework as described in section 6.1	

Table 7.2:

As illustrated in Figure 7.1, an increase in emission intensity of a point source leads to a decrease in health damage costs (per unit of electricity generated). At the EU modelling domain, this reduction amounts to 3%, whereas it is more pronounced at the France domain (- 15%) and at the Île-de-France domain (- 27%).



Figure 7.1: Annual health damage costs related to the normal emission intensity scenario (S2b) and the high emission intensity scenario (S2c) at different modelling domains

7.1.2 Background emission inventory

Table 7.3 summarises the objective and approach to analyse the influence of the background emission inventory on health damage costs. Such an inventory includes all natural and anthropogenic emissions considered during the atmospheric modelling. Globally, emissions of all relevant pollutants are expected to be lower in the future (year 2020) scenario than in the year 2009. The analysis was only conducted at the French modelling domain in order to limit the computational effort.

Table 7.3: O or	bjective and research approach to assess the influence of the background emission inventory n health damage costs
Objective	What is the influence on health damage costs of using an assumed future back- ground emission inventory as compared to the standard background emission in- ventory?
Approach	
What	Assessing the influence of the background emission scenario on the health damage costs related to $PM_{2.5}$, PM_{10} , and O_3 by comparing results for the standard (2009) and a hypothetical future (2020) background emission inventory
Where	France (FR), cf. Figure 6.3
For	Scenario S2c, compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1 with a variation in the background emission inventory, cf. Table 6.4

As shown in Figure 7.2, changing the background emission scenario leads to a minor (+ 3%) increase in health damage costs at the France modelling domain.



Figure 7.2: Annual health damage costs related to emission scenario S2c using the default background emission inventory (left) and a hypothetical future background emission inventory 2020 (right) at the France modelling domain

7.1.3 Temporal modelling resolution

Table 7.4 summarises the objective and approach to analyse the influence of the chosen temporal modelling resolution. In the given case, all atmospheric modelling results are calculated at an hourly basis. By evaluating a scenario with a constant hourly emission rate, the functioning of an averaged modelling approach that does not account for variations in emission patterns throughout the year is simulated. Such an average modelling is notably underlying the parameterised results of the EcoSenseWeb model (cf. section 5.5.2). This average modelling approach is compared to a more highly resolved modelling approach that accounts for variations in emissions throughout the year.

Table 7.4: 0	Objective and research approach to assess the influence of temporal resolution on health dam- age costs
Objective	What is the influence on health damage costs of using a higher temporal model- ling resolution, accounting for variable operation profiles of the emission source throughout the year?
Approach	
What	Assessing the influence of temporal modelling resolution on the health damage costs related to $PM_{2.5}$, PM_{10} , and O_3 by comparing a constant emission scenario to a variable emission scenario with equivalent overall emissions during one year
Where	Île-de-France (IDF), France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4
For	Scenarios S2 (variable operation) and S2b (constant operation), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1

Figure 7.3 shows the influence of using a different temporal resolution on health damage costs, all other input data being equal.



Figure 7.3: Annual health damage costs related to the variable emission scenario (S2) and the equivalent constant emission scenario (S2b) at different modelling domains

Compared to the variable emission scenario (S2), the constant emission scenario (S2b) leads to an increase in health damage costs by a factor of 1.69 at the European domain, whereas for the French (factor 2.93) and IDF domain (factor 4.11), the increase is even higher.

To better understand these differences, it is instructive to look at the concentration fields of annual mean $PM_{2.5}$ concentration, e.g. in France (Figure 7.4). Apart from a globally higher concentration increase in the constant emission scenario, the spatial distribution pattern is also different, reflecting further influences, for instance different wind directions throughout the year (Figure 7.5).



Figure 7.4: Change in annual ambient mean PM_{2.5} concentration due to the variable operation scenario S2 (left) and the equivalent constant emission scenario S2b (right) at the France modelling domain



Figure 7.5: Wind directions and wind speed simulated using the WRF model at emission release height over the France modelling domain during the variable operation scenario (left) and during the constant operation scenario (right); Source: Legorgeu (2016)

7.1.4 Spatial modelling resolution

Table 7.5 summarises the objectives and related approaches to analyse the influence of spatial modelling resolution, i.e. the resolution of the grid cells for which results are estimated.

Table 7.5:	Objectives and research approach to assess the influence of spatial resolution on health damage costs	
Objec- tives	 What is the influence on health damage costs of using a different spatial modelling resolution for a fixed spatial perimeter? 	
	2) What is the influence on health damage costs of using a so-called plume-in- grid modelling approach, i.e. a local Gaussian modelling embedded into the less resolved Eulerian modelling?	
Approad	h for objective 1)	
What	Assessing the influence of spatial modelling resolution on the health damage costs related to $PM_{2.5},PM_{10},and\;O_3$	
Where	Île-de-France (IDF): high (~3 km x ~5 km), medium (~17 km x ~25 km) and low (~35 km x ~50 km) modelling resolution, cf. Table 6.4 and Figure 6.3	
	France (FR): medium (~17 km x ~25 km) and low (~35 km x ~50 km) modelling res- olution, cf. Table 6.4 and Figure 6.3	
For	Scenario S2, compared to S0 (baseline), cf. Table 6.2	
Using	Standard modelling framework as described in section 6.1, adapted by using the age group fractions, baseline rates and long-term exposure mortality impact func- tion of the European modelling domain at all domains and thus allowing for a con- sistent comparison	
Approad	h for objective 2)	
What	Comparing health damage costs related to $PM_{2.5}$, PM_{10} , and O_3 based on a plume-in-grid modelling approach with those based on the standard modelling approach	
Where	Île-de-France (IDF), France (FR), cf. Figure 6.3	

- For Scenarios S1, S1PIG, S2 and S2PIG, compared to S0 (baseline), cf. Table 6.2
- Using Standard modelling framework as described in section 6.1

Figure 7.6 shows the influence of modelling grid resolution on health damage costs, all other elements being equal. For the modelling domain France, using a higher spatial resolution results in a 9% decrease in health damage costs. For the Île-de-France domain, the direction of change depends on the resolution: moving from a low to medium spatial resolution results in a 2% increase, whereas moving to a high modelling resolution results in a 3% decrease in health damage costs. Using a nesting approach at European level would thus slightly decrease the health damage costs compared to the standard approach without nesting.



Figure 7.6: Annual health damage costs related to emission scenario S2 at different modelling domains and assessed using different spatial modelling resolutions

Regarding the second objective, Figure 7.7 shows the influence of the plume-in-grid (PIG) modelling on the results.

In all analysed cases, the PIG modelling leads to a considerable increase in health damage costs, ranging from a factor of 1.75 (France domain) to a factor of 2.51 (Île-de-France domain). At the local scale, the effect is more pronounced than at the national scale.



Figure 7.7: Health damage costs caused by emission scenarios S1 and S2 using the standard or the plume-ingrid (PIG) modelling approach at the France and Île-de-France modelling domain

To better understand these results, dominated by PM_{2.5}-related health damage costs, the PM_{2.5} concentration changes at the local (IDF) domain are displayed using either a medium or high spatial resolution (Figure 7.8) and for both types of atmospheric modelling (i.e. with or without PIG, Figure 7.9). The colour ranges have been harmonised in order to allow for direct comparisons within each figure, however not across the two figures.

For the IDF modelling domain and the area around the emission source (highlighted by a white dot), a medium spatial modelling resolution leads to a slightly higher level of PM_{2.5} concentrations than the higher modelling resolution (Figure 7.8), partly explaining the small difference in health damage costs observed above.

Using the plume-in-grid modelling approach leads to a different pattern of concentration changes around the emission source as well as to remarkably higher concentration increases compared to the standard modelling approach (Figure 7.9). This explains the increases in health damage costs for the PIG modelling observed above.



Figure 7.8: Change in annual ambient mean PM_{2.5} concentration due to operation scenario S2 at the Île-de-France modelling domain using a medium (left) or high (right) spatial modelling resolution



Figure 7.9: Change in annual ambient mean PM_{2.5} concentration due to the variable operation scenario S2 using the standard (left) or plume-in-grid (PIG) modelling approach (right) at the Île-de-France modelling domain

7.2 Exploring the influence of methodological choices in health impact assessment

Apart from methodological choices in atmospheric modelling, the input data and parameter choices in health impact assessment have already been shown to substantially influence quantifiable health damage costs (cf. chapter 4). Complementary to these findings and following latest recommendations by the WHO (2013a), some additional elements shall be analysed here. Most importantly in view of recent findings, the influence of NO₂-related health impacts on overall quantifiable health damage costs is studied. Given the predominant role of mortality-related health damage costs, particularly related to long-term exposure to PM_{2.5}, an alternative quantification approach for this endpoint is tested. This is followed by a case study in which the concentration threshold for ozonerelated health impacts is varied.

7.2.1 Inclusion of NO₂-related health impacts

Table 7.6 summarises the objective and approach to analyse the influence of additionally quantifying NO₂-related health impacts. The scenarios S2 and S2b were chosen for illustrative purposes. The influence of including NO₂-related health impacts in social CBA is assessed in section 7.3.3.

.

...

. .

Table 7.6:	Dbjective and research approach to assess the influence of additionally quantifying direct NO_2 -related health effects
Objective	What is the effect on health damage costs of quantifying direct NO_2 -related endpoints in addition to PM and ozone-related endpoints?
Approach	
What	Including direct NO_2 -related endpoints in the health damage cost assessment
Where	Île-de-France (IDF), France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4
For	Scenarios S2 (variable operation) and S2b (constant operation), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1 and additionally ac- counting for NO ₂ -related health endpoints (Table 6.7), whilst multiplying the long- term NO ₂ exposure-related mortality impact function with a factor of 0.67 in or- der to account for potential double counting risks with regard to PM (WHO 2013a)

Figure 7.10 shows the absolute contribution of NO₂-related health damage costs at the three modelling domains under different operation scenarios. In addition, the relative contribution of the only substantial NO₂-related endpoint, i.e. long-term NO₂ exposure-related mortality risks (above an annual threshold of 20 μ g/Nm³), is displayed. NO₂-related health damage costs vary considerably across the modelling domains. Whilst their contribution amounts to only a few percent at European level, it rises to at least 14% at the French level and at least 25% at the Île-de-France level, depending on the emission scenario.



Figure 7.10: Annual health damage costs related to emission scenarios S2 (variable) and S2b (constant) showing the percentage contribution of NO₂-related mortality impacts due to long-term exposure (annual average above 20 μ g/m³) at different modelling domains

Interestingly, the absolute level of NO₂-related health damage costs is approximately equal at Île-de-France level and France level and lower at European level. This can be mainly explained by the fact that the largest share of NO₂-related impacts occurs only above an annual ambient concentration threshold of 20 μ g/m³. As shown in Figure 7.11, this threshold is only exceeded in the surroundings of the metropolitan area eastwards of the emission source. This means that a relatively small area is responsible for the quantified health damage costs at national level, explaining the quasi-parity of health damage costs at the local and national level. At the European scale, the ambient concentration increases at a lower absolute level (Figure 7.12), which can partly be explained by the coarser modelling resolution. Concentrations are thus diluted over a larger area. Even though some additional exposure occurs in the Benelux area and south-eastwards of France, the quantified health damage costs are lower overall than at the national level alone.



Figure 7.11: Change in annual ambient mean NO₂ concentration (above 20 μg/m³) due to operation scenario S2b at the Île-de-France (left) and France (right) modelling domain



Figure 7.12: Change in annual ambient mean NO₂ concentration (above 20 μg/m³) due to operation scenario S2b at the European modelling domain (excerpt of modelling domain)

7.2.2 Using an alternative approach for long-term exposure mortality impact assessment

As shown in chapter 4, the approach for assessing long-term PM_{2.5} exposure-related mortality impacts has a crucial influence, given the large share of damage costs that are regularly attributed to this single endpoint. As an alternative to quantifying years of life lost due to an increased mortality risk, the number of premature deaths are estimated here, following key international institutions (OECD 2012, U.S. EPA 2012c). Subsequently, these premature deaths are valued using the so-called VSL (Value of a Statistical Life, cf. also section 6.1.3.2). The analysis uses the following input parameters:

- The impact function is composed of the relative risk as recommended by WHO (2013a) and French and EU mortality baseline rates as given in Table A.4 (Appendix);
- For monetary valuation, the same VSL parameter as for all-cause infant mortality is used for all-cause mortality from long-term exposure, cf. Table 6.8, i.e. amounting to 1.9 million €2015 per premature death, net of income adaption.

Table 7.7 summarises the objective and approach to analyse the influence of the approach for long-term exposure mortality impact assessment. In section 7.3.3, the effect of using an alternative approach in an exemplary social CBA is assessed.

Table 7.7: C	Dejective and research approach to assess the influence of different approaches for long-term xposure mortality impact assessment
Objective	What is the impact of using an alternative approach to long-term $PM_{2.5}$ exposure mortality impact assessment?
Approach	
What	Assessing the influence of using the VSL approach instead of the VOLY approach for long-term exposure mortality impact assessment on health damage costs
Where	Île-de-France (IDF), France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4
For	Scenarios S2 (variable operation) and S2b (constant operation), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1, however using an al- ternative approach for long-term exposure mortality impact assessment and valu- ation, described above

Figure 7.13 shows the result of applying the alternative approach for mortality risk assessment. Depending on the modelling domain (and underlying different baseline mortality rates used), the VSL-based approach leads to an increase by a factor of 3.69 (European domain) and 3.09 (France and Île-de-France domain) respectively. The difference between the domains can be explained by the different mortality impact assessment approaches used (cf. section 6.1.3.2).



Figure 7.13: Annual mortality-related health damage costs due to long-term PM_{2.5} exposure of the emission scenarios S2 (variable) and S2b (constant) at different modelling domains and estimated using two alternative approaches

7.2.3 Lowering the concentration threshold for ozone-related health impacts

Instead of the standard metric for ozone impacts, i.e. SOMO35 (Sum Of (maximum daily 8h) Means Over 35 parts per billion), the alternative metric SOMO10 (Sum Of (maximum daily 8h) Means Over 10 parts per billion) is used, as recommended by the WHO (2013a) for sensitivity analyses.

Table 7.8 summarises the objective and approach to analyse the influence of a lower concentration threshold for ozone-related health effects. The scenarios were chosen for illustrative purposes, i.e. reflecting different operation patterns and different emission intensities. Objective and research approach to assess the influence of different concentration threshold lev-

e	els for ozone-related health effects	
Objective	What is the effect of assuming a lower effect threshold for assessing ozone-related health damages (SOMO10 instead of SOMO35)?	
Approach		
What	Assessing the influence of the concentration threshold above which ozone-related health effects are assumed to occur on ozone-related health damage costs	
Where	Île-de-France (IDF), France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4	
For	Scenarios S2, S2b and S2c, compared to S0 (baseline), cf. Table 6.2	
Using	Standard modelling framework as described in section 6.1, however considering ozone effects above a concentration threshold of 10 ppb (instead of 35 ppb)	

Table 7.8:

Figure 7.14 shows the influence of the alternative ozone metric on related quantified health damage costs.



Figure 7.14: Annual ozone-related health damage costs for emission scenarios S2 (variable), S2b (constant, normal emission intensity) and S2c (constant, high emission intensity) at different modelling domains and using two alternative concentration thresholds

Interestingly, assuming a lower concentration threshold leads to lower health damage costs for all scenarios and at all scales. This somewhat paradox outcome cannot be fully explained, as it depends on the non-linear processes underlying tropospheric ozone formation that are dependent on the emission source, its location and background conditions during the year (cf. section 2.4.3).

A closer look at the results reveals further particularities. For instance, different emission intensities lead to either positive (S2b) or negative (= avoided) health damage costs (S2c) at the IDF and FR modelling domain, whereas they lead to positive health damage costs for both scenarios at the EU modelling domain. For the variable emission scenario (S2) and at the EU modelling domain, switching from SOMO35 to SOMO10 induces a change from positive health damage costs to negative health damage costs.

7.3 Social cost-benefit analysis of emission control measures

The health damage cost assessment framework (section 6.1) is combined with the costing methodology (section 6.2) in order to analyse the efficiency of installing selected primary and secondary emission control measures at the exemplary power plant using social CBA. The objective is to demonstrate the feasibility and methodological particularities when carrying out social CBA at a specific industrial site.

7.3.1 Social CBA of primary and secondary emission control measures

Table 7.9 summarises the objectives and related approaches in order to conduct social CBAs of emission control measures.

Table 7.9:	Objectives and research approach to assess the private costs and health-related benefits of pri- mary and secondary emission control measures		
Objec- tives	1)	What are the private costs and health-related benefits of reducing NO_x and SO_2 emissions at a heavy fuel oil-fired power plant and are these measures socially efficient?	
	2)	What are the private costs and health-related benefits of additionally reduc- ing PM emissions and is this measure socially efficient?	

Approach for objective 1)

What	Assessing the health benefits (reduced damage costs related to $PM_{2.5}$, PM_{10} , and O_3) and associated private costs of primary NO_x and SO_2 emission control measures
Where	France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4
For	Scenarios S1 (initial emission level) and S2 (reduced emission level), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1 and private cost assessment described in sections 6.2.2.1 and 6.2.2.2

Approach for objective 2)

What	Assessing the health benefits (reduced damage costs related to $PM_{2.5}$, PM_{10} , and O_3) and associated private costs of a secondary PM emission control measure
Where	France (FR), Europe (EU), cf. Figure 6.3 and Figure 6.4
For	Scenarios S2 (initial emission level, including primary NO_x and SO_2 emission control measures) and S3 (reduced emission level), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1 and private cost assessment described in section $6.2.2.3$

The following parameter choices are made for the social CBAs carried out here:

- Benefits consist of avoided health damage costs due to emission control measures at the power plant;
- Costs consist of annualised investment and operating costs of the respective emission control measures (cf. section 6.2.2), estimated using a 6% weighted average cost of capital by default; an equipment lifetime of 20 years and 500 operating hours per year are assumed (cf. Table 6.9);
- Discount rate: Both future costs and benefits are discounted using a social discount rate of 4% (cf. section 6.3);
- Affected individuals: Given that the benefits of emission reductions occur within and beyond the French border, the analysis considers the population of the European modelling domain. In order to estimate the share of effects at national level, results for France are given in addition.

The health benefits of installing primary NO_x and SO_2 emission control measures, i.e. a low- NO_x burner and switching to heavy fuel oil with lower sulphur contents, can be derived from Figure 7.15.



Figure 7.15: Annual health damage costs related to the initial emission level (S1) and after installing primary emission control measures for NO_x and SO₂ (S2) at different modelling domains

Moving from emission scenario S1 to emission scenario S2 reduces health damage costs by 7.36 \notin_{2015} per MWh_{el} (Europe) and 1.71 \notin_{2015} per MWh_{el} (France). When multiplied with the annual electricity generation, i.e. 300 000 MWh per power plant unit, this leads to annual benefits of 300 000 * 1.71 = 513 000 \notin_{2015} (France) and 300 000 * 7.36 = 2 206 788 \notin_{2015} (Europe). Hence, approximately one fourth of benefits occur at the national scale.

The annualised private costs of the envisaged emission reduction measures amount to (cf. section 6.2.2):

- 1 145 538 €2015 per year and unit for installing the low NO_x burner;
- 4 706 280 €2015 per year and unit for switching to heavy fuel oil with a lower sulphur content.

Figure 7.16 compares the private costs and societal benefits at European level resulting from the installation of primary emission control per power plant unit, showing that costs exceed benefits by about 3.6 million ξ_{2015} per year under the given conditions. Even before entering the annual data into the CBA decision rule, thereby considering the complete project lifetime, the primary emission control measures can be classified as disproportionate, mainly owing to the high costs of reducing the fuel sulphur content. Applying the CBA decision rule (equation 3.1) results in a net present value of -49 537 143 ξ_{2015} , confirming that the investment is not socially efficient. Whilst it would be interesting to separately analyse the two primary emission reduction measures treated as a bundle here, the available emission scenarios do not allow such a separate analysis.

In order for annualised costs to break even with annual health benefits, the costs of fuel switching would have to decrease to a level of $1\ 061\ 250\ \epsilon_{2015}$ annually, all other elements remaining equal. This would imply marginal costs of $5\ 896\ \epsilon_{2015}$ per tonne of SO₂ removed and a substantial reduction when compared to the current default value (26 146 ϵ_{2015} per tonne of SO₂ removed). Alternatively, for a break even, the health benefits would need to rise to 19.5 ϵ_{2015} per MWh_{el} at the European scale.



Figure 7.16: Annualised costs of primary emission control measures and related benefits through avoided health damage costs per power plant unit and for the European modelling domain



The benefits of installing a secondary PM emission control measure, i.e. an electrostatic precipitator (ESP), can be derived from the health damage costs displayed in Figure 7.17.

Figure 7.17: Annual health damage costs related to the initial emission level with primary NO_x and SO₂ control measures installed (S2) and after installing an additional secondary emission control measure for PM (S3) at different modelling domains

The reduction in health damage costs amounts to around $0.26 \in_{2015}$ per MWh_{el}. Combining these with 300 000 MWh_{el} generated per power plant unit leads to annual benefits of 78 872 \in_{2015} . When compared to the annualised cost of the ESP, amounting to 471 664 \in_{2015} per unit and year (cf. section 6.2.2.3), costs exceed benefits by 392 792 \in_{2015} per unit and year (Figure 7.18). Applying the CBA decision rule (cf. equation 3.1) results in a net present value of -5 338 178 \in_{2015} , implying that the investment is not socially efficient.


Figure 7.18: Annualised costs of secondary emission control and related benefits through avoided health damage costs per power plant unit and for the European modelling domain

7.3.2 Comparison with simplified approaches for benefit assessment

The health damage cost assessment approach developed in this thesis enables an advanced consideration of the emission source characteristics and its location, however at the expense of a relatively high computational effort for atmospheric modelling (cf. also section 8.4). For this reason, the results obtained above shall be compared to two simplified approaches for benefit assessment:

- First, the benefit assessment is carried out using adapted EcoSenseWeb modelling results (including updated impact functions and monetary valuation in order to enable a better comparability with the standard social CBA results, cf. sections 4.3.1 and 5.2). This yields health damage costs that are site-specific to some extent but fail to account for variability in the operation pattern;
- Second, the benefits assessment is carried out using national average damage cost factors per quantity of pollutant emitted based on guidance by the European Commission (2006a) within the framework of BAT Reference documents (cf. section 2.4.6). These unit cost factors are dated (and could not be updated retrospectively) and account neither for the specific emission place nor for variability in the operation pattern.

Table 7.10 summarises the objectives and related approaches to compare different variants of the benefit assessment within the social CBA of emission control measures.

Table 7.10:	Objectives and research approach for the social CBA of primary emission control measures using simplified methods for health benefit assessment
Objec- tives	1) What is the outcome of the social CBAs on primary and secondary emission control measures when using the EcoSenseWeb model for the health benefit assessment?
	2) What is the outcome of the social CBAs on primary and secondary emission control measures when using unit damage cost factors from European Com- mission (2006a) for the health benefit assessment?
Approach	n for objective 1)
What	Assessing health benefits (reduced damage costs related to $PM_{2.5}$, PM_{10} , and O_3) and associated private costs of primary and secondary emission control measures
Where	Europe (EU28 + 11 non EU countries, larger than the Europe domain defined in section 6.1.1.2)
For	Scenarios S1 (initial emission level) as well as S2 and S3 (reduced emission levels), compared to S0 (baseline), cf. Table 6.2
Using	EcoSenseWeb model, adapted as described in section 4.2 in order to match the input parameters of section 6.1, followed by an adjustment for inflation and income growth (cf. section 6.1.4.2); private cost assessment, cf. section 6.2.2
Approach	n for objective 2)
What	Assessing health benefits (reduced damage costs related to $PM_{2.5}$, PM_{10} , and O_3) and associated private costs of primary and secondary emission control measures
Where	Europe (EU28 + 11 non EU countries, larger than the Europe domain defined in section 6.1.1.2)
For	Scenarios S1 (initial emission level) as well as S2 and S3 (reduced emission levels), compared to S0 (baseline), cf. Table 6.2
Using	Unit damage costs as described in (Holland et al. 2005c), adjusted regarding infla- tion and income growth between the year 2000 and 2015 (cf. section 6.1.4.2); pri- vate cost assessment, cf. section 6.2.2

For objective 1, dedicated calculations using the model EcoSenseWeb have been carried out. The input data is consistent with the case presented in section 6.1.1. To allow for a better comparison, health impact functions (including age group and risk group fractions) and monetary valuation parameters have been harmonised following the approach described in chapter 4. This means that the remaining differences in resulting health damage costs are due to:

- Dispersion modelling: The atmospheric model used in EcoSenseWeb is a parameterised version of the EMEP model that does not account for variable operation profiles. Moreover, different background meteorology and emission inventory data is used compared to the modelling framework described in section 6.1;
- Exposure modelling: Given that the modelling domain in EcoSenseWeb is larger than the Europe modelling domain used before, more people are considered for the health impact assessment. Moreover, the statistics on population distribution and numbers are older than those used in the modelling framework described in section 6.1.

For objective 2, unit damage cost factors given in European Commission (2006a) include impacts on human health and crops. To allow for a better comparison and using back-ground information by Holland et al. (2005c), only the human health-related damage costs are included here, amounting to:

- 7 200 €₂₀₀₀ per tonne of NO_x emitted in France, translating into 10 475 €₂₀₁₅ after inflation and income adaption as described in section 6.1.4.2;
- 8 035 €2000 per tonne of SO2 emitted in France, translating into 11 689 €2015 after inflation and income adaption as described in section 6.1.4.2;
- 44 000 €2000 per tonne of PM2.5 emitted in France, translating into 64 011 €2015 after inflation and income adaption as described in section 6.1.4.2.

The differences as compared to the approach described in section 6.1 are as follows:

• Dispersion modelling: The results are based on a parameterised version of the EMEP model that does not account for variable operation profiles. This modelling is similar to the one underlying EcoSenseWeb, however it uses an older version of the EMEP model. Background meteorology and emission inventory data differ as well;

- Exposure modelling: Given that the modelling domain is larger than the Europe modelling domain used before, more people are considered for the health impact assessment. Moreover, the statistics on population distribution and numbers are different from those in section 6.1;
- Health impact assessment: As data harmonisation is not possible here, results differ in terms of considered health endpoints and respective impact functions;
- Monetary valuation: No harmonisation of valuation factors was carried out, apart from adjusting the values for inflation and income growth. Particularly for the valuation of mortality risks due to long-term exposure towards PM, a 20% higher valuation factor is used compared to the approach described in section 6.1.

For the benefit assessment the unit damage cost factors (as indicated above) are multiplied by the amount of tonnes of pollutant avoided through the emission control measures (180 tonnes per year of NO_x and SO₂ and 15 tonnes of $PM_{2.5}^{26}$).

The above described approaches lead to the results displayed in Figure 7.19 (primary emission control measures) and Figure 7.20 (secondary emission control measure).

In case of installing the primary emission control measures, both alternative approaches lead to an increase in health damages per tonne of pollutant emitted. Therefore, the societal benefit of reducing emissions is higher when compared to the standard modelling approach ('own assessment'). The highest benefit is observed when relying on the unit damage cost factors by European Commission (2006a). Yet, the private costs of emission control remain globally higher than the associated benefits for all of the approaches compared here and the conclusion of the social CBA does not change, i.e. the pollution control measures are not socially efficient.

For the secondary PM emission control measure, using the simplified assessment approaches again leads to substantially increased societal benefits. When relying on the unit damage cost factors by European Commission (2006a), the investment can be classified as socially efficient according to the social CBA. This is mainly due to the high damage costs per tonne of PM_{2.5} emitted (cf. above). Given that a national average unit damage cost factor is used that does not account for the high height of emissions from the oil-fired power plant, it is likely that this approach overestimates the benefits of emission reduction in this case.

²⁶ It is assumed that the 18 tonnes of PM avoided consist of 84% PM_{2.5} (Dreiseidler et al. 2000).



Figure 7.19: Annualised costs of primary emission control measures and related benefits through avoided health damage costs per power plant unit and for the European modelling domain using different approaches for benefit assessment



Figure 7.20: Annualised costs of secondary PM emission control and related benefits through avoided health damage costs per power plant unit and for the European modelling domain using different approaches for benefit assessment

7.3.3 Sensitivity analysis

In order to demonstrate the influence of key modelling and parameter choices on the social CBA outcome, a sensitivity analysis is carried out. Given that the methodological choices are interdependent, a step-wise procedure is followed. Based on experience from chapter 4 and sections 7.1 and 7.2, specific methodological choices are analysed. Table 7.11 summarises the objectives and related approaches.

 Table 7.11:
 Objectives and research approach for the sensitivity analysis of input parameters used in the social CBA of primary emission control measures

1)	What is the outcome of the social CBA on primary emission control measures
	when varying key modelling choices regarding the benefit assessment?
2)	What is the outcome of the social CBA on primary emission control measures when varying key parameters in the emission control cost assessment?
	1) 2)

Approach for objective 1)

What	Assessing health benefits and private costs of primary NO _x and SO ₂ emission control measures, when including additional NO ₂ -related health impacts and when varying the long-term exposure mortality impact assessment (benefit side)
Where	Europe (EU), cf. Figure 6.4
For	Scenarios S1 (initial emission level) and S2 (reduced emission level), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1 and private cost assessment described in sections 6.2.2.1 and 6.2.2.2, varying modelling choices for the benefit assessment, described above

Approach for objective 2)

What	Assessing health benefits and private costs of primary NO_x and SO_2 emission control measures, when varying the equipment lifetime and the weighted average cost of capital (cost side)
Where	Europe (EU), cf. Figure 6.4
For	Scenarios S1 (initial emission level) and S2 (reduced emission level), compared to S0 (baseline), cf. Table 6.2
Using	Standard modelling framework as described in section 6.1 and private cost assessment described in sections 6.2.2.1 and 6.2.2.2, varying input parameters for the cost assessment, described above

As results are similar for the primary and secondary emission control measures, only the former are presented here.

The following adjustments are carried out regarding the benefit assessment:

- inclusion of direct NO2-related health impacts;
- use of a VSL-based approach for the assessment of long-term exposure mortality impacts (both due to PM_{2.5} and NO₂).

Other important influencing factors have been disregarded for the following reasons:

- Using the plume-in-grid modelling instead of the default modelling (cf. section 7.1.4) was disregarded given that the results are not available at the European domain and that the confidence in the results is not necessarily higher than for the standard modelling (cf. section 8.1.1);
- Using the constant instead of the variable emission scenario (cf. section 7.1.3) was likewise disregarded because such a constant scenario is not realistic for a peak load power plant.

When using the annuity method (cf. section 3.3.1) for the private cost assessment, the annuity is defined by the capital recovery factor that in turn depends on the project life-time and interest rate (weighted average cost of capital), as displayed in Figure 7.21.



Figure 7.21: The capital recovery factor of the annuity method as a function of project lifetime and weighted average cost of capital for typical ranges of values

Short project lifetimes lead to a substantial increase of the capital recovery factor. As a consequence, annuities increase with shorter project lifetime and with higher weighted average costs of capital. On this basis and by considering further information, the following parameter choices have been made for the sensitivity analysis regarding the private cost assessment:

- Equipment lifetime: The annuities of the low NO_x burner are recalculated using an assumed life time of 10 years (instead of 20 years);
- Weighted average cost of capital: Recommendations given in a recent European reference document (Capros et al. 2016) for 'companies in competitive energy supply markets' are followed, leading to a weighted average cost of capital of 8.5% (instead of 6%).

The cost of fuel switching for SO_2 emission reduction includes only a variable component and is therefore independent from the assumed equipment lifetime and the weighted average cost of capital.



Figure 7.22 shows the result of applying the above listed sensitivity assumptions.

Figure 7.22: Annualised costs of primary emission control measures and related benefits through avoided health damage costs per power plant unit and for the European modelling domain using different assumptions and parameters for the cost and benefit assessment

On the cost side, a decrease of the project lifetime as well as an increase of the weighted average cost of capital both lead to an increase of annualised costs. On the benefit side, considering direct NO2-related health benefits leads only to a slight increase of benefits, whereas switching to a VSL-based approach for the assessment of long-term exposure mortality risk assessment leads to a very substantial increase in health benefits. In the latter case, the CBA would yield a positive net present value for all sensitivity cases analysed for the private cost assessment.

7.4 Summary of chapter 7

The results of several case studies are presented, providing answers to the following overarching objectives: exploring the influence of 1) emission patterns and atmospheric modelling features, and 2) of methodological choices in health impact assessment, as well as 3) demonstrating the application of the methodological framework for social cost-benefit analysis of emission control measures at site level. Using deterministic modelling approaches, the most remarkable observed influence on quantified damage costs concerns the temporal emission pattern (and hence temporal modelling resolution), the use of a plume-in-grid modelling approach, and the approach for long-term exposure mortality impact assessment. Neither the primary nor the secondary emission control measures are socially efficient according to the default assessment following the methodology developed within this thesis. However, using an alternative mortality assessment approach or relying on simplified benefit assessment approaches would, in some cases, alter the outcome of the social cost-benefit analysis.

8 Discussion

The first section of this chapter discusses the case studies of chapter 7 in view of providing further contextual information (cf. section 8.1). Apart from demonstrating the feasibility of social cost-benefit analysis (CBA) at the level of one individual industrial emission source, another merit of these case studies is to provide insights into the influence of methodological choices on health damage costs and, more generally, on the results of social CBAs. As these insights relate to the topic of uncertainty, a large part of this chapter is devoted to uncertainty underlying health damage costs: first, existing evidence on uncertainty is reviewed and put into perspective (cf. section 8.2.1); second, a quantitative (cf. section 8.2.2) and qualitative (cf. section 8.2.3) uncertainty characterisation is presented, largely based on the case studies of this thesis; last but not least, the implications of uncertainty on decision-making in the business and public policy context are discussed (cf. section 8.3). A critical appraisal of the developed methodology beyond uncertainty considerations serves to identify future research opportunities (cf. section 8.4). Based on the experience with transferring the social CBA method from the public to the private context, a set of methodological recommendations for private decision-makers concludes this chapter (cf. section 8.5).

8.1 Discussion of the case study results

Following the structure of chapter 7, the results of the case studies are discussed. The influence of methodological choices on health damage costs is discussed mainly in terms of the underlying mechanisms and put into perspective with related existing work. For the exemplary social CBA, the discussion focuses on the relevance and the reliability of the results.

8.1.1 Influence of atmospheric modelling features

Globally, it is difficult to generalise findings that are related to atmospheric modelling, given the non-linearity and thresholds involved in atmospheric chemistry. Moreover, as discussed below, the modelling results depend on the specific emission source, time, place, and background conditions.

In the specific cases analysed in chapter 7, all results are influenced by the fact that the emission source is located near a large metropolitan area (i.e. the city of Paris), characterised by a high population density, and surrounding areas with a lower population density (Figure 6.6). This spatial variability in population density has a crucial influence on population exposure and resulting health damage costs. It also implies that the sometimes substantial differences due to different modelling features at the local level (Île-de-France domain) would be less pronounced in the case of a less densely populated area near the emission source. Apart from higher population densities, metropolitan areas are typically also characterised by higher background emission levels, particularly due to road traffic. This has an influence on NO₂-related impacts and on the formation mechanisms of secondary pollutants, notably secondary PM or ozone, further discussed below.

Emission intensity

The differences in health damage costs due to different emission intensities were moderate (i.e. ranging from 3% at the Europe domain to 27% at the Île-de-France domain, cf. section 7.1.1). Evidence exists that numerical modelling uncertainty reduces with an increasing emission intensity (Brandt et al. 2012, Tarrasón 2009). As a consequence, the health damage costs per MWh for the constant emission scenario at higher emission intensity can be considered more robust than those at lower emission intensity.

Background emission inventory

Using a different background emission inventory at national level only had a minor influence on the estimated health damage costs (cf. section 7.1.2). Yet, this result strongly depends on the place of emission as well as the substances concerned (those emitted and those already present in the background), especially with regard to the formation of secondary PM.

The robustness of results at local level could be further improved by increasing the spatial resolution of the background emission inventory. In the modelling framework described in section 6.1 and used for the case study, the inventory has a resolution of 50 km, complemented by some additional information derived from land use data. Work by Fountoukis et al. (2013) on the Paris metropolitan region, albeit using a different air quality model than the one in this thesis, has shown that increasing the spatial resolution of the emission inventory leads to larger spatial concentration gradients, thereby improving the precision of exposure estimates.

Temporal modelling resolution

The average modelling approach (constant operation) leads to considerably higher health damage costs than the more highly resolved modelling approach (variable operation) for the case study presented in section 7.1.3. As the annual emission quantities are equal in both scenarios, the underlying causes can be generally attributed to the atmospheric modelling, including the influence of meteorology and background emissions. These lead to a considerably different population exposure for which three reasons are identified.

An analysis of exposure data under the constant emission scenario reveals a seasonal influence regarding PM formation. Notably, under the given conditions and when comparing monthly average exposure data, the same quantity of power plant emissions leads to more pronounced ambient PM_{2.5} concentration increases during the summer months than during the winter months²⁷. Combined with the fact that, under the variable emission scenario, the power plant operates mainly during the winter months (Figure 6.2), this helps to partly explain the observed differences between the two operation scenarios.

Second, looking at the wind directions for both scenarios (Figure 7.5) reveals that the constant emission scenario is characterised by prevailing westerly winds, whereas wind directions are more diverse in the variable emission scenario. Given that a large metropolitan area is situated eastwards of the emission source, this partly explains the higher damage costs observed in the constant emission scenario, particularly at the Île-de-France domain.

Another, though minor part of the differences between the constant and variable emission scenario is explained by differences in ozone-related health damage costs. Tropospheric ozone impacts are assumed to occur only above a given concentration threshold. In addition, ozone forms in a non-linear reaction mainly during the summer season. As a result, the operating time of the emission source as well as the meteorological and background emission conditions during the year are relevant influencing factors. In the given case, this leads to lower ozone-related health damages in the variable emission scenario, in which emissions mainly occur during the winter season (Figure 6.2). According to the European Environment Agency²⁸, the meteorological year 2009, used for the modelling, was characterised by very low summer ozone concentrations compared to other years.

²⁷ Results available upon request.

²⁸ <u>http://www.eea.europa.eu/highlights/summer-ozone-record-low-concentrations-in-2009</u>, last accessed: 2017-05-18

Spatial modelling resolution

The case study presented in section 7.1.4 shows that the direction of change in health damage costs with regard to the spatial modelling resolution depends on the analysed zone as well as the resolution itself: increasing the spatial resolution at the France domain (from 35 km x 50 km to 17 km x 25 km) led to a 9% reduction in health damage costs. At the Île-de-France domain, the same increase in resolution slightly increased health damage costs, whereas moving to a high (3 km x 5 km) resolution slightly reduced health damage costs. The latter can be explained by the fact that the high modelling resolution leads to lower concentration levels surrounding the emission source, where large shares of the exposed population are located (cf. Figure 7.8). A similar explanation applies to the France domain, where the averaging of concentrations over a larger area using a lower modelling resolution leads to a higher exposure level in the metropolitan area (results not included graphically in this thesis).

These findings are partly consistent with further work on the same topic. Likhvar et al. (2015) evaluated the influence of spatial modelling resolution on PM_{2.5} and ozone-related health impacts at a European (50 km resolution) and Île-de-France (4 km resolution) domain. However, the analysis differs from the case presented in this thesis in two central points: First, a different air quality model (CHIMERE) was used. Second, Likhvar et al. (2015) assessed the impacts of large-scale emission changes assumed to occur under future European energy scenarios rather than of specific point emission sources. The results (ibid., p. 447) indicate that "working at the IdF [Île-de-France] scale, the results [for PM_{2.5}] were more than 20% larger than those estimated at the European scale for the same area of interest. Here the discrepancy can be mainly due to the resolution change." Moreover, it was found that the effect of modelling resolution on ozone-related impacts was more pronounced than for PM_{2.5} and that ozone-related effects were strongly influenced by the presence of a large metropolitan area.

Thompson et al. (2014) analysed the effects of different emission reduction scenarios on PM_{2.5} and ozone-related health impacts in 9 US regions at 36, 12, and 4 km modelling resolution. Here again, a different air quality model (CAMx) as well as different emission source characteristics (i.e. evaluating emission changes between the year 2005 and 2014) were used. Regarding PM_{2.5}, it is found that the influence of spatial modelling resolution on health impacts is different for primary and secondary PM species. While a higher resolution generally resulted in higher impacts due to primary PM, the direction of change for secondary PM depended on the specific region. Overall, PM_{2.5} related impacts were likewise found to be less sensitive towards spatial modelling resolution than ozone-related impacts.

Increasing the spatial resolution through a plume-in-grid modelling approach led to remarkably higher concentration levels when compared to the standard modelling approach (cf. section 7.1.4). Yet, it is not possible to state which of the two modelling approaches performs globally better, as the precision of the respective modelling approaches depends on the specific emission scenario and the modelling parameterisations used (Kim et al. 2014). In order to assess the model performance, modelled concentrations using both approaches would need to be compared to surface monitoring data. This was out of scope of the current thesis.

Other findings concerning the plume-in-grid modelling of this thesis are consistent with observations made by Kim et al. (2014). First, plume-in-grid impacts are less pronounced at coarser modelling resolutions due to a higher dilution of the pollution plume. Second, under a plume-in-grid approach, effects from point sources tend to be lower near the point source and more pronounced further downwind from the source (cf. Figure 7.9).

8.1.2 Influence of methodological choices in health impact assessment

Methodological choices related to health impact assessment influence the magnitude of health damage costs (cf. section 7.2). Similarly to atmospheric modelling features, differences in results due to quantifying direct NO₂-related health impacts and due to changing the ozone concentration threshold depend on the emission source characteristics and the population distribution. By contrast, the differences in results observed due to different mortality impact assessment approaches are independent from source characteristics and population distribution and are therefore more generally valid.

Inclusion of NO2-related health impacts

Like for PM_{2.5}-related endpoints, the increase in mortality risk due to long-term exposure above a concentration threshold of 20 μ g of NO₂/m³ has the highest influence on health damage costs among all NO₂-related health endpoints (Figure 7.10). For this reason the choice of the threshold is of crucial importance. Even though the 20 μ g/m³ threshold is officially recommended by the WHO (2013a), it is nonetheless discussed as potentially being too high, given that health effects have also been shown to occur at lower concentration levels (Héroux et al. 2015, Walton et al. 2015). Keeping in mind the quite limited areas for which the threshold was actually exceeded in section 7.2.1 (cf. Figure 7.12), it is clear that decreasing the threshold would lead to a potentially important increase in health damage costs. Another important influencing factor related to this concentration threshold is the proximity of a large metropolitan area. As such areas are typically characterised by a high volume of road traffic that causes substantial NO₂ emissions, the probability of exceeding a threshold of 20 μ g/m³ is much higher, as illustrated by the case study result (cf. Figure 7.11).

Still related to the threshold, the spatial modelling resolution also plays a decisive role. The higher this resolution, the higher the chances to capture local exceedances of thresholds, as concentrations get more diluted when being averaged over a larger area. This effect can be observed for the European modelling domain. The concentration increase estimated over the metropolitan area is lower when compared to the higher resolution modelling at the Île-de-France modelling domain (cf. section 7.2.1). The effects at European level thus tend to be underestimated in the presented case. Indeed, measurement data for 2009^{29} shows local exceedances of the $20 \,\mu\text{g/m}^3$ concentration threshold at many places all over Europe, while these exceedances are not identified by the atmospheric modelling at the European domain. As a consequence, in order to obtain more precise results at European level, a higher modelling resolution would be needed for countries other than France. This is however not deemed necessary, given that the exemplary emission source is located quite centrally within France. Therefore, the majority of the total effects is expected to be captured by the higher modelling resolutions at the Île-de-France and France domains.

In addition to the questions surrounding the appropriate effect threshold, there is a risk of double counting related to long-term exposure mortality impacts, if effects due to PM_{2.5} exposure and additionally due to NO₂ exposure are estimated. According to WHO (2013a), the overlap is said to be in the range from 0 to 33%. For this reason and taking a rather conservative approach, only 67% of the direct NO₂-related mortality health damage costs were considered in section 7.2.1 and in section 7.3.3.

Using an alternative approach for long-term exposure mortality impact assessment

The way in which long-term exposure mortality risks are assessed is the single most influential factor within health damage costs assessment. It includes the choice of the approach as such (i.e. VOLY versus VSL) and of the valuation parameter to be used (i.e. how to value a VOLY or VSL).

²⁹ See <u>http://www.eea.europa.eu/themes/air/interactive/no2</u>, last accessed: 2017-05-18

As shown in section 7.2.2, using a VSL-based approach instead of a VOLY-based approach increases related health damage costs by at least a factor of 3.1. This outcome also depends on the health baseline rates and age group fractions used, explaining the differences between the French and European modelling domain. At the same time, the central VSL estimate of 1.9 million ξ_{2015} (net of income adaption) used, can be characterised as rather conservative when compared to alternative values. For instance, the OECD (2012) proposes a value of 3.6 million US ξ_{2005} for the EU context; the U.S. EPA proposes a central VSL estimate of 7.9 million US ξ_{2008} for policy analysis (U.S. EPA 2010). This shows an important discrepancy between what is currently recommended at both sides of the Atlantic with potentially important policy implications.

The impact of using different VOLY estimates on health damage costs caused by an exemplary power plant has been analysed in section 4.4.1. An increase of this parameter by about 30% leads to an increase in overall health damage costs of about 20%. Although this difference does not seem substantial, it may be sufficient to shift the binary efficiency decision criterion based on a social CBA in either direction.

Lowering the concentration threshold for ozone-related health impacts

Ozone-related health impacts are strongly influenced by the presence of large agglomerations (Likhvar et al. 2015, Markakis et al. 2014, Thompson et al. 2014). Owing to the nonlinearity in ozone formation, ozone-related health impacts as estimated in section 7.2.3 additionally depend on the background emission inventory, meteorology, the spatial and temporal modelling resolution, and, as illustrated, the chosen threshold level. The strong sensitivity of results with regard to changes in concentration threshold and modelling domain also imply a generally higher uncertainty of ozone-related health damage costs. Given however that ozone-related health damage costs are minor when compared to those related to PM or NO₂, the effect of these uncertainties on decision-making is limited.

8.1.3 Social CBA of emission control measures

Regarding the practical relevance of the social CBA results (cf. section 7.3) for private decision-making, it should be noted that the emission levels obtained after retrofitting emission control measures (cf. Table 6.2) comply with requirements of the Industrial Emission Directive (IED, cf. section 2.4.6), applicable in the year 2016. For power plants operating less than 1500 full load hours per year, however, derogations are foreseen within Annex V of the IED (European Parliament and Council of the European Union 2010), resulting in less stringent emission limit levels than those assumed in the social CBA. A further tightening of emission limit values in the future will result from the recently concluded revision of the BREF on large combustion plants (cf. section 2.4.6). Due to uncertainty regarding the national implementations and potential flexibilities, the social CBA carried out in section 7.3 is to be seen as a proof of concept rather than being concerned with decisionmaking under realistic conditions.

While both the primary and secondary emission control measures assessed were qualified as disproportionate using the highly resolved modelling approach implemented in this thesis, the outcome for the secondary control measure changed when using a simplified approach for the benefit assessment (cf. section 7.3.2). As already mentioned, the unit cost factors for PM_{2.5} in the simplified approach do neither account for the specific location and height of the emission source nor for variability in the operation pattern. Moreover, given that they were derived from an older impact pathway implementation with dated risk and valuation parameters, they cannot be considered a robust choice, even though they stem from an official European Commission reference document.

The sensitivity analysis revealed that changing parameters for the private cost assessment had a relatively limited influence, whereas changes in the benefit assessment methodology were more influential. While the inclusion of NO₂-related health impacts had a fairly limited impact on the overall benefits, switching from a VOLY- to a VSL-based mortality impact assessment altered the final outcome of the social CBA.

For a critical appraisal of the social CBA methodology, refer to section 8.4.3.

8.2 Uncertainties in health damage cost assessment: quantitative and qualitative evidence

This section serves to present existing evidence regarding the uncertainties underlying health damage cost assessment in general as well as quantitative and qualitative insights generated through the case studies within this thesis.

8.2.1 Existing evidence on uncertainties underlying health damage cost assessment

As introduced in section 3.4.3, the assessment of health damage costs based on the impact pathway approach is subject to various types of uncertainty. In the European research context, particularly related to the ExternE project series, a few comprehensive uncertainty assessments exist:

- Rabl and Spadaro (1999), Spadaro and Rabl (2008): The central statistical theorem states that, given the multiplicative chain underpinning the impact pathway approach (cf. equation 3.3), the resulting damage costs are lognormally distributed. Under this condition, the 68% confidence interval around the geometric mean (μ_g) can be conveniently expressed using the geometric standard deviation (σ_g), i.e. spanning a range of [$\mu_g * 1/\sigma_g$; $\mu_g * \sigma_g$]. For instance, the following estimates for σ_g are given for mortality-related health damage costs due to directly emitted PM (Spadaro and Rabl 2008):
 - Exposure modelling (including dispersion, chemical transformation and background emissions): $\sigma_g = 1.5$ for non-reactive primary pollutants (for secondary PM in the form of sulfate or nitrate σ_g amounts to respectively 1.76 and 1.9);
 - Health impact assessment (including relative risk, PM component toxicity, quantification of YOLL): σ_g = 1.88;
 - Monetary valuation of YOLL: $\sigma_g = 2$ for valuation based on WTP from surveys (for market-based valuations σ_g is assumed to be in a range of 1.1 to 1.3).

These assumptions lead to a σ_g of 2.78 for mortality-related health damage costs due to direct PM emissions. For emissions of SO₂ and NO_x, as precursors for secondary PM-related mortality impacts, σ_g amounts to respectively 3.42 and 3.55. Based on this and some further results, the authors (ibid.) conclude that inhalation-related health damage costs due to emissions of classical air pollutants can be characterised by a σ_g of approximately 3.

It should be noted that the algebraic models underlying these uncertainty estimates are relatively simple and not comparable to the atmospheric models used in this thesis and that some of the assumptions taken, e.g. on differing PM component toxicity for secondary PM, are not consistent with latest WHO recommendations (cf. section 3.4.1). Yet, these results are useful in that they provide order of magnitude values of uncertainty ranges for central parameters during different assessment stages and for different substances.

Holland et al. (2005b): This relatively broad uncertainty assessment was carried out in the frame of the social CBA underpinning the Clean Air For Europe (CAFE) programme, i.e. an evaluation of air quality legislation at the European level. It includes a probabilistic uncertainty assessment using Monte Carlo analysis, sensitivity analysis and a qualitative discussion of possible biases. For instance, based on probabilistic Monte Carlo analysis, the 95% confidence interval of the so-called aggregate damage function³⁰ for PM-related health impacts is found to range from approximately a factor of 2.5 below to a factor of 1.7 above the mean value.

A critical step during Monte Carlo analysis is the definition of input data (parameter estimates and corresponding probability distribution functions), given that the results are directly impacted by these assumptions. In the present case (ibid.), these assumptions were partly taken on subjective grounds and should therefore be interpreted with caution. The sensitivity analyses address similar elements as this thesis, notably the approach for mortality impact assessment or different effect thresholds for ozone, however at a more aggregated level. Further on, the influence of the choice of the meteorological year is addressed: results indicate that the variation in exposure due to differences in meteorology at the national level are up to 50%, but are reduced when taking a pan-European perspective.

Although building upon the impact pathway approach, the parameter choices and further assumptions used for these uncertainty analyses are outdated compared to the current thesis.

Holland (2014b): Closely related to the HRAPIE project (WHO 2013a) and the recent European Clean Air Policy Package (cf. section 2.4.5), this report includes a rather brief and incomplete quantitative and qualitative uncertainty assessment of health-related damage costs. The 95% confidence intervals around the central HRAPIE health risk parameters as well as subjective indications of confidence intervals and probability distributions for population at risk and incidence rates are presented. Moreover, based on the assumed confidence intervals of alternative

³⁰ The aggregate damage function represents the health damage costs per person and ambient pollutant concentration increment, thus excluding exposure modelling-related uncertainties.

monetary valuation factors for mortality impacts, an uncertainty framework for social CBA is presented that aims at estimating the probability of benefits exceeding costs. On a qualitative level, a bias assessment for unquantified uncertainties is also included.

Regarding the central parameters used for health impact assessment and monetary valuation, most choices correspond to the ones used in this thesis. By contrast, Holland (2014b) uses a higher default monetary valuation for mortality-related impacts compared to this thesis (cf. also section 4.3.5) and disregards atmospheric modelling-related aspects.

8.2.2 Quantitative uncertainty characterisation based on the results of this thesis

Health damage costs caused by atmospheric emissions from point emission sources are influenced by numerous influencing factors, as demonstrated in this thesis. The variation in health damage costs brought about by changing specific influencing factors can be compared on a quantitative basis and delivers new insights for decision-making. In terms of atmospheric modelling and particularly when assessing one individual emission source, the analyses cover more aspects than the existing uncertainty analyses discussed above.

In order to capture the influence of different emission source characteristics more broadly, Figure 8.1 reproduces recent work carried out in the frame of updating the German methodological convention on environmental costs (van der Kamp et al. forthcoming). The associated health damage costs are based on the EcoSenseWeb model and have been updated to reflect recent recommendations in the same way as described in section 4.2 (using the WHO HRAPIE recommendations for impact assessment and monetary valuation according to the "Year 2013" implementation, however updating the monetary base year to 2015).



Figure 8.1: Health damage costs per tonne of substance emitted in 4 German sub regions (DE1 – DE4) and per height of release; adapted from van der Kamp et al. (forthcoming)

In terms of their generalisability, the influencing factors on health damage costs, presented below (Table 8.1), can be distinguished into two groups:

- Input data and methodological features related to atmospheric modelling: Input data includes emission source characteristics, such as location, height of release, and emission pattern that, in conjunction with background emission inventory and meteorological data, determine the variations in ambient pollutant concentrations. Due to the non-linear processes underlying atmospheric chemistry, it is generally not possible to derive general conclusions from related analyses, but rather to present the range of possible results per setting (cf. Figure 8.1). Choices regarding spatial and temporal resolution have an influence on the precision of the results, but the direction and size of their influence depends likewise on the specific case and cannot be generalised (cf., for instance, section 7.1.4).
- Elements that are related to the remaining steps of the impact pathway approach, i.e. exposure modelling, health impact assessment, and monetary valuation. Given the linear assessment chain used for the implementation in this thesis (equation 3.3), the influence of these elements on the results can be generalised.

8 Discussion

The bulk of analyses carried out within this thesis concerns the first group, i.e. the atmospheric modelling inputs and features.

To characterise the sensitivity of health damage costs with regard to the variation per influencing factor, relative differences are calculated, i.e. dividing a (higher) reference value by a (lower) baseline value. Given however that these sensitivities refer to results that have been calculated using different models and methodologies, they cannot be directly compared. Rather, the intention is to provide order of magnitude values for the influence of single elements (input data and methodological choices) on health damage costs. For reasons of completeness, Table 8.1 also contains influencing factors for which no specific analysis has been carried out in the frame of this thesis.

Next to presenting the outcome of the sensitivity analyses, Table 8.1 also indicates which types of uncertainty are underlying the respective influencing factor, following the classification of section 3.4.3.

 Table 8.1:
 Overview on influencing factors, underlying types of uncertainty and sensitivity of health damage costs with regard to changing the influencing factors; based on case studies within this thesis or related work

Influencing factor	Varia bility	Parameter uncertainty	Model uncertainty	Decision-rule uncertainty	Sensitivity of health damage costs with re- gard to changes in the influencing factor	Reference
	Varia	Parai	Mod	Decis		

Atmospheric modelling: emission source characterisation and modelling features

Emission source: geographic location	x	х	x	Increase of a factor of 1.8 when locating a power plant (high height of release) at a densely populated area compared to a remote location (low population density); European assessment domain	section 5.4.2
	x	х	x	Differences of up to a factor of 2.49 between different regions in Germany for direct PM _{2.5} emissions at a low height of release (regional modelling); European assessment domain	Figure 8.1

Influencing factor	Variability	Parameter uncertainty	Model uncertainty	Decision-rule uncertainty	Sensitivity of health damage costs with re- gard to changes in the influencing factor	Reference
Emission source: height of release	x	x	x		Increase of up to a factor of 2.45 when mov- ing from a high to a low height of release for direct PM _{2.5} emissions in a given German re- gion (regional modelling); European assess- ment domain	Figure 8.1
Emission source: substances emitted	x	x	x		Considerable variations per tonne of primary substance emitted. E.g. damage costs per tonne of primary PM _{2.5} up to a factor of 58 higher than those per tonne of NMVOC at a low height of release in a given German re- gion; European assessment domain	Figure 8.1
Emission source: emission in- tensity	x	x	x		Decrease of between 3% (Europe modelling domain) and 27% (Île-de-France modelling domain) when moving from low to high emission intensity	section 7.1.1
Background emission inventory	x	x	x	x	Increase of 3% at the France modelling do- main between the year 2009 and a hypothet- ical inventory of the year 2020	section 7.1.2
Use of plume-in-grid model	x	x	x	x	Increase of between a factor of 1.75 (France modelling domain) and 2.51 (Île-de-France modelling domain) when using the plume-in- grid modelling	section 7.1.4
Spatial modelling resolution	x	x	x		Decrease of 9% when moving to a higher resolution (France modelling domain); direction of change depending on the chosen resolution at Île-de-France modelling domain	section 7.1.4

Influencing factor	Variability	Parameter uncertainty	Model uncertainty	Decision-rule uncertainty	Sensitivity of health damage costs with re- gard to changes in the influencing factor	Reference
Temporal modelling resolution/ emission pattern	x	×	×		Increase of between a factor of 1.69 (Europe modelling domain) and 4.11 (Île-de-France modelling domain) when moving from an av- erage (constant emission pattern) to a spe- cific modelling (variable emission pattern)	section 7.1.3
Exposure asses Population	smer x	nt x	x		See above (Emission source: geographic loca-	section
Health impact	asses	smen	t			
Concentra- tion- response function		x	x		Not analysed separately in chapter 7; highest influence for endpoints with important con- tribution to overall damage costs (i.e. mortal- ity impacts due to long-term PM _{2.5} exposure, cf. below)	section 4.3.4
Health base- line rates and age group fractions	x	x			Not analysed separately here; highest influ- ence for endpoints with important contribu- tion to overall damage costs (i.e. mortality impacts due to long-term PM _{2.5} exposure, cf. next item)	n. a.
Approach for mortality impact assessment		x	x	x	Difference of between a factor of 3.09 (France modelling domain) and 3.69 (Europe modelling domain) when switching from an approach based on years of life lost to an ap- proach based on cases of premature deaths	section 7.2.2

Influencing factor	Variability	Parameter uncertainty	Model uncertainty	Decision-rule uncertainty	Sensitivity of health damage costs with re- gard to changes in the influencing factor	Reference
Differential PM compo- nent toxicity		x	x	x	Increase of between a factor of 1.47 (remote location) and 2.02 (densely populated area) when using equal toxicity instead of differen- tial toxicity; European assessment domain; the WHO (2013a) recommends equal PM component toxicity	section 5.4.4.2
Choice of health endpoints		x	x	x	Highest influence for endpoints with im- portant weight in overall damage costs, i.e. all-cause natural mortality due to long-term PM _{2.5} exposure (contributing around 68% to the overall quantified damages when exclud- ing direct NO ₂ -related damages, Europe modelling domain)	section 7.3
		x	x	x	NO ₂ -related long-term exposure mortality impacts contribute to between 3% (Euro- pean modelling domain) and 39% (Île-de- France modelling domain) of total damage costs; strongly dependent on the effect threshold and treatment of double counting with PM effects	section 7.2.1

Monetary valuation

Use of differ- ent base years: inflation and income adjustment		x	x	×	20% increase for an inflation adaptation from the year 2005 to 2015 within the Euro zone; case-specific otherwise. Influence of income adjustment depending on the geographic zone and elasticity of will- ingness-to-pay with regard to personal in- come; 5.2% increase for an adaptation from the year 2005 to 2015 in the EU28	section 6.1.4.2
---	--	---	---	---	---	--------------------

Influencing factor	Variability	Parameter uncertainty	Model uncertainty	Decision-rule uncertainty	Sensitivity of health damage costs with re- gard to changes in the influencing factor	Reference
Discounting		x	x	х	Influence depending on assumptions con- cerning the discount rate; no influence on the outcome of the social CBA in section 7.3, given that annualised costs and benefits are constant over the project lifetime and that an equal social discount rate was applied	section 6.3
Valuation per health endpoint	x	x	x	x	Highest influence for endpoints with important contribution to overall damage costs, i.e. all-cause natural mortality due to long-term $PM_{2.5}$ exposure; Different assumptions on the VOLY have led to a 20% difference in health damage costs	section 4.4.1
Valuation components				x	If only tangible (marketable) components were included in the valuation, the benefits of emission control measures would be sub- stantially lower, owing mainly to the fact that mortality-related impacts would be excluded, cf. Table 6.8	section 6.1.4

As a conclusion, the most notable sensitivities according to the methodological choices analysed here are observed concerning:

• Emission source characteristics: Location, height of release as well as the substances concerned are shown to have a high influence on health damage costs. The variation in health damage costs due to differences in these characteristics can be used as a proxy for the error induced when working with generic or average damage cost factors (cf. the simplified benefit assessment approach in section 7.3.2) instead of more specific modelling approaches;

- Temporal modelling resolution and the use of a plume-in-grid approach (related to spatial resolution at local level) are both shown to have a considerable influence on health damage costs. While a higher temporal modelling resolution generally improves the precision of the results, the added value of the plume-in-grid modelling in terms of robustness remains unclear in the absence of a validation with ambient monitoring data;
- PM component toxicity: Although an important influencing factor in theory, the practical relevance is limited as current recommendations are to treat all particle components as equal in terms of toxicity (WHO 2013a). However, due to ongoing discussions on this point (cf. section 8.4.1), additional sensitivity analyses may be envisaged. This requires to differentiate between primary and secondary particle fractions in the results of the atmospheric modelling;
- The choice of the mortality impact assessment approach is crucial given its high influence on the total health damage costs. In the absence of a clear recommendation concerning the preferred approach, it appears reasonable to carry out a sensitivity analysis using alternative assumptions when conducting social CBA.

8.2.3 Qualitative assessment of uncertainties underlying the social CBA on emission control measures

A qualitative bias review of unquantified elements with an impact on the social CBA of emission control measures (cf. section 7.3) is carried out in the following (Table 8.2), including information on:

- Element: what is the unquantified element?
- Impact on social CBA: describing whether the omission is likely to increase or decrease the societal benefit or private cost (ceteris paribus, i.e. all other elements remaining equal);
- Mechanism: description of how the element influences the outcome of the social CBA.

Note that the information presented below is a first step towards a critical appraisal of the methodology developed within this thesis (cf. section 8.4).

Element	Impact on	Mechanism				
	social CBA					
Emission source ch	aracterisation					
Auto electricity consumption of emission control equipment	Decrease in societal benefits	Relevant mainly for secondary pollution control measures (European Commission 2006b); power plant efficiency de- creases, which results in a slight increase in atmospheric emissions and related damage costs per unit of electricity produced.				
Power plant effi- ciency losses due to part-load operation and cycling	Direction unclear	Although efficiency losses lead to higher emission levels per unit of electricity produced for some pollutants (CO_2 , SO_2), non-linear relationships exist in case of NO _x , also de- pending on the power plant type (Lew et al. 2013); the overall effect on health damage costs is thus unclear.				
Emissions of trace pollutants	Decrease in societal benefits	Depends on the extent to which emissions of trace pollutants are influenced by the installation of emission control equipment. Damage costs due to trace pollutants are highly site-dependent and their quantification is rather complex (Bachmann 2006).				
Emissions of greenhouse gases	Decrease in societal benefits	Depends on the extent to which greenhouse gas emissions are influenced by the installation of emission control equipment. Small increase of damage costs per unit of electricity generated in case of auto electricity con- sumption or other efficiency losses (cf. above).				
Releases into other media, e.g. water or soil	Decrease in societal ben- efits; not rel- evant here	Secondary emission control measures may lead to addi- tional discharges, e.g. into water, causing environmental or health impacts. Quantification is highly site-specific and complex.				
Solid waste discharge	Direction unclear	Solid waste resulting from certain emission control equip- ment needs to be discharged and, depending on the method of discharge and potential valorisation, may cause positive or negative effects.				
Atmospheric modelling and exposure assessment						

Table 8.2: Review of unquantified elements with an impact on the social CBA of emission control measures

Larger geographical	Increase in societal	Extending the assessment to further countries would lead to additional health benefits, however expected to be
zone	benefits	small in the present case.

Element	Impact on social CBA	Mechanism
Dynamic modelling and exposure assess- ment	Direction unclear	During the social CBA, modelling results based on one exemplary year are projected statically into the future. Accounting for (expected) changes in meteorology, background emissions, and population characteristics and distribution over time would add to the precision of the results.
Impact assessment	t	
Effects on crops, building materials and ecosystems	Potentially important increase in societal benefits	Effects limited according to existing quantification meth- ods (cf. section 5.4.2) but potentially important, notably when comprehensively accounting for ecosystem (services).
Additional health endpoints	Increase in societal benefits	As discussed by WHO (2013b), recent evidence exists to quantify further health effects, thereby increasing health impacts of air pollution.
Country-specific health risk parameters and baseline rates	Direction unclear	Effect on health damage costs depending on population characteristics and national health statistics (cf. also section 8.4.1).
Life cycle assess- ment of emission control equip- ment	Decrease in societal benefits	Depending on the place and characteristics of for instance production processes, additional environmental and health damage costs occur, that are however out of scope of the current thesis.
Monetary valuatio	n	
Further valuation components, non-use values in particular	Increase in societal benefits	The more components of an environmental good or ser- vice are assigned a monetary value, the higher the socie- tal benefit related to protecting this good or service.
Private cost assess	ment	
More sophisti- cated private cost assessment	Direction unclear	The costs of emission control measures are specific for each setting and are best assessed by the operators them- selves.

Element	Impact on social CBA	Mechanism
Decreasing secu- rity of supply due to power plant shut downs	Potentially important in- crease in societal costs	If security of supply, as a positive externality of flexible power plants, is not sufficiently accounted for by market mechanisms, there is a risk that these plants are shut down due to unprofitability. This can be exacerbated, for instance, by environmental regulations. Related to this, the IED foresees derogations for peak load power plants.
Distortions in markets for emission control measures	Direction unclear	If costs of emission control measures are distorted, e.g. by market failures such as monopolies or state interventions, the outcome of a social CBA may not reflect societal effi- ciency any more.

8.3 Implications of uncertainty on private and public decision-making

Uncertainties are an obstacle to robust decision-making. Some general remarks regarding uncertainty and decision-making can be made:

- While the uncertainty around single elements may be very high, its impact on decision-making depends on the global influence of the respective elements (cf. section 8.2.2);
- The impact of modelling and parameter uncertainties depends on the way in which results are processed. In case that one single scenario is used to inform decisionmaking, the possible errors induced by key uncertainty factors are potentially very large. The error is reduced, however, when conducting comparisons between different scenarios (e.g. power plant operation scenarios) and when analysing relative differences, such as during the social CBA of emission control measures (cf. section 7.3);
- The relative weight of different types of uncertainties differs as a function of the analysed case and the methods used, as shown in the following examples:

- For instance, Fountoukis et al. (2013) found that uncertainty underlying input data to atmospheric modelling, such as background emissions or meteorology, lead to larger discrepancies between measured and modelled concentrations than differences in the spatial modelling resolution;
- Thompson et al. (2014) found that the uncertainty represented by the confidence interval around the concentration-response function for assessing PM-related mortality is in a similar order of magnitude as using different modelling resolutions;
- Mansfield et al. (2009) found that decision-rule uncertainty (e.g. selecting endpoints and corresponding risk functions) is more influential than underlying parameter uncertainty (i.e. the confidence intervals around the selected risk functions).

Beyond these general remarks, the different ways in which uncertainty impacts decisionmaking in the private and public sector shall be discussed.

8.3.1 Implications of uncertainty for private decision-making

Given that private decision-making differs from public decision-making in certain key aspects (e.g. no public consultations, less competing interests from diverse stakeholder groups, etc.), certain safeguard mechanisms that help to reduce the potential consequences of uncertainty on decision-making are lacking. Especially in case of high financial or political stakes, it is therefore crucial to provide the private decision-maker with robust guidance and methodologies.

In order to produce robust outcomes, a key choice is to determine the general approach for health damage cost assessment. As shown in section 7.3.2, using a simplified approach based on a dated methodology induces the risk of obtaining a different social CBA outcome than the more robust specific assessment approach. Given the lack of methodological guidance for the private sector, this thesis discusses how to choose an appropriate assessment approach under different conditions (cf. section 8.5).

While reducing considerable uncertainties is imperative for robust private decision-making, it needs to be weighed against increasing resources and expertise required for conducting specific modelling approaches. As long as official methodological guidance is lacking, the tolerable level of uncertainty thus depends on the decision context and the financial stakes. Regarding the latter and keeping in mind the substantial costs related to power plant retrofits, it can reasonably be expected that conducting a specific modelling exercise in order to obtain a regulatory derogation will in most cases be justifiable.

Regarding the current methodological state-of-the-art, the approach and valuation parameter for assessing long-term exposure-related mortality crucially influence the quantified health damage costs. In this thesis, a rather conservative approach for mortality impact valuation was used by default, i.e. using a valuation parameter that is around 20 % lower than the central parameter used in recently concluded European air quality legislation (cf. section 6.1.4.1). Such a rather cautious approach appears justified from the perspective of the private decision-maker, given the uncertainties related to valuing nontangible goods through surveys. Similarly, a rather cautious approach to quantifying NO₂ exposure-related mortality risks appears reasonable (i.e. using a concentration threshold of 20 μ g of NO₂/m³ and reducing the impact by a third in order to account for possible overlap with PM, cf. section 7.2.1).

8.3.2 Implications of uncertainty for policy decision-making

While uncertainties underlying damage costs are considerable (cf. section 8.2.1), Rabl et al. (2004) state that the risk of cost penalties due to uncertainties underlying damage costs is overall rather small in policy decision-making. Binary (e.g. as in social CBA) and continuous (e.g. when defining environmental quality standards) decision situations are distinguished.

Quite intuitively, the impact of uncertainties on the result of a social CBA (binary decision) is small in cases where the difference between costs and benefits is initially very large. For continuous choices, the effect of uncertainty is likewise said to be rather small "[...] *because near an optimum, the total social cost varies only slowly as individual cost components are varied*. [...]" (ibid., page 400). However, these findings cannot be transferred to private decision-making where the results are usually much more sensitive towards single choices, as shown in this thesis (cf. section 8.2.2).

The joint consideration of several individual emission sources or even sectors plays a crucial role in typical policy decision-making. Given the many substances emitted at many different places, the effects of influencing factors are more likely to be evened-out at the aggregated scale than at the scale of a single installation. Moreover, the uncertainties underlying atmospheric modelling are lower as the emission signals are typically much higher (cf. section 8.1). Finally, model uncertainty is typically less critical in policy assessments concerned with comparing (inherently uncertain) future emission scenarios. Last but not least, while social CBA supports policy decision-making, it is typically used to prepare political negotiations. Given the diverse stakeholders involved in the decision process, these negotiations typically lead to compromises that are less ambitious than the originally recommended objectives based on social CBA³¹.

8.4 Critical appraisal of the developed methodology framework and associated research opportunities

First elements for a critical appraisal of the social CBA methodology developed in this thesis have already been presented in section 8.2.3 (i.e. the qualitative bias review of unquantified impacts). The current section aims at providing complementary information regarding these elements and to highlight further methodological constraints as well as associated research opportunities. It distinguishes between the health damage cost assessment, private cost assessment and social CBA.

8.4.1 Health damage cost assessment

By definition, any kind of modelling involves a simplified representation of reality and is thus subject to limitations. This is particularly true when valuing intangible goods such as human health and when modelling complex processes related to atmospheric chemistry, as already discussed in section 8.2. Following the steps of the impact pathway approach, selected methodological shortcomings are discussed below.

Emission source characterisation: consideration of cycling and part-load operation on emission levels

The variable emission scenarios underlying the social CBA in chapter 7 are based on the assumption that emission levels are proportional to the electric load. As a result, efficiency losses due to part-load operation and associated changes in emission levels are ignored. Although this limitation is not deemed too relevant when comparing equally affected scenarios as in this thesis, it is nonetheless interesting to briefly explore the associated consequences.

³¹ An example is the recently revised NEC Directive (European Parliament and Council of the European Union 2016), whose final ambition level is lower than the "optimally" defined ambition level based on social CBA (European Commission 2013).

The US Western Wind and Solar Integration Study (Lew et al. 2013) studied the influence of cycling and part-load operation (induced by intermittent renewables) on emission levels of fossil fuel power plants. As a consequence of cycling (load shifts), CO_2 and SO_2 emissions per unit of electricity are found to increase. For NO_x emissions, the case is less obvious. While ramping (i.e. changing the load level) generally leads to higher NO_x emissions, stable part-load operation may avoid NO_x formation when compared to full load operation due to lower levels of (thermal) NO_x formation. However, this effect also depends on the type of power plant technology (Lew et al. 2013).

Whilst being less relevant for social CBA of emission control measures at site-level, the influence of this limitation in social CBA at an aggregate level, e.g. for future policy assessment, could be more substantial. This is due to the increasing flexibility requirements of fossil fuel power plants expected in future energy scenarios (cf. section 2.3).

Atmospheric modelling

Atmospheric modelling based on an Eulerian chemistry transport model as used in this thesis (cf. section 6.1.1) is highly complex and therefore subject to various constraints. Constraints with particular relevance for this thesis are listed below:

- Spatial resolution of background emission inventory: While a relatively high spatial atmospheric modelling resolution at Île-de-France (3 km x 5 km) and French (17 km x 25 km) level was used in the case studies of chapter 7, the resolution of the background emission inventory can be qualified as rather coarse (50 km x 50 km). Even though some adjustments based on land use data were carried out in addition (Legorgeu 2016), the spatial resolution of the emission inventory could be further increased. This would lead to a higher consistency with the grid cell resolution of the atmospheric modelling and, as a consequence, would increase the precision in results (cf. section 8.1.1).
- Plume-in-grid modelling: As shown in this thesis, the influence of using a plume-ingrid approach on health damage costs can be substantial (cf. section 7.1.4). Though consuming a lot of computational resources, such an approach could be justified by the gain in precision in modelled concentrations around the emission source. Yet, the general performance of a plume-in-grid modelling compared to a modelling without plume-in-grid treatment is not automatically higher and depends on various factors, as discussed by Kim et al. (2014). Therefore, further research should aim at increasing the robustness of the plume-in-grid modelling approach

and clarify under which conditions such an approach could represent an added value.

Secondary PM formation: Comparisons of modelled ambient pollutant concentrations based on the Polair3D model (cf. section 6.1.1) with ambient monitoring data have found the largest deviations regarding the formation of secondary inorganic aerosols (Lecœur and Seigneur 2013, Tombette and Sportisse 2007). Secondary inorganic PM is shown to represent a considerable fraction of the overall PM mass (Lecœur and Seigneur 2013). At the same time, PM_{2.5} causes the highest share of quantifiable health damages in this thesis (cf. section 7.3.1). Therefore, it is important to conduct further research on the formation mechanisms of secondary PM and to improve atmospheric models like Polair3D accordingly.

Health impact assessment

A crucial element in health impact assessment are the concentration-response functions that describe the risk increase of developing a disease or dying prematurely per increase in ambient air pollutant concentration. Mainly related to this, some limitations and associated research opportunities shall be discussed:

- Generally, when establishing the relation between concentration change and health risk increase, different functional forms can be used:
 - In the EU context, a linear relationship between concentration and risk change is assumed (Holland 2014b), as adopted in this thesis and conceptualised in equation 3.3;
 - In the US context, a log-linear relationship is commonly assumed (U.S. EPA 2012b), however being approximately linear under common European background conditions;
 - A recently published approach for mortality impact assessment uses socalled integrated exposure-response functions, including effect thresholds at very low concentrations and an increasing saturation at very high concentrations (Burnett et al. 2014). These curves are approximately linear in the concentration ranges typically found under European background conditions.

To sum up, under current European background conditions, the alternative approaches do not deviate much from the linear relationship assumed within this
thesis (cf. equation 3.3). If future conditions shall be assessed, characterised by lower pollution concentration levels, it appears justified to test the implications of alternative functional forms.

- As mentioned in section 6.1.3, concentration-response function are often derived from epidemiological studies, based on long-term population cohort observations. Under ideal settings, the characteristics between the population for which a health impact assessment is carried out and the population observed during epidemiological studies, which underpin the health impact assessment, should be similar. This concerns elements such as gender and age distribution, health status, ambient pollutant concentration levels, exposure patterns etc. However, as epidemiological studies are costly and not universally available, results are often transferred from one context to another. Depending on the level of dissimilarity between population characteristics, the error induced by this transfer can be considerable (Anenberg et al. 2016, Ready et al. 2004). In the particular case of this thesis, the transfer of concentration-response functions is less of an issue given that only parameters recommended for use in Europe by the WHO (2013a) were used. These have been either derived directly for the European context or transferred from the US, where population characteristics are assumed to be similar.
- In the social CBA conducted in section 7.3.1, health impacts based on scenario calculations for one exemplary year are largely extrapolated statically into the future. The only dynamic element is the use of a life table approach for the assessment of future mortality impacts due to exposure during one year (cf. section 6.1.3.2). This static approach represents a simplification of reality, given that background conditions and assessment parameters are evolving over time, e.g. concerning pollution levels, health status and associated baseline rates, life expectancy, concentrationresponse functions etc. Proceeding towards a more dynamic social CBA, particularly when carrying out prospective studies, thus represents an interesting research opportunity. However, given an inherently uncertain future, it should be based on solid assumptions regarding future developments. Carrying out sensitivity analyses appears important.
- Evidence about the relative toxicity of different particles is ambiguous (Krall et al. 2013, Levy et al. 2012, Reiss et al. 2007, Stanek et al. 2011, WHO 2013b). The WHO recommends equal toxicity for all particle components (WHO 2013a), as adopted in the case studies of chapter 7 in this thesis. At the same time, different particle components are acknowledged to trigger different health effects (WHO 2013b), showing a need for further research.

- Another research opportunity is the cause-specific assessment of mortality impacts. According to latest WHO recommendations, specific causes of death are to be assessed in a sensitivity analysis only, while all-cause mortality assessment remains the default approach. This is mainly due to data availability and reliability reasons (WHO 2013a). Yet, related sources state that using cause-specific risk functions and baseline rates would yield more accurate results (Burnett et al. 2014, Miller et al. 2011, WHO 2013b). In an exploratory study, Amann and Schöpp (2011) found damage costs to generally increase when assessing cause-specific mortality impacts. However, own exemplary estimates for France (not included in this thesis) yield opposite results: combining risk slopes from the Global Burden of Disease study (Institute for Health Metrics and Evaluation 2014), based on WHO (2013a), with country-specific background rates (WHO 2014) yields lower mortality impacts per pollution increment than an all-cause mortality risk approach. This is mainly due to the substantially lower baseline rates of the four specific cause categories compared to the all-cause mortality baseline rate. Strongly depending on the context and in particular on the choice of baseline rates and risk slopes, the implications of a cause-specific mortality valuation approach thus deserve more research.
- An ethical concern is related to using an approach based on life years lost for the assessment of mortality impacts due to long-term pollution exposure, as already discussed in section 6.1.3. The fact of implicitly assigning fewer life years lost to the death of elderly people is questioned, as it could, under certain circumstances, lead to age discrimination (OECD 2016).
- Finally, a more regionalised health impact assessment at European scale could be envisaged. At the European domain, EU-wide average baseline rates, age group fractions and concentration-response functions were used. Country-specific parameters were only used at the French and Île-de-France domain (cf. section 6.1.3). A potential improvement would be to proceed towards a more regionalised assessment at European level by using country-specific data (IOM 2011, WHO 2013a). Higher baseline rates, for instance, imply elevated mortality impacts due to PM_{2.5} for Eastern European compared to Central or Western European residents (Miller et al. 2011). However, such an approach may suggest region-specific emission reduction strategies. Less developed countries, often suffering from higher pollution levels, therefore risk to be confronted with substantially increased emission reduction requirements and ultimately higher costs. Although being economically efficient, a more regionalised health impact assessment may thus raise associated equity concerns.

Monetary valuation

The most critical parameter in the monetary valuation of health impacts concerns the assessment of premature deaths and the associated life years lost. Apart from the discussion on the preferred approach and the preferred central estimate (cf. section 8.1.1), more fundamental ethical objections exist.

For instance, when using surveys to value what people are willing to spend in order to avoid the risk of dying prematurely due to air pollution or other causes, the available income plays a key role (Walz 2009). Inevitably, this leads to large discrepancies in the valuation of human life between developed and developing countries, which gives rise to criticism. For this thesis, however, this is less of a concern, given that the same valuation parameter for mortality impacts is applied over the whole European assessment domain.

Moreover, using surveys for the valuation of intangible goods is a major topic of scientific debate, given that these surveys can only be conducted for given population samples with specific characteristics. Yet, results from such surveys are subsequently applied to whole populations. Moreover, surveys are itself subject to biases and constraints (Bateman et al. 2002, Pearce et al. 2006, Turner 2007).

These constraints, the differing recommendations at European and US level and the high weight of mortality-related health damage costs clearly point to a need for further research regarding the monetary valuation of mortality impacts.

8.4.2 Private cost assessment

Regarding the limitations of the private cost assessment, two methodological issues and one assumption underlying the private cost assessment of emission control measures shall be discussed here.

Generally, even though the TFTEI methodology used for cost characterisation (cf. section 6.2) is relatively detailed, it cannot replace a truly site-specific cost assessment, carried out by the operator of an individual emission source himself. Especially in the case of retrofitting emission control measures, technical constraints and other local specificities can lead to considerably different costs than the ones estimated in this thesis and used in the social CBA in section 7.3.1.

As another methodological constraint, a relatively simple cost characterisation was carried out in section 6.2. In particular, the input data used for the cost characterisation, e.g.

energy or resource costs, was assumed to remain constant over the project lifetime. Although the impact of this simplification on the outcome of the social CBA presented in section 7.3.1 is expected to be limited, exploring the effects of a more dynamic cost assessment represents an interesting research opportunity.

A critical assumption concerns the costs of switching to a very low-sulphur fuel, as explored in section 6.2. This cost is likely overestimated given that the source of the cost estimate is somewhat outdated and does neither reflect recent market developments (period of low oil prices in general), nor legislative changes (e.g. stricter European regulations regarding the sulphur content of marine fuels, cf. Directive 2012/33/EU). Although the impact of using a different cost for fuel switching on the social CBA outcome in section 7.3.1 is once again expected to be limited, an updated cost estimate would nonetheless lead to more robust conclusions.

8.4.3 Social CBA of emission control measures

The limitations discussed hereafter are mainly general concerns regarding the social CBA methodology, complemented by some specific limitations concerning the social CBA of emission control measures carried out in this thesis.

Limitations underlying the neoclassical economic approach to dealing with environmental problems, also applying to social CBA, have already been discussed in section 3.1.4. These include most notably concerns related to inter- and intra-generational equity, the concept of weak sustainability and the anthropocentric viewpoint taken in environmental-economic valuation studies.

Another critique is related to the strong focus on efficiency and the binary decision criterion of (social) CBA. This allows to consider only those effects for decision-making that can be assigned a monetary value, potentially missing other important tangible or intangible impacts (cf. section 8.2.3). It is argued that CBA should therefore be used as a decision support, complemented by further relevant information (Boardman et al. 2006, Turner 2007).

A major critique, also evident through the results presented in section 8.2.2, is the risk of biased assessments due to selective methodological choices according to the user's preferences. One way to address this issue is by transparently describing the complete input data into the social CBA as well as using latest scientific recommendations, as demonstrated in this thesis. Standardised methodological guidelines at national or international level are therefore important prerequisites for a harmonised approach to social CBA. No

such guidelines for site-specific social CBA of emission control measures at European level exist yet, although some activities are currently ongoing³². Another factor is a better awareness on the potential influencing factors to which this thesis brought new insights.

Comprehensive uncertainty assessments improve the robustness of social CBA results for use in public and private decision-making. As discussed in section 8.2.1, existing probabilistic uncertainty analyses at aggregate level partly ignore elements related to atmospheric modelling. Moreover, they sometimes include subjective choices regarding probability estimates. An interesting research opportunity would thus be to strive for a more comprehensive probabilistic uncertainty assessment of health damage costs in Europe for use in policy assessments. To this end, existing evidence from different disciplines, particularly air quality modelling, should be combined and comprehensively reviewed.

Due to considerable expertise and resources needed, such a comprehensive probabilistic uncertainty assessment appears less feasible for private decision-making, e.g. the social CBA conducted in chapter 7 of this thesis. Increasing the modelling precision and performing sensitivity analyses are seen as more useful than an exhaustive probabilistic uncertainty assessment in a private decision-making context.

8.5 Using the social CBA methodology in the business context: methodological recommendations

Environmental regulations aiming at the internalisation of environmental and health damages push the private sector to consider related impacts, potentially even to quantitatively assess these impacts. As this thesis shows, there are opportunities to avoid an excessive cost burden in cases where overall welfare is expected to be reduced by too stringent policies (cf. section 2.4.6). To this end, the social CBA methodology needs to be transferred from the aggregate policy level to the level of an individual emission source, as demonstrated throughout this thesis. Applying social CBA at the level of an individual site bears several methodological challenges, e.g. related to choosing the appropriate methods, the appropriate modelling resolution and the treatment of uncertainty.

³² Cf. <u>http://ec.europa.eu/environment/industry/stationary/studies.htm</u>, last accessed: 2017-05-18

Based on the insights generated within this thesis, this section aims at providing methodological guidance for the private sector concerning the use of social CBA for the assessment of emission control measures, notably concerning health damage cost assessment. Private cost characterisation methods are not addressed here, as they are usually well established in the private sector.

How to choose an appropriate method and implementation for health damage cost assessment?

Operators aiming to perform an environmental-economic health damage cost assessment face several possible methods, each with related strengths and weaknesses:

Option 1: The most simple and straightforward way, particularly in the context of the EU's Industrial Emission Directive, is the use of national average unit damage cost factors (cf. section 7.3.2) from the Integrated Pollution Prevention and Control Reference Document on Economics and Cross-Media Effects (European Commission 2006a). As a strength, these values stem from an official European Commission guidance document, include sensitivity results concerning mortality impact assessment, and are readily available for the most common classical air pollutants. As a drawback (cf. also section 8.1.3), these average damage cost factors are highly uncertain given that they fail to account for emission source characteristics such as the specific place and height of release and given that the underlying assessment parameters are out of date.

Option 2: As a next best solution, unit damage cost factors based on parameterised atmospheric modelling, e.g. as implemented in the EcoSenseWeb model can be used (cf. section 7.3.2 and Figure 8.1). Regarding strengths, particularly when assessing effects from large combustion plants, specific results are available for a high height of release and for different sub regions in larger European countries such as France or Germany. Moreover, different background emission inventory data and different meteorological data can be used to generate results. Nevertheless, results per substance emitted are not readily available and need to be extracted using the EcoSenseWeb model, as recently done for Germany by van der Kamp et al. (forthcoming). Taking a business perspective, the most relevant weaknesses are the limited spatial and temporal modelling resolution, as well as partly outdated input data to atmospheric modelling and health impact assessment (cf. section 5.5.2). This lack in resolution is particularly critical when effect thresholds shall be accounted for. Finally, no option to assess direct NO₂-related health impacts is currently available. Option 3: The most sophisticated option for a health damage cost assessment uses specific atmospheric modelling and case specific input data and parameters to assess the impacts of emission scenarios, such as developed and presented throughout chapters 6 and 7 of this thesis. Regarding strengths, such an approach enables the user to: account for temporal variations in emission patterns; choose a spatial modelling resolution that is adapted to the case (e.g. high resolution in case of steep population density gradients near the emission source); increase the modelling precision for a more robust consideration of non-linear atmospheric processes; conduct a detailed assessment of uncertainty and variability underlying the health damage cost assessment. Regarding the limitations and drawbacks (beyond those discussed in section 8.4), the most notable are: the high degree of freedom in choosing modelling features, input data, and further methodological assessment principles. Inevitably, this involves the risk of biased assessments, where users would influence the outcome according to their personal preferences. This risk can be confronted through full methodological transparency, using publicly available data and models as far as available, and relying on the latest scientific state-of-the-art for parameter choices. Moreover, sensitivity analysis of key parameter or modelling choices help showing the range of plausible results. A rather practical and resource-oriented drawback is the high complexity involved in modelling atmospheric processes and chemistry by Eulerian models, such as used in this thesis. Expert knowledge and substantial computational resources³³ are needed for such a task that may only be warranted where the stakes are sufficiently high. Moreover, as a technical constraint, a site-specific approach using Eulerian models at larger geographical scale requires a minimum emission intensity in order to generate a sufficiently visible and robust signal in terms of concentration changes, as discussed, for instance, by Brandt et al. (2012).

As a conclusion from the above presented options, no "one-size fits all" approach to health damage cost assessment exists. A higher precision in the results comes at the expense of higher resource costs and an increasing risk of biased choices. Therefore, the preferred approach depends on the specific question to be addressed, as outlined in the following:

• In case of a simple screening exercise, i.e. in order to get an idea about the order of magnitude of environmental and health damage costs related to a given emission scenario or a given emission source, it may be sufficient to rely on the simplified approach mentioned above (option 1);

³³ The simulations for the case studies carried out within this thesis took several months on a high-speed parallel computer to complete.

- In case that a higher precision is needed, for instance in the context of regulatory requirements, a parameterised or even specific modelling are generally to be preferred. Under the following conditions, a parameterised modelling (e.g. using a 50 km grid cell resolution as in the case of the EcoSenseWeb model) can be considered sufficiently robust (option 2):
 - where local results (i.e. over an area of around 50 km around the emission source) are of minor importance and the focus is on regionalised results (national or European domain);
 - where population densities are relatively even around the emission source (i.e. over an area extending to 50 km or larger around the emission source);
 - where the temporal emission pattern of the emission source is relatively constant throughout the year (e.g. in the case of base load power plants);
 - where effect thresholds play a minor role.
- On the contrary, if these conditions are reversed, a site-specific modelling appears recommendable (option 3). Beyond that, a site-specific modelling is also justified where the stakes (political or financial) are very high.

When aiming to increase the spatial modelling resolution while limiting resource requirements, a pragmatic solution is to combine existing parameterised modelling results with new specific modelling results:

- A specific, more highly resolved modelling is used for the local or national scale, enabling to capture spatial and temporal variability and reducing related uncertainties;
- Results at the larger scale, e.g. European level, are derived on the basis of already existing, parameterised modelling results at a lower resolution.

When aiming to increase the temporal modelling resolution at moderate resource requirements, parameterised model results could be enhanced by accounting for seasonal variations. Given that the formation of secondary PM and ozone is known to be seasondependent (Kaneyasu et al. 1995), such an approach would improve the precision of parameterised models while at the same time limiting computational demands. The choice of the appropriate modelling resolution remains crucial and strongly depends on the case to be analysed. Generally, a higher need for precision at local level and higher stakes justify a higher modelling resolution.

The methodological recommendations given above are summarised by means of a flowchart in Figure 8.2.

How to deal with uncertainty beyond choosing the assessment approach?

Globally, it is important to take a step-wise approach to reducing uncertainty, given that uncertainties are propagated throughout the assessment chain of the impact pathway approach. Reducing uncertainties at an early stage, i.e. starting with atmospheric modelling, is therefore critical.

The different types of uncertainty underlying health damage costs (cf. section 3.4.3) can be reduced as follows:

- Variability is inherent in the modelled systems and, as such, cannot be avoided. However, spatial or temporal variability can be captured in the modelling approach through an increase in resolution of input data and spatial or temporal coverage;
- Parameter uncertainty is best reduced by complying with the scientific state-ofthe-art, such as followed in this thesis. Concretely, this implies using latest WHO (2013a) recommendations for health impact assessment at EU level and for further input parameters, e.g. discount rates. Given that official guidance at EU level is currently lacking for monetary valuation, sensitivity analysis regarding the most influential parameters (i.e. valuation of mortality impacts) should be conducted;
- Model uncertainty is best reduced by selecting models that are adequate for the specific assessment context (e.g. in terms of scope and complexity) and by using input data that is as representative as possible for the case to be assessed, e.g. in terms of temporal and spatial resolution and in terms of up-to-dateness. Where feasible, modelled concentrations should be compared with ambient monitoring data in order to assess the performance of the atmospheric modelling. Advanced probabilistic uncertainty assessment methods can be used to assess the combined effect of parameter and model uncertainty;
- Decision-rule uncertainty is best confronted by the use of sensitivity analysis.



Figure 8.2: Flowchart representing the choice of the basic methodological approach to be used for health damage cost assessment at the level of an industrial point emission source (own conception)

8.6 Summary of chapter 8

This chapter starts by discussing the case studies of chapter 7 in view of underlying causal mechanisms, relevance and robustness of results.

Uncertainty underlying health damage costs is dealt with in various ways. For a quantitative view on uncertainty, the sensitivity of health damage costs with regard to modelling features and methodological choices is summarised (cf. Table 8.1). Among the various case studies presented in this thesis, emission source characteristics, spatial and temporal modelling resolution, background meteorology, and the approach for mortality impact assessment are shown to be most influential. Due to the proximity of a metropolitan area, NO₂-related impacts at local level are likewise found to considerably contribute to overall health damage costs, however depending crucially on the concentration threshold level and modelling resolution.

Owing to the scope of this thesis, certain elements remain unquantified that have the potential to affect the conclusion of the social cost-benefit analysis of emission control measures. These elements are qualitatively reviewed (cf. Table 8.2). Important omissions concern dynamic aspects within the social cost-benefit analysis (direction of effect unclear), ecosystem-related impacts (increase in societal benefits), and issues related to security of supply (potential increase in societal costs).

The impact of uncertainties on decision-making depends on the context, type, and scope of the assessment. Whilst the consequences of uncertainty are argued to be less critical in policy decision-making than typically assumed, the impact at the level of individual industrial emission sources is higher. Yet, the tolerable level of uncertainty depends strongly on the decision context and related political or financial stakes.

Extending beyond uncertainty, a critical appraisal of the methodology developed in this thesis reflects further limitations. Many of these are related to health impact assessment and monetary valuation of intangible goods, mortality impacts in particular. Moreover, global constraints of social cost-benefit analysis are discussed, such as the narrow focus on monetarily quantifiable effects or the high degree of freedom in methodological choices.

Building upon the above findings, the chapter concludes by giving methodological recommendations. These should facilitate the use of social cost-benefit analysis and health damage cost assessment in particular at the level of an individual emission source. Gains in precision through a specific modelling come at the expense of higher resource requirements and a risk of biased decision-making. Resource requirements can be limited by relying on existing modelling results, complemented with specific modelling results for particular applications, e.g. in the presence of large spatial or temporal variations. Biased decision-making is best avoided through a high level of transparency and following guidance from official parties, where available. In addition to this, sensitivity analysis is highly recommended for more robust decision-making, serving to confront some key constraints of social cost-benefit analysis.

9 Summary and outlook

Recent environmental policy developments in Europe open up the possibility of using social cost-benefit analysis (CBA) to assess emission control measures at the level of an industrial point emission source. One typical aim of social CBA in the given context is to assess whether investments into emission control measures are disproportionate by comparing private costs with societal benefits (essentially avoided health damage costs). This method is well established for aggregate policy assessments, e.g. at sectorial or European level. However, as discussed in this thesis (cf. sections 3.5 and 5.5.2), most existing assessment models fail to account for temporal and spatial variations related to atmospheric modelling and exposure assessment. As a result, they are not deemed appropriate for decision support at the level of an individual point emission source.

Against this background, this thesis set out the following objectives:

- Extend the existing scientific methods for the assessment of human health damage costs caused by atmospheric emissions from fossil fuel power plants in order to cope with temporal variability in operation profiles and spatial variability in population densities;
- 2) Transfer the methodology for social CBA of emission control measures from the public policy (aggregate level) to the private business context (site level);
- 3) Derive methodological recommendations accounting for uncertainty in health damage costs.

All three objectives were achieved through the research conducted in this thesis, further summarised below.

Regarding objective 1, a new modelling framework for health damage cost assessment was developed, as transparently and comprehensively described in section 6.1. It considers the latest scientific state-of-the-art and follows the stages of the impact pathway approach, i.e.:

• atmospheric modelling of emission scenarios using an advanced Eulerian chemistry transport model at three spatial domains (with respectively differing spatial modelling resolutions) and at hourly temporal resolution during one year;

- population exposure assessment based upon spatially resolved data within a geographic information system, allowing the assessment of concentration changes, caused by the different emission scenarios, in combination with population data;
- health impact assessment regarding mortality- and morbidity-related impacts, caused by human exposure towards ambient air pollutants (i.e. particulate matter (PM_{2.5} and PM₁₀), ozone (O₃), and nitrogen dioxide (NO₂);
- monetary valuation, assigning tangible and intangible valuation components to the physical health effects assessed in the previous step.

Given that different power plant operation scenarios are compared in this thesis, it was decided to express the resulting health damage costs in \in per MWh of electricity generated.

For the purpose of objective 2, the newly developed approach for health damage cost assessment at site level (cf. objective 1) was combined with a private cost characterisation method. The annuity method was used to determine annual costs of selected primary and secondary emission control measures at an exemplary fossil-fired peak-load power plant (cf. section 6.2). This permitted to carry out an exemplary social CBA at site-level (cf. section 7.3). The emission reduction measures considered were not found to be socially efficient according to the decision rule of social CBA. Yet, as shown in complementary analyses, switching to a simplified health damage cost assessment or using an alternative mortality impact assessment approach altered the result, making the investments socially efficient. This is a typical example of decision-rule uncertainty related to selecting an appropriate health damage cost assessment approach.

Using the newly developed approach for health damage costs, key modelling features and methodological choices were varied and the quantitative sensitivities of health damage costs with regard to single elements were summarised (Table 8.1) and discussed (cf. section 8.1).

Choices concerning spatial and temporal modelling resolution were confirmed to be important influencing factors. However, their influence depends on further, site-specific elements, particularly emission source characteristics and the population distribution. For this reason, findings regarding the sensitivity of health damage costs with regard to changes underlying atmospheric modelling cannot be generalised. Nonetheless, these findings make it possible to estimate the potential error induced by using an average modelling approach instead of a highly resolved specific modelling approach. As shown in the case studies of this thesis, this error led to at least a doubling of the resulting health damage costs, thus being crucial for decision-making.

Methodological choices in health impact assessment were likewise shown to be influential, particularly those related to quantifying mortality impacts due to long-term PM_{2.5} exposure, which is the single most influential endpoint (contributing around 68% to the overall quantified damages when excluding direct NO₂-related damages, cf. section 7.3). Switching from an assessment approach based on years of life lost to an approach based on cases of premature deaths universally increased associated health damage costs at least threefold (cf. section 7.2.2).

At the local modelling domain, characterised by the presence of a large metropolitan area, NO₂-related mortality impacts were shown to contribute considerably to overall health damage costs (cf. section 7.2.1). This result, however, depends crucially on the concentration threshold, above which impacts are assumed to occur. As a consequence, the spatial atmospheric modelling resolution should be sufficiently high when aiming to properly account for threshold effects.

For the purpose of objective 3, uncertainties underlying social CBA and particularly the health damage cost assessment were comprehensively reviewed in a quantitative and qualitative way (cf. section 8.2), including but not limited to the sensitivity analyses presented above. This review also provided the basis for methodological recommendations (cf. section 8.5 and below).

While health damage costs were shown to be very sensitive towards modelling parameters and assumptions, the tolerable level of uncertainty in decision-making depends on the decision context and the associated financial or political stakes (cf. section 8.3). Moreover, the impact of uncertainties on decision-making depends on the type and scope of the assessment. In public decision-making and generally when conducting scenario comparisons, e.g. within social CBA, modelling and parameter uncertainties are argued to have a smaller influence. For decision-making concerned with assessing industrial point emission sources from a private or regulatory perspective, however, the consequences of uncertainty can be substantial. This points to the need to establish methodological guidance regarding the use of social CBA of emission control measures at the level of an individual emission source in the regulatory process. Due to the fact that private cost assessment is common practice for businesses, official guidance in this field can be limited to key parameters, e.g. sector-specific interest rates. Methodological guidance is more critically needed in view of the health damage cost assessment, also due to the ongoing methodological evolutions (cf. section 8.4). Such methodological guidance, aimed primarily at private decision-makers, was therefore provided in this thesis (cf. section 8.5) and shall be briefly summarised hereafter.

An elementary choice is related to the general approach and model for health damage cost assessment. As shown in section 7.3.2, the use of simplified approaches, based on average modelling, induces a risk of erroneous decisions. At the same time, the increase in precision and robustness enabled by a site-specific modelling needs to be weighed against an increase in complexity, resource requirements, and potential biases in results. The latter is due to the manifold methodological choices that could be used to influence the outcome according to the user's preferences. This risk is notably among the key constraints regarding the use of social CBA for decision-making (cf. section 8.4.3). Yet, where the financial or political stakes are sufficiently high and the temporal or spatial variations are expected to be important, a specific modelling approach appears justified. This is especially true for the case studied in this thesis, characterised both by variations in the emission pattern (peak load power plant) and by variations in population density (proximity of large metropolitan area). Alternative ways of assessing health damage costs with lower resource requirements are also proposed, e.g. by combining a local site-specific assessment with parameterised modelling over a larger zone.

In order to reduce uncertainty and avoid biased results, a site-specific modelling approach should be transparently documented and should follow the latest official methodological recommendations. This underlines the importance for policy makers to provide such methodological guidance. Stakeholder consultations present a possibility to gather further decision-relevant information beyond social CBA. Generally, sensitivity analysis of key parameter and assumptions should be conducted for private and political decision-making alike in order to enhance the robustness of the social CBA outcome and to provide a broader picture on potential impacts.

To conclude, even though the use of social CBA and environmental-economic assessment methods is subject to limitations (cf. section 8.4), it is the author's opinion that human wellbeing and environmental resources are served better by applying such methods than by ignoring or rejecting them.

Outlook

If welfare economic instruments such as social CBA are increasingly integrated into regulatory processes, a harmonised and robust framework for estimating damage costs is needed, allowing competent authorities to base their decisions on comparable grounds. Regarding health-related damage costs of classical air pollutant emissions in Europe, the impact functions proposed by the WHO's HRAPIE project can currently be considered the first choice (cf. section 6.1.3). A similar harmonised effort for monetary valuation, especially with regard to valuing mortality impacts, is currently lacking. When aiming to provide a consistent methodological basis regarding atmospheric modelling, sufficiently resolved standardised modelling results at European level appear as a promising starting point (e.g. unit damage cost factors per sector, zone of release, and height of release). Depending on the spatial and temporal modelling resolution and the number of emission zones and sectors distinguished, the computational calculation effort could, however, be substantial. For decision-making at site level, it appears useful to combine such regional results with more highly-resolved local modelling results.

Several methodological developments are worth noting that are expected to lead to more robust health damage cost assessments in the future:

- Advances in computational power allow ever more refined atmospheric modelling and the associated assessment of health impacts or damages up to the global scale (Apte et al. 2015, Brauer et al. 2016, Lelieveld et al. 2015). Related to this, there is a trend to improve the quality of modelled concentration data through a combination of chemistry transport modelling, ambient monitoring data, and remote sensing data (Anenberg et al. 2016, Silver et al. 2016).
- The internet of things, characterised amongst other by the widespread use of decentralised, low-cost sensors, also enables new ways of monitoring ambient concentrations of air pollutants (Deville Cavellin et al. 2016, Holstius et al. 2014, Snyder et al. 2013). This could be useful for the validation of atmospheric modelling results and the improvement of the modelling performance. Further research efforts are nonetheless needed in order to assess the reliability and performance of such sensors, especially when used outdoor.
- When quantifying health impacts and damage costs, there is a trend to regionalised assessments. These will lead to a higher precision in results but may provoke new discussions on ethical grounds, e.g. related to a fair distribution of the burden of pollution reduction between more or less developed countries (cf. section 8.4).
- Generally, there is an ever growing scientific literature on the health impacts caused by ambient air pollution exposure (WHO 2013b). A crucial next step for increased precision would be to proceed towards a cause-specific mortality impact assessment (cf. section 8.4).

• Environmental economic survey methods for the valuation of intangible goods are likewise subject to continuous scientific improvement (Johnston et al. 2017), lead-ing to a higher confidence in the non market-based valuation components underlying health damage costs.

Appendix

Referring to section 2.4, Table A.1 gives a comprehensive overview on the potential links between industrial emissions (pressures), states, and impacts for the case of a large fossil fuel power plant (driver). The potential environmental impacts are classified according to the so-called areas of protection:

- Human health: Upon exposure, air pollution increases the risk for human health impacts;
- Man-made environment or society: Air pollution affects building materials due to corrosion; heat discharges into rivers can impact on other heat discharging installations downstream; even though involving ecosystems, impacts on forestry and agriculture are also classified to fall into this category, as both are economic sectors;
- Ecosystems: Acid rain impacts soils and thus habitats of species; heat discharges in water bodies may lead to a lack of oxygen.

Pressures	State changes induced	Areas of protection	Examples of Impacts
Greenhouse gas (GHG) emissions	GHG concentra- tions in the air im- plying changes in, for instance: Tem- perature, glacier volumes, intensi- ties and frequen- cies of natural hazards, sea level	Human health Man-made environment/ society	Negative:Heat stress, increasedprevalence of infectious diseasesPositive:Less cold stressNegative:Abandoning coastal ar-eas, dyke building, increased de-mand for cooling, migration ofpeople, water shortages, social in-stabilitiesPositive:New agricultural opportunities (e.g. in Nordic countries), re-duced demand for heating, ferti-lizer effect of CO2 in forestry andagriculture

 Table A.1:
 Overview on selected environmental pressures and related impacts caused by fossil fuel-fired power generation; adapted from Bachmann and van der Kamp (2014)

Ap	pend	ix
----	------	----

Pressures	State changes induced	Areas of protection	Examples of Impacts
		Ecosystems	<u>Negative:</u> Decline of coral reefs due to ocean acidification, sea level rise, shift in climatic zones to which ecosystems/species cannot adapt, biodiversity and habitat im- pacts due to water shortages and sea-level rise, crops and trees not adapted to increased tempera- tures
Emissions of classical air pollutants	Concentration of classical air pollu- tants indoor and outdoor, also im- plying state changes in pH of precipitation	Human health	<u>Negative:</u> Cardio-pulmonary mor- bidity and increasing mortality risk, cancers
		Man-made environment/ society	<u>Negative:</u> Corrosion and soiling of building materials, crop yield losses, acidification <u>Positive:</u> Fertilizer effect of SO ₂ and NO _x on agricultural land (at low input rates)
		Ecosystems	<u>Negative:</u> Biodiversity and ecosys- tem service impacts due to eu- trophication and acidification
Releases of heavy metals and other toxic substances	Concentration of heavy metals and hazardous chemi- cals in soils, water and air	Human health	<u>Negative:</u> Cardio-pulmonary mor- bidity, cancers
		Man-made environment/ society	<u>Negative</u> : Certain areas can no longer be used for specific pur- poses (e.g. playgrounds for chil- dren, residential areas)
		Ecosystems	<u>Negative:</u> Biodiversity and ecosys- tem service impacts due to soil and water pollution
Heat emissions	Air temperature, water tempera- ture	Human health	<u>Negative:</u> Local heat production may aggravate heat stress <u>Positive:</u> Local heat production may alleviate cold stress

Pressures	State changes induced	Areas of protection	Examples of Impacts
		Man-made environment/ society	<u>Negative:</u> Restrictions of power plants to inject cooling water into rivers due to heat inputs upstream
		Ecosystems	<u>Negative:</u> Biodiversity and ecosys- tem service impacts due to lack of oxygen
Noise (and vibrations)	Noise level	Human health	<u>Negative:</u> Hypertension and is- chaemic heart disease, disamenity (e.g. sleep disturbances)
		Man-made environment/ society	<u>Negative:</u> Building material dam- ages.
		Ecosystems	<u>Negative:</u> Effects on animal's phys- iology and behaviour

Table A.2 shows the input data requirements of the damage cost assessment model EcoSenseWeb, used throughout chapters 4 and 5.

Characteristics	Unit
Plant design	
Year of assessment ^{a)}	Year
Period of assessment (in years) ^{a)}	Years
Stack height	m
Stack diameter	m
Flue gas volume stream ^{b)}	Nm³/h
Flue gas temperature	К
Latitude	°N
Longitude	°E

Table A.2:	Data input requirements in EcoSenseWeb for a fossil-fired power plant;
	based on Preiss and Klotz (2008)

Characteristics	Unit
Full load hours per year ^{c)}	h/a
Electricity production per year ^{d)}	GWh/a
Emissions of classical air pollutants	
NO _x	mg/Nm ³
SO ₂	mg/Nm ³
Primary particles with an aerodynamic diameter of less than 10 μm (PM_{10})	mg/Nm ³
Primary particles with an aerodynamic diameter of less than 2.5 μm (PM_{2.5})	mg/Nm ³
Emissions of trace pollutants	
Cadmium (Cd)	µg/Nm³
Mercury (Hg)	µg/Nm³
Arsenic (As)	μg/Nm ³
Lead (Pb)	µg/Nm³
Chromium (Cr)	μg/Nm ³
Chromium-VI (Cr-VI)	μg/Nm ³
Nickel (Ni)	μg/Nm ³
Dioxins and dioxin-like compounds e)	ng/Nm ³
Emissions of greenhouse gases (during operation)	
CO ₂ , CH ₄ , N ₂ O	t/a
Emissions of greenhouse gases (upstream and downstream)	
CO ₂ , CH ₄ , N ₂ O	t
Land use change	
Land use change ^{f)}	-

^{a)} Only needed for valuing greenhouse gas emissions, i.e. not relevant here

^{b)} Flue gas volume stream is given under standard conditions denoted by "N" in the unit (e.g. according to the German standard DIN 1343 or ISO norm 2533). These are a temperature of 0 °C and an ambient air pressure of 1013 hPa. It shall further be verified that the flue gas volume stream and the pollutant concentrations are consistent in terms of oxygen-content.

c) EcoSenseWeb internally distributes the emissions associated with the indicated full load hours equally across the year.

^{d)} The parameter "electricity production per year" means the net amount of electricity delivered to the electricity grid, i.e. gross electricity production minus on-site electricity demand/consumption.

- e) Input of Toxic Equivalent (TEQ) values is required (cf. (European Commission 1999a, Watkiss 2007)).
- ^{f)} 22 land use types are distinguished. Land use changes are intended to be due to the construction of the power plant. However, land use changes can be specified at any step of the fuel cycle. The underlying assumptions, however, confine those land use changes to Europe; irrelevant in the context of the current work.

The following tables (Table A.3 to Table A.5) provide background statistics used for the health impact assessment described in section 6.1.3.

	Île-de-France 2009	France 2009	EU-average 2009
all	100%	100%	100%
Infants	1.43%	1.22%	1.07%
children5to14	12.63%	12.20%	10.43%
children5to19	18.91%	18.43%	16.24%
children6to12	8.90%	8.58%	7.28%
adults18+	76.79%	78.03%	80.84%
adults20to64	61.56%	58.63%	61.17%
adults30+	59.54%	63.04%	65.52%
adults65+	12.62%	16.82%	17.34%

 Table A.3:
 Age group fractions and corresponding years used in the health impact assessments of this thesis; sources: Île-de-France and France: (INSEE 2009c); EU-average: (WHO 2016)

Table A.4:EU and French baseline rates for 2009 per health endpoint used in the health impact assessments
of this thesis

Endpoint detailed	Age group	EU baseline	FR baseline	Unit (per person)	Source
all-cause natural mortality (acute)	all	0.0108	0.0080	deaths/year	MDB
cardiovascular hospital admission (excl. stroke)	adults65+	0.0578	0.0763	cases/year	HMDB

Appendix

Endpoint detailed	Age group	EU	FR	Unit	Source
	00 9. ovh	baseline	baseline	(per person)	
respiratory hospital admission	adults65+	0.0297	0.0268	cases/year	HMDB
minor restricted activity day (MRAD)	all	7.8	7.8	days/year	(Ostro and Rothschild 1989)
all-cause natural mortality (VSL)	adults30+	0.0146	0.0127	deaths/year	DMDB
cardiovascular hospital admission	all	0.0240	0.0228	cases/year	HFADB
respiratory hospital admission	all	0.0139	0.0128	cases/year	HFADB
net restricted activity day (netRAD)	all	19	19	days/year	(Ostro and Rothschild 1989)
work loss day (WLD)	adults20to64	9.8739	9.8739	days/year	HFADB ^a
asthma symptom day	children5to19	0.1700	0.1700	daily incidence	HRAPIE
all-cause infant mortality	infants	0.0014	0.0012	deaths/year	HFADB
bronchitis prevalence	children6to12	0.1860	0.1860	annual prevalence	HRAPIE
chronic bronchitis case	adults18+	0.0039	0.0039	cases/year	HRAPIE
bronchitis symptom days	children5to14	0.2990	0.2990	annual prevalence	HRAPIE

^a considering only working population; EU average used for France, since latest statistics for France date from year 1988; EU-average employment rate taken from eurostat (2015b)

Synonyms and sources: MDB = mortality database, HMDB = Hospital Morbidity Database (WHO 2016); HRAPIE = Health Risks of Air Pollution In Europe (WHO 2013a)

	Life expectancy at birth, year 2009	YOLL [1/(10 µg/m ³)*person*a]
Austria	80.6	0.00948
Belgium	80.3	0.00961
Bulgaria	73.8	0.01287
Croatia	76.4	0.01143
Cyprus	81.3	0.00920
Czech Republic	77.5	0.01089
Denmark	80.1ª	0.00970
Estonia	75.3	0.01201
Finland	80.2	0.00966
France ^b	81.8	0.00900
Germany	80.4	0.00956
Greece	80.3	0.00959
Hungary	74.5	0.01249
Iceland	81.9	0.00893
Ireland	80.0	0.00975
Italy	82.1	0.00888
Latvia	73.3	0.01316
Lithuania	73.2	0.01319
Luxembourg	81.6	0.00908
Malta	80.5	0.00954
Netherlands	81.0	0.00931
Poland	75.9	0.01170
Portugal	79.6	0.00991
Romania	73.6	0.01296
Slovakia	75.4	0.01195

 $\begin{array}{ll} \mbox{Table A.5:} & \mbox{Life expectancy in EU28 member states in the year 2009 and YOLL (population aged 30+) per 1-year increase of 10 \mbox{} \mu g/m^3 \mbox{PM}_{2.5} \mbox{ concentration (life expectancy data from HFA DB (WHO 2016); YOLL estimates based on Miller et al. (2011)) } \end{array}$

	Life expectancy at birth, year 2009	YOLL [1/(10 µg/m³)*person*a]
Slovenia	79.5	0.00998
Spain	81.9	0.00894
Sweden	81.6	0.00906
United Kingdom	80.6	0.00950
EU (average)	79.9	0.00980

^a 2011 data, since no 2009 data available

^b note that detailed life table results for France are used in the impact assessment, cf. section 6.1.3.2

References

- Amann, M. (2012), "The GAINS Integrated Assessment Model". European Consortium for Modelling of Air Pollution and Climate Strategies - EC4MACS, p. 43. Online available at: <u>http://www.ec4macs.eu/content/report/EC4MACS_Publications/MR_Final%20</u> in%20pdf/GAINS_Methodologies_Final.pdf
- Amann, M., Borken-Kleefeld, J., Cofala, J., Hettelingh, J.-P., Heyes, C., Hoglund, L., Holland, M., Kiesewetter, G., Klimont, Z., Rafaj, P., Posch, M., Sander, R., Schöpp, W., Wagner, F., Winiwarter, W. (2014), "The Final Policy Scenarios of the EU Clean Air Policy Package - TSAP Report #11, Version 1.1a". International Institute for Applied Systems Analysis (IIASA), Laxenburg, Austria. p. 43. Online available at: http://www.iiasa.ac.at/web/home/research/researchPrograms/MitigationofAir

http://www.iiasa.ac.at/web/home/research/researchPrograms/MitigationofAir PollutionandGreenhousegases/TSAP_11-finalv1-1a.pdf

- Amann, M., Schöpp, W. (2011), "Calculation of cause-specific mortality impacts of fine particulate matter in GAINS". Centre for Integrated Assessment Modelling (CIAM), International Institute for Applied Systems Analysis (IIASA), Laxenburg, Austria. p. 18.
- Andersen, M.S., Frohn, L.M., Nielsen, J.S., Nielsen, M., Jensen, J.B., Jensen, S.S., Christensen, J., Brandt, J. (2006), "EVA - a non-linear Eulerian approach for assessment of health-cost externalities of air pollution".9th Biennial Conference of the International Society for Ecological Economics (ISEE2006). New Delhi, India. p. 25. Online available at: <u>http://www2.dmu.dk/Pub/EVA_artikel.pdf</u>.
- Anenberg, S.C., Belova, A., Brandt, J., Fann, N., Greco, S., Guttikunda, S., Heroux, M.-E., Hurley, F., Krzyzanowski, M., Medina, S., Miller, B., Pandey, K., Roos, J., Van Dingenen, R. (2016), "Survey of Ambient Air Pollution Health Risk Assessment Tools". *Risk Analysis*, 36 (9), p. 1718-1736.
- Apte, J.S., Marshall, J.D., Cohen, A.J., Brauer, M. (2015), "Addressing Global Mortality from Ambient PM2.5". *Environmental Science & Technology*, 49 (13), p. 8057-8066.
- Assemblée Nationale (2015), "Projet de loi relatif à la transition énergétique pour la croissance verte". ISBN: TEXTE ADOPTÉ n° 519. Online available at: <u>http://www.assemblee-nationale.fr/14/ta/ta0519.asp</u>.
- Atkinson, S.E., Lewis, D.H. (1974), "A cost-effectiveness analysis of alternative air quality control strategies". *Journal of Environmental Economics and Management*, 1 (3), p. 237-250.

- Bachmann, T.M. (2006), *Hazardous substances and human health: exposure, impact and external cost assessment at the European scale*. Elsevier, Amsterdam. ISBN: 0-444-52218-2.
- Bachmann, T.M., van der Kamp, J. (2014), "Environmental cost-benefit analysis and the EU (European Union) Industrial Emissions Directive: Exploring the societal efficiency of a DeNOx retrofit at a coal-fired power plant". *Energy*, 68 (0), p. 125-139.
- Bateman, I.J., Carson, R., Day, B., Hanemann, M., Hett, T., and others (2002), *Economic Valuation with Stated Preference Techniques. A Manual*. Edward Elgar, Cheltenham, UK.
- Baumbach, G. (1994), *Luftreinhaltung*. Springer, Berlin ; Heidelberg [u.a.]. ISBN: 3-540-56823-9.
- Baumol, W.J., Oates, W.E. (1988), *The theory of environmental policy*. Cambridge Univ. Pr., Cambridge. ISBN: 0-521-32224-3, 0-521-31112-8 (pbk.).
- BGR (2015), "Energy Study 2015. Reserves, resources and availability of energy resources (19)". Federal Institute for Geoscience and Natural Resources (BGR), Hannover, Germany. Online available at: <u>http://www.bgr.bund.de/EN/Themen/Energie/Produkte/energy_study_2015_s</u> <u>ummary_en.html?nn=1548106</u>
- BGR (2016), "Energiestudie 2016. Reserven, Ressourcen und Verfügbarkeit von Energierohstoffen (20)". Bundesanstalt für Geowissenschaften und Rohstoffe, Hannover, Germany. p. 180. Online available at: <u>http://www.bgr.bund.de/DE/Themen/Energie/Produkte/energiestudie2016 Zu</u> <u>sammenfassung.html</u>
- BMBU (2016), "Klimaschutzplan 2050 Klimaschutzpolitische Grundsätze und Ziele der Bundesregierung ". Bundesministerium für Umwelt, Naturschutz, Bau und Reaktorsicherheit, Berlin, Germany. Online available at: <u>http://www.bmub.bund.de/fileadmin/Daten BMU/Download PDF/Klimaschut z/klimaschutzplan 2050 bf.pdf</u>
- BMWi (2014), "Second Monitoring Report "Energy of the future" Summary". Federal Ministry for Economic Affairs and Energy (BMWi), Berlin, Germany. p. 12. Online available at:

http://www.bmwi.de/EN/Service/publications,did=639404.html

- Boardman, A., Greenberg, D., Vining, A., Weimer, D. (2006), *Cost Benefit Analysis: Concepts and Practice*. Prentice Hall, Upper Saddle River, NJ. ISBN: 0131435833, 9780131435834.
- Bockstael, N.E., Freeman, III, A.M., Kopp, R.J., Portney, P.R., Smith, V.K. (2000), "On Measuring Economic Values for Nature". *Environmental Science & Technology*, 34 (8), p. 1384-1389.

- Boutahar, J., Lacour, S., Mallet, V., Quélo, D., Roustan, Y., Sportisse, B. (2004), "Development and validation of a fully modular platform for numerical modelling of air pollution : POLAIR". *International Journal of Environmental Pollution*, 22 (1-2)
- Brandt, J., Silver, J.D., Christensen, J.H., Andersen, M.S., Bønløkke, J.H., Sigsgaard, T., Geels, C., Gross, A., Hansen, A.B., Hansen, K.M., Hedegaard, G.B., Kaas, E., Frohn, L.M. (2013), "Contribution from the ten major emission sectors in Europe and Denmark to the health-cost externalities of air pollution using the EVA model system an integrated modelling approach". *Atmos. Chem. Phys.*, 13 (15), p. 7725-7746.
- Brandt, J., Silver, J.D., Frohn, L.M., Geels, C., Gross, A., Hansen, A.B., Hansen, K.M., Hedegaard, G.B., Skjøth, C.A., Villadsen, H., Zare, A., Christensen, J.H. (2012), "An integrated model study for Europe and North America using the Danish Eulerian Hemispheric Model with focus on intercontinental transport of air pollution". *Atmospheric Environment*, 53 (0), p. 156-176.
- Brauer, M., Freedman, G., Frostad, J., van Donkelaar, A., Martin, R.V., Dentener, F., Dingenen, R.v., Estep, K., Amini, H., Apte, J.S., Balakrishnan, K., Barregard, L., Broday, D., Feigin, V., Ghosh, S., Hopke, P.K., Knibbs, L.D., Kokubo, Y., Liu, Y., Ma, S., Morawska, L., Sangrador, J.L.T., Shaddick, G., Anderson, H.R., Vos, T., Forouzanfar, M.H., Burnett, R.T., Cohen, A. (2016), "Ambient Air Pollution Exposure Estimation for the Global Burden of Disease 2013". *Environmental Science & Technology*, 50 (1), p. 79-88.
- Brauner, G., Glaunsinger, W., Bofinger, S., John, M., Magin, W., Pyc, I., Schüler, S., Schulz, S., Schwing, U., Seydel, P., Steinke, F. (2012), "Erneuerbare Energie braucht flexible Kraftwerke – Szenarien bis 2020". VDE, ETG-Task Force Flexibilisierung des Kraftwerksparks, Frankfurt am Main, Germany. p. 75. Online available at: <u>https://www.vde.com/de/fg/ETG/Arbeitsgebiete/V1/Aktuelles/Oeffentlich/Seit</u> en/StudieFlexibilisierung.aspx
- Breun, P., Comes, T., Doll, C., Fröhling, M., Hiete, M., Ilsen, R., Krail, M., Lützkendorf, T., Schultmann, F., Stengel, J., Unholzer, M. (2012), "National Integrated Assessment Modelling zur Bewertung umweltpolitischer Instrumente -Entwicklung des otello-Modellsystems und dessen Anwendung auf die Bundesrepublik Deutschland". Karlsruher Institut für Technologie (KIT), Karlsruhe, Germany. p. 514. Online available at: <u>http://dx.doi.org/10.5445/KSP/1000028216</u>
- Bundesregierung (2010), "Das Energiekonzept Beschluss des Bundeskabinetts vom 28. September 2010". p. 40. Online available at: <u>http://www.bundesregierung.de/ContentArchiv/DE/Archiv17/ Anlagen/2012/</u> <u>02/energiekonzept-final.pdf?</u><u>blob=publicationFile&v=5</u>

- Burnett, R.T., Pope, C.A., Ezzati, M., Olives, C., Lim, S.S., Mehta, S., Shin, H.H., Singh, G., Hubbell, B., Brauer, M., Anderson, H.R., Smith, K.R., Balmes, J.R., Bruce, N.G., Kan, H., Laden, F., Prüss-Ustün, A., Turner, M.C., Gapstur, S.M., Diver, W.R., Cohen, A. (2014), "An integrated risk function for estimating the global burden of disease attributable to ambient fine particulate matter exposure". *Environmental Health Perspectives*, 122 (4), p. A88–A111 & 325–430.
- Capros, P., De Vita, A., Tasios, N., Papadopoulos, D., Siskos, P., Apostolaki, E., Zampara, M., Paroussos, L., Fragiadakis, K., Kouvaritakis, N., Hoglund-Isaksson, L., Winiwarter, W., Purohit, P., Bottcher, H., Frank, S., Havlik, P., Gusti, M., Witzke, H.P. (2013), "EU Energy, Transport and GHG Emissions Trends to 2050 -Reference Scenario 2013". European Commission, Luxembourg. p. 168. Online available at: <u>http://ec.europa.eu/transport/media/publications/doc/trends-to-2050-update-2013.pdf</u>
- Capros, P., Vita, A.D., Tasios, N., Siskos, P., Kannavou, M., Petropoulos, A.,
 Evangelopoulou, S., Zampara, M., Papadopoulos, D., Nakos, C., Paroussos, L.,
 Fragiadakis, K., Tsani, S., Karkatsoulis, P., Fragkos, P., Kouvaritakis, N., HöglundIsaksson, L., Winiwarter, W., Purohit, P., Gomez-Sanabria, A., Frank, S., Forsell,
 N., Gusti, M., Havlík, P., Obersteiner, M., Witzke, H.P., Kesting, M. (2016), "EU
 Reference Scenario 2016 Energy, transport and GHG emissions Trends to 2050
 ". Publications Office of the European Union, Luxembourg. p. 217. Online
 available at: http://ec.europa.eu/energy/en/data-analysis/energy-modelling
- CDC, 2015. The Different Types of Health Assessments. available at: <u>http://www.cdc.gov/healthyplaces/types_health_assessments.htm</u>. Accessed on 2015-02-24.
- Couvidat, F., Debry, É., Sartelet, K., Seigneur, C. (2012), "A hydrophilic/hydrophobic organic (H2O) aerosol model: Development, evaluation and sensitivity analysis". *Journal of Geophysical Research: Atmospheres*, 117 (D10), p. n/a-n/a.
- Curtiss, P.S., Rabl, A. (1996), "Impacts of air pollution: general relationships and site dependence". *Atmospheric Environment*, 30 (19), p. 3331-3347.
- Czarnowska, L., Frangopoulos, C.A. (2012), "Dispersion of pollutants, environmental externalities due to a pulverized coal power plant and their effect on the cost of electricity". *Energy*, 41 (1), p. 212-219.
- Debry, E., Fahey, K., Sartelet, K., Sportisse, B., Tombette, M. (2007), "Technical Note: A new Size REsolved Aerosol Model (SIREAM)". Atmos. Chem. Phys., 7 (6), p. 1537-1547.
- DECC (2013), "Electricity Generation Costs (December 2013)". Department of Energy & Climate Change, London, United Kingdom. p. 61. Online available at: <u>https://www.gov.uk/government/uploads/system/uploads/attachment_data/fi</u> <u>le/269888/131217_Electricity_Generation_costs_report_December_2013_Fina</u> <u>l.pdf</u>

DEFRA (2015), "Air quality economic analysis - Damage costs by location and source". Department for Environment, Food and Rural Affairs, UK. p. 4. Online available at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/fi le/460398/air-quality-econanalysis-damagecost.pdf

- Desaigues, B., Ami, D., Bartczak, A., Braun-Kohlová, M., Chilton, S., Czajkowski, M., Farreras, V., Hunt, A., Hutchison, M., Jeanrenaud, C., Kaderjak, P., Máca, V., Markiewicz, O., Markowska, A., Metcalf, H., Navrud, S., Nielsen, J.S., Ortiz, R., Pellegrini, S., Rabl, A., Riera, R., Scasny, M., Stoeckel, M.E., Szántó, R., Urban, J. (2011), "Economic valuation of air pollution mortality: A 9-country contingent valuation survey of value of a life year (VOLY)". *Ecological Indicators*, 11 (3), p. 902-910.
- Desaigues, B., Ami, D., Hutchison, M., Rabl, A., Chilton, S., Metcalf, H., Hunt, A., Ortiz, R., Navrud, S., Kaderjak, P., Szántó, R., Nielsen, J.S., Jeanrenaud, C., Pellegrini, S., Kohlová, M.B., Scasny, M., Máca, V., Urban, J., Stoeckel, M.-E., Bartczak, A., Markiewicz, A., Riera, P., Farreras, V. (2007), "Final Report on the monetary valuation of mortality and morbidity risks from air pollution, Delivery n° 6.7 - RS 1b of the NEEDS project". UParis, Paris. p. 55.
- Deville Cavellin, L., Weichenthal, S., Tack, R., Ragettli, M.S., Smargiassi, A., Hatzopoulou, M. (2016), "Investigating the Use Of Portable Air Pollution Sensors to Capture the Spatial Variability Of Traffic-Related Air Pollution". *Environmental Science & Technology*, 50 (1), p. 313-320.
- Dreiseidler, A., Straub, D., Baumbach, G. (2000), "Korngrößenverteilung (PM₁₀ und PM_{2,5}) von Staubemissionen relevanter stationärer Quellen (Teil II zu 297 44 853)". Abteilung Reinhaltung der Luft (RdL), Institut für Verfahrenstechnik und Dampfkesselwesen (IVD), Universität Stuttgart, Stuttgart. p. 192.
- Droste-Franke, B. (2005), "Quantifizierung von Umweltschäden als Beitrag zu Umweltökonomischen Gesamtrechnungen". Institut für Energiewirtschaft und Rationelle Energieanwendung, Universität Stuttgart. p. 232. Online available at: <u>http://elib.uni-stuttgart.de/opus/volltexte/2005/2204/</u>
- Droste-Franke, B., Krewitt, W., Friedrich, R., Trukenmuller, A., 2005. Attribution of air damages to countries and economic sectors of origin. In: Markandya, A., Tamborra, M. (Eds.). Green accounting in Europe. Edward Elgar Publishing, Celtenham, pp. 226-292. ISBN: 1-84542-114-0.
- Edenhofer, O., Hirth, L., Knopf, B., Pahle, M., Schlömer, S., Schmid, E., Ueckerdt, F. (2013), "On the economics of renewable energy sources". *Energy Economics*, 40, Supplement 1 (0), p. S12-S23.

EGTEI (2006), "Draft Background Document on Combustion sector". Prepared in the framework of EGTEI, Online available at: <u>http://citepaax.alias.domicile.fr/forums/egtei/LCP%20background%20docume</u> <u>nt.pdf</u> EGTEI, 2012. Expert Group on Techno-Economic Issues (EGTEI) BACKGROUND DOCUMENTS - Combustion Installations in energy, transformation industries and industries available at:

http://citepaax.alias.domicile.fr/forums/egtei/egtei_doc-Comb-Ind.htm. Accessed on 2013-01-02.

- Endres, A., Fraser, I.L. (2011), *Environmental Economics*. Cambridge University Press Textbooks. ISBN: 9781280879173.
- EPRI (2001), "Damage to Power Plants Due to Cycling". 1001507. Palo Alto, CA. p. 190. Online available at: <u>http://www.epri.com/abstracts/Pages/ProductAbstract.aspx?ProductId=00000</u> 0000001001507
- European Commission (1995), "ExternE Externalities of Energy Vol. 2 Methodology". European Commission, p. 399.
- European Commission (1999a), "ExternE Externalities of Energy Vol. 7 Methodology 1998 update". Office of Publications for the European Commission, Luxembourg. Online available at: http://www.externe.info/externe_d7/sites/default/files/vol7.pdf
- European Commission (1999b), "ExternE Externalities of Energy Vol. 10 National Implementation". Office for Official Publications of the European Communities, Luxembourg. p. 606. Online available at: http://www.externe.info/reportex/vol10.pdf
- European Commission (2003), "External Costs: Research results on socio-environmental damages due to electricity and transport". Office for Official Publications of the European Communities, Luxembourg. p. 28. Online available at: http://ec.europa.eu/research/energy/pdf/externe en.pdf
- European Commission (2004), "New Elements for the Assessment of External Costs from Energy Technologies (NewExt) - Final Report to the European Commission, DG Research, Technological Development and Demonstration (RTD)". European Commission, Brussels. p. 333.
- European Commission (2005a), "ExternE Externalities of Energy Methodology 2005 Update". Office for Official Publications of the European Communities, Luxembourg. Online available at: http://www.externe.info/brussels/methup05a.pdf
- European Commission (2005b), "Thematic Strategy on air pollution". EC, Brussels, Belgium. p. 13. Online available at: <u>http://eur-</u> lex.europa.eu/LexUriServ/LexUriServ.do?uri=CELEX:52005DC0446:EN:NOT
- European Commission (2006a), "Integrated Pollution Prevention and Control Reference Document on Economics and Cross-Media Effects". EC, p. 159.

- European Commission (2006b), "Reference Document on Best Available Techniques for Large Combustion Plants". European Commission, Joint Research Centre, Institute for Prospective Technological Studies, Seville, Spain. p. 580. Online available at: <u>http://ftp.jrc.es/eippcb/doc/lcp_bref_0706.pdf</u>
- European Commission (2008a), "20 20 by 2020 Europe's climate change opportunity". COM(2008) 30 final. EC, Brussels, Belgium. Online available at: <u>http://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52008DC0030</u>
- European Commission (2008b), "Energy Sources, Production Costs and Performance of Technologies for Power Generation, Heating and Transport". EC, Brussels, Belgium. p. 29. Online available at:

http://ec.europa.eu/energy/strategies/2008/2008 11 ser2 en.htm

European Commission (2009), "Impact Assessment Guidelines - SEC(2009) 92". EC, Bruxelles. p. 50. Online available at: <u>http://ec.europa.eu/governance/impact/commission_guidelines/commission_guidelines_en.htm</u>

European Commission (2011), "Energy Roadmap 2050". COM(2011) 885 final. EC, Brussels, Belgium. p. 20. Online available at: <u>http://eur-lex.europa.eu/legalcontent/EN/TXT/?uri=COM:2011:0885:FIN</u>

European Commission (2013), "Impact Assessment accompanying the documents Communication from the Commission to the Council, the European Parliament, the European Economic and Social Committee and the Committee of the Regions - a Clean Air Programme for Europe". European Commission (EC), Brussels, Belgium. p. 351. Online available at:

http://ec.europa.eu/environment/archives/air/pdf/Impact_assessment_en.pdf

- European Commission, 2014a. The Clean Air Policy Package. available at: <u>http://ec.europa.eu/environment/air/clean_air_policy.htm</u>. Accessed on 2014-07-22.
- European Commission (2014b), "A policy framework for climate and energy in the period from 2020 to 2030 ". COM/2014/015 final. EC, Brussels, Belgium. p. 18. Online available at: <u>http://eur-lex.europa.eu/legalcontent/EN/TXT/?uri=CELEX:52014DC0015</u>
- European Commission (2015a), "Energy Economic Developments Investment perspectives in electricity markets". European Commission, Directorate-General for Economic and Financial Affairs, p. 112. Online available at: <u>http://ec.europa.eu/economy_finance/publications/eeip/ip003_en.htm</u>
- European Commission, 2015b. Energy Efficiency Directive. available at: <u>http://ec.europa.eu/energy/en/topics/energy-efficiency/energy-efficiency-</u> <u>directive</u>. Accessed on 2015-08-06.

European Commission (2016), "Proposal for a Directive of the European Parliament and of the Council amending Directive 2012/27/EU on energy efficiency". EC, Brussels, Belgium Online available at:

http://ec.europa.eu/energy/sites/ener/files/documents/1 en act part1 v16.pdf.

- European Environment Agency (2011), "Revealing the costs of air pollution from industrial facilities in Europe". EEA, Copenhagen, Denmark. p. 72. Online available at: <u>http://bookshop.europa.eu/en/revealing-the-costs-of-air-</u> pollution-from-industrial-facilities-in-europe-pbTHAK11015/
- European Environment Agency (2014a), "Costs of air pollution from European industrial facilities 2008–2012 an updated assessment (EEA Technical report No 20/2014)". EEA, Copenhagen, Denmark. p. 74. Online available at: http://www.eea.europa.eu/publications/costs-of-air-pollution-2008-2012
- European Environment Agency (2014b), "European Union emission inventory report 1990–2012 under the UNECE Convention on Long-range Transboundary Air Pollution (LRTAP)". EEA, Copenhagen, Denmark. p. 130. Online available at: <u>http://www.eea.europa.eu/publications/Irtap-2014</u>
- European Environment Agency, 2015. Overview of the electricity production and use in Europe. available at: <u>http://www.eea.europa.eu/data-and-</u> <u>maps/indicators/overview-of-the-electricity-production/assessment-2</u>. Accessed on 2015-02-05.
- European Environment Agency (2016), "European Union emission inventory report 1990–2014 under the UNECE Convention on Long-range Transboundary Air Pollution (LRTAP) - Annex D: EU gap-filled inventory". EEA, Copenhagen, Denmark Online available at: <u>http://www.eea.europa.eu//publications/lrtap-</u> emission-inventory-report-2016.
- European Parliament, Council of the European Union (2008), "Directive 2008/1/EC of the European Parliament and of the Council of 15 January 2008 concerning integrated pollution prevention and control (Codified version) ". Online available at: <u>http://eur-</u>

lex.europa.eu/LexUriServ/LexUriServ.do?uri=CELEX:32008L0001:EN:NOT.

European Parliament, Council of the European Union (2009), "Directive 2009/28/EC of the European Parliament and of the Council of 23 April 2009 on the promotion of the use of energy from renewable sources and amending and subsequently repealing Directives 2001/77/EC and 2003/30/EC". Online available at: <u>http://eur-lex.europa.eu/legal-</u>

content/EN/ALL/?uri=CELEX:32009L0028&qid=1437572664787.

European Parliament, Council of the European Union (2010), "Directive 2010/75/EU of the European Parliament and of the Council of 24 November 2010 on industrial emissions (integrated pollution prevention and control) (Recast)". Official Journal of theEuropean Union. p. 103. Online available at: <u>http://eurlex.europa.eu/LexUriServ/LexUriServ.do?uri=CELEX:32010L0075:EN:NOT</u>.

- European Parliament, Council of the European Union (2016), "Directive (EU) 2016/2284 of the European Parliament and of the Council of 14 December 2016 on the reduction of national emissions of certain atmospheric pollutants, amending Directive 2003/35/EC and repealing Directive 2001/81/EC (Text with EEA relevance)". Online available at: <u>http://eur-lex.europa.eu/legal-</u> content/EN/TXT/?uri=uriserv:OJ.L .2016.344.01.0001.01.ENG.
- eurostat, 2013. Harmonized Indices of Consumer Prices (HICP) Main tables. available at:

http://epp.eurostat.ec.europa.eu/portal/page/portal/hicp/data/main_tables. Accessed on 2013-07-10.

- eurostat (2015a), "Administrative units / Statistical units (© EuroGeographics for the administrative boundaries)". eurostat Online available at: <u>http://ec.europa.eu/eurostat/web/gisco/geodata/reference-</u> <u>data/administrative-units-statistical-units</u>.
- eurostat (2015b), "Employment rate, by sex (code: tsdec420)". eurostat, Online available at: <u>http://ec.europa.eu/eurostat/tgm/refreshTableAction.do?tab=table&plugin=1</u> &pcode=tsdec420&language=en.
- eurostat (2015c), "Population density by NUTS 3 region (code: demo_r_d3dens)". eurostat, Online available at: <u>http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=demo_r_d3dens&l</u> <u>ang=en</u>.
- eurostat (2016a), "Greenhouse gas emissions by source sector (source: EEA), (env_air_gge)". eurostat Online available at: <u>http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=env_air_gge&lang=</u> <u>en</u>.
- eurostat (2016b), "Harmonized Indices of Consumer Prices (HICP) Main tables". eurostat Online available at: <u>http://ec.europa.eu/eurostat/web/hicp/data/database</u>.
- eurostat (2016c), "Real GDP per capita, growth rate and totals (tsdec100)". eurostat Online available at: <u>http://ec.europa.eu/eurostat/tgm/table.do?tab=table&init=1&plugin=1&langu</u> age=en&pcode=tsdec100.
- eurostat (2017), "Supply, transformation and consumption of electricity annual data (nrg_105a)". Online available at: http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=nrg_105a.
- Finlayson-Pitts, B.J., Pitts, J.N. (1993), "Atmospheric Chemistry of Tropospheric Ozone Formation: Scientific and Regulatory Implications". *Air & Waste*, 43 (8), p. 1091-1100.

- Fountoukis, C., Koraj, D., Denier van der Gon, H.A.C., Charalampidis, P.E., Pilinis, C., Pandis, S.N. (2013), "Impact of grid resolution on the predicted fine PM by a regional 3-D chemical transport model". *Atmospheric Environment*, 68, p. 24-32.
- Geldermann, J., Rentz, O. (2004), "The reference installation approach for the technoeconomic assessment of emission abatement options and the determination of BAT according to the IPPC-directive". *Journal of Cleaner Production*, 12 (4), p. 389-402.
- Grahame, T.J., Schlesinger, R.B. (2007), "Health effects of airborne particulate matter: do we know enough to consider regulating specific particle types or sources?". *Inhal Toxicol*, 19 (6-7), p. 457-481.
- Hallquist, M., Wenger, J.C., Baltensperger, U., Rudich, Y., Simpson, D., Claeys, M., Dommen, J., Donahue, N.M., George, C., Goldstein, A.H., Hamilton, J.F., Herrmann, H., Hoffmann, T., Iinuma, Y., Jang, M., Jenkin, M.E., Jimenez, J.L., Kiendler-Scharr, A., Maenhaut, W., McFiggans, G., Mentel, T.F., Monod, A., Prévôt, A.S.H., Seinfeld, J.H., Surratt, J.D., Szmigielski, R., Wildt, J. (2009), "The formation, properties and impact of secondary organic aerosol: current and emerging issues". *Atmos. Chem. Phys.*, 9 (14), p. 5155-5236.
- Harrison, M. (2010), "Valuing the Future: The Social Discount Rate in Cost-Benefit Analysis". Productivity Commission, Canberra, p. 175. Online available at: http://ssrn.com/abstract=1599963
- Hartley, A., Pekel, J.-F., Ledwith, M., Champeaux, J.-L., Badts, E.D., Bartalev, S.A. (2006), "The Land Cover Map for Europe in the Year 2000". European Commision Joint Research Centre. Online available at: http://forobs.jrc.ec.europa.eu/products/glc2000/glc2000.php.
- HEI (2003), "Assessing Health Impact of Air Quality Regulations: Concepts and Methods for Accountability Research". Health Effects Institute Accountability Working Group, p. 99. Online available at: http://pubs.healtheffects.org/getfile.php?u=261
- Héroux, M.-E., Anderson, H.R., Atkinson, R., Brunekreef, B., Cohen, A., Forastiere, F., Hurley, F., Katsouyanni, K., Krewski, D., Krzyzanowski, M., Künzli, N., Mills, I., Querol, X., Ostro, B., Walton, H. (2015), "Quantifying the health impacts of ambient air pollutants: recommendations of a WHO/Europe project". *Int J Public Health*, 60 (5), p. 619-627.
- Hohmeyer, O. (1988), Social Costs of Energy Consumption. Springer Verlag, Berlin, Germany.
- Hohmeyer, O., Ottinger, R.L., (Eds.) (1994), *Social Costs of Energy Present Status and Future Trends*. Springer, Berlin ; Heidelberg [u.a.]. ISBN: 3-540-57841-2.
- Holland, M. (2014a), "Cost-benefit Analysis of Final Policy Scenarios for the EU Clean Air Package - Version 2, Corresponding to IIASA TSAP Report 11, Version 2a".
 EMRC, p. 67. Online available at: http://ec.europa.eu/environment/air/pdf/TSAP%20CBA.pdf
- Holland, M. (2014b), "Implementation of the HRAPIE Recommendations for European Air Pollution CBA work". EMRC, UK, p. 54. Online available at: <u>http://ec.europa.eu/environment/air/pdf/review/CBA%20HRAPIE%20implement.pdf</u>
- Holland, M., Hunt, A., Hurley, F., Navrud, S., Watkiss, P. (2005a), "Methodology for the Cost-Benefit analysis for CAFE: Volume 1: Overview of Methodology". AEA Technology Environment, Harwell, UK. p. 112.
- Holland, M., Hurley, F., Hunt, A., Watkiss, P. (2005b), "Methodology for the Cost-Benefit Analysis for CAFE: Volume 3: Uncertainty in the CAFE CBA: Methods and First Analysis". AEA Technology Environment, Harwell, UK. p. 53. Online available at: <u>http://www.cafe-cba.org/assets/volume 3 methodology 05-05.pdf</u>
- Holland, M., Pye, S., Watkiss, P., Droste-Franke, B., Bickel, P. (2005c), "Damages per tonne emission of PM2.5, NH3, SO2, NOx and VOCs from each EU25 Member State (excluding Cyprus) and surrounding seas". EMRC, AEA Technology and IER, p. 25. Online available at: <u>http://www.cafecba.org/assets/marginal_damage_03-05.pdf</u>
- Holland, M., Watkiss, P., Pye, S., Oliveira, A.d., Regemorter, D.V. (2005d), "Cost-benefit analysis of policy option scenarios for the Clean Air for Europe programme". AEA Technology, Didcot, United Kingdom. p. 82.
- Holstius, D.M., Pillarisetti, A., Smith, K.R., Seto, E. (2014), "Field calibrations of a low-cost aerosol sensor at a regulatory monitoring site in California". *Atmos. Meas. Tech.*, 7 (4), p. 1121-1131.
- Hunt, A., Navrud, S., Maca, V., Scasny, M. (2011), "Monetary values for health endpoints used in the HEIMTSA/INTARESE Common Case Study - D 4.1.2 of the HEIMTSA project". UBath, p. 10. Online available at: <u>http://www.heimtsa.eu/LinkClick.aspx?fileticket=Z79uJ1ZKuX8%3d&tabid=293</u> <u>7&mid=6403&language=en-GB</u>
- Hunt, A., Navrud, S., Maca, V., Scasny, M. (2013), "Monetary values for health endpoints used in the HEIMTSA/INTARESE Common Case Study D 4.1.2 of the HEIMTSA project". UBath, p. 10. Online available at:
 <u>http://www.heimtsa.eu/LinkClick.aspx?fileticket=Z79uJ1ZKuX8%3d&tabid=293</u> <u>7&mid=6403&language=en-GB</u>
- Hurley, F., Hunt, A., Cowie, H., Holland, M., Miller, B., Pye, S., Watkiss, P. (2005),
 "Methodology for the Cost-Benefit analysis for CAFE: Volume 2: Health Impact Assessment". AEA Technology Environment, Harwell, UK. p. 149.

- IEA (2014), The Power of Transformation Wind, Sun and the Economics of Flexible Power Systems. IEA (International Energy Agency), Paris, France. ISBN: 978 92 64 20803 2.
- IER (2004), "EcoSense 4.0 User's Manual". Institute of Energy Economics and the Rational Use of Energy (IER), University of Stuttgart, p. 54. Online available at: <u>http://ecoweb.ier.uni-</u>

stuttgart.de/ecosense web/ecosensele web/ecosense4um.pdf

- IGN (2013), "GEOFLA® 2011 Communes France Métropolitaine". Institut National de l'Information Géographique et Forestière Online available at: <u>http://professionnels.ign.fr/geofla</u>.
- INSEE (2009a), "Données détaillées des statistiques d'état civil sur les décès en 2009". Institut National de la Statistique et des Études Économiques Online available at: <u>http://www.insee.fr/fr/themes/detail.asp?ref_id=ir-sd20093</u>.
- INSEE (2009b), "Populations légales 2009 des départements et des collectivités d'outremer". Institut National de la Statistique et des Études Économiques Online available at: <u>http://www.insee.fr/fr/ppp/bases-de-</u> <u>donnees/recensement/populations-legales/france-</u> <u>departements.asp?annee=2009#departements</u>.
- INSEE (2009c), "RP2009, exploitation principale, POP1B Population par sexe et âge". Institut National de la Statistique et des Études Économiques Online available at:

http://www.insee.fr/fr/themes/tableau_local.asp?ref_id=POP1B&millesime=20 09&niveau=2&nivgeo=FE&codgeo=1#infos-connexes.

- Institute for Health Metrics and Evaluation, 2014. Global Burden of Disease Study 2010 (GBD 2010) - Ambient Air Pollution Risk Model 1990 - 2010. available at: <u>http://ghdx.healthdata.org/record/global-burden-disease-study-2010-gbd-</u> <u>2010-ambient-air-pollution-risk-model-1990-2010</u>. Accessed on 2014-05-20.
- Institute of Medicine (2013), "Environmental decisions in the face of uncertainty". The National Academies Press, Washington, DC. p. 258. Online available at: <u>http://nap.edu/12568</u>
- IOM (2011), "Final report on risk functions used in the case studies D 3.1.2/3/4 of the HEIMTSA project". Institute of Occupational Medicine (IOM); TNO; JRC, Edinburgh, UK; Delft, The Netherlands; Ispra, Italy. p. 244. Online available at: <u>http://www.heimtsa.eu/LinkClick.aspx?fileticket=a6n5dYwOWqU%3d&tabid=2</u> <u>937&mid=6403&language=en-GB</u>
- IPCC (1996), "Good Practice Guidance and Uncertainty Management in National Greenhouse Gas Inventories - Annex 1: Conceptual Basis for Uncertainty Analysis". Intergovernmental Panel on Climate Change, p. 22. Online available at: <u>http://www.ipcc-nggip.iges.or.jp/public/gp/english/</u>

- IPCC (2014a), "Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (Final draft)". Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA. Online available at: <u>http://mitigation2014.org/report/final-draft/</u>
- IPCC (2014b), "Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change". IPCC, Geneva, Switzerland. p. 151. Online available at: <u>http://www.ipcc.ch/report/ar5/syr/</u>
- JCGM (2008), "Evaluation of measurement data Guide to the expression of uncertainty in measurement". Joint Committee for Guides in Metrology (JCGM), Online available at: www.bipm.org/utils/common/documents/jcgm/JCGM 100 2008 E.pdf
- Johnston, R.J., Boyle, K.J., Adamowicz, W., Bennett, J., Brouwer, R., Cameron, T.A.,
 Hanemann, W.M., Hanley, N., Ryan, M., Scarpa, R., Tourangeau, R., Vossler,
 C.A. (2017), "Contemporary Guidance for Stated Preference Studies". *Journal of the Association of Environmental and Resource Economists*, 4 (2), p. 319-405.
- Joskow, P.L. (2011), "Comparing the Costs of Intermittent and Dispatchable Electricity Generating Technologies". *American Economic Review*, 101 (3), p. 238-241.
- Just, R.E., Hueth, D.L., Schmitz, A. (2004), *The welfare economics of public policy: a* practical approach to project and policy evaluation. Edward Elgar, Cheltenham. ISBN: 1-84376-688-4.
- Kalli, J., Karvonen, T., Makkonen, T. (2009), "Sulphur content in ships bunker fuel in 2015

 A study on the impacts of the new IMO regulations and transportation costs".
 University of Turku, Finland. p. 38. Online available at:
 www.lvm.fi/docs/en/339549 DLFE-8042.pdf
- Kaneyasu, N., Ohta, S., Murao, N. (1995), "Seasonal variation in the chemical composition of atmospheric aerosols and gaseous species in Sapporo, Japan". *Atmospheric Environment*, 29 (13), p. 1559-1568.
- Kim, Y., Sartelet, K., Raut, J.-C., Chazette, P. (2015), "Influence of an urban canopy model and PBL schemes on vertical mixing for air quality modeling over Greater Paris". *Atmospheric Environment*, 107, p. 289-306.
- Kim, Y., Seigneur, C., Duclaux, O. (2014), "Development of a plume-in-grid model for industrial point and volume sources: application to power plant and refinery sources in the Paris region". *Geosci. Model Dev.*, 7 (2), p. 569-585.
- Klingelhöfer, H.E. (2006), *Finanzwirtschaftliche Bewertung von Umweltschutzinvestitionen*. DUV, Wiesbaden, Germany. ISBN: 978-3-8350-9464-2.
- Kolb, J.A., Scheraga, J.D. (1990), "Discounting the benefits and costs of environmental regulations". *Journal of Policy Analysis and Management*, 9 (3), p. 381-390.

- Korsakissok, I., Mallet, V. (2010), "Development and application of a reactive plume-ingrid model: evaluation over Greater Paris". *Atmos. Chem. Phys.*, 10 (18), p. 8917-8931.
- Krall, J.R., Anderson, G.B., Dominici, F., Bell, M.L., Peng, R.D. (2013), "Short-term exposure to particulate matter constituents and mortality in a national study of U.S. urban communities". *Environ Health Perspect*, 121 (10), p. 1148-1153.
- Krewitt, W. (2002), "External costs of energy--do the answers match the questions?: Looking back at 10 years of ExternE". *Energy Policy*, 30 (10), p. 839-848.
- Krewitt, W., Heck, T., Trukenmüller, A., Friedrich, R. (1999), "Environmental damage costs from fossil electricity generation in Germany and Europe". *Energy Policy*, 27 (3), p. 173-183.
- Krewitt, W., Mayerhofer, P., Friedrich, R., Trukenmüller, A., 1997. External Costs of Fossil Fuel Cycles. In: Hohmeyer, O., Ottinger, R.L., Rennings, K. (Eds.). Social costs and sustainability: valuation and implementation in the energy and transport sector. Proceedings of an international conference, held at Ladenburg, Germany, May 27 - 30, 1995. Springer, Berlin, pp. 121-136. ISBN: 3-540-60177-5.
- Krewitt, W., Schlomann, B. (2006), "Externe Kosten der Stromerzeugung aus erneuerbaren Energien im Vergleich zur Stromerzeugung aus fossilen Energieträgern. Gutachten im Rahmen von Beratungsleistungen für das Bundesministerium für Umwelt, Naturschutz und Reaktorsicherheit".
 Deutsches Zentrum für Luft- und Raumfahrt (DLR), Fraunhofer Institut für System- und Innovationsforschung (ISI), Stuttgart, Karlsruhe. p. 59.
- Krewitt, W., Trukenmüller, A., Bachmann, T.M., Heck, T. (2001), "Country-specific Damage Factors for Air Pollutants: A Step Towards Site Dependent Life Cycle Impact Assessment". *The International Journal of Life Cycle Assessment*, 6 (4), p. 199-210.
- Kuik, O. (2007), "The Avoidance Costs of Greenhouse Gas Damage: A Meta-Analysis". Institute for Environmental Studies (IVM), VU University of Amsterdam, Amsterdam. p. 17.
- Kumar, N., Besuner, P.M., Lefton, S.A., Agan, D.D., Hilleman, D.D. (2012), "Power Plant Cycling Costs". NREL/SR-5500-55433. Intertek APTECH, Sunnyvale, California, US. p. 83. Online available at: http://www.nrel.gov/electricity/transmission/western-wind-2.html
- Lecœur, È., Seigneur, C. (2013), "Dynamic evaluation of a multi-year model simulation of particulate matter concentrations over Europe". *Atmos. Chem. Phys.*, 13 (8), p. 4319-4337.
- Lee, C.J., Martin, R.V., Henze, D.K., Brauer, M., Cohen, A., Donkelaar, A.v. (2015), "Response of Global Particulate-Matter-Related Mortality to Changes in Local Precursor Emissions". *Environmental Science & Technology*, 49 (7), p. 4335-4344.

- Legorgeu, C. (2016), "Modélisation à différentes échelles spatiales des rejets atmosphériques de la centrale de Porcheville – version 2". Centre d'Enseignement et de Recherche en Environnement Atmosphérique (CEREA), p. 29.
- Lelieveld, J., Evans, J.S., Fnais, M., Giannadaki, D., Pozzer, A. (2015), "The contribution of outdoor air pollution sources to premature mortality on a global scale". *Nature*, 525 (7569), p. 367-371.
- Levy, J.I., Diez, D., Dou, Y., Barr, C.D., Dominici, F. (2012), "A meta-analysis and multisite time-series analysis of the differential toxicity of major fine particulate matter constituents". Am J Epidemiol, 175 (11), p. 1091-1099.
- Lew, D., Brinkman, G., Ibanez, E., Florita, A., Heaney, M., Hodge, B.-M., Hummon, M., Stark, G., King, J., Lefton, S.A., Kumar, N., Agan, D., Jordan, G., Venkataraman, S. (2013), "The Western Wind and Solar Integration Study Phase 2". NREL/TP-5500-55588. NREL (Golden, Colorado, US), RePPAE, Intertek-APTECH, GE Energy, p. 199. Online available at: <u>http://www.nrel.gov/electricity/transmission/western-wind-2.html</u>
- Likhvar, V.N., Pascal, M., Markakis, K., Colette, A., Hauglustaine, D., Valari, M., Klimont, Z., Medina, S., Kinney, P. (2015), "A multi-scale health impact assessment of air pollution over the 21st century". *Science of The Total Environment*, 514 (0), p. 439-449.
- Maca, V., Scasny, M., Hunt, A., Anneboina, L., Navrud, S. (2013), "Presentation of unit values for health end-points: country-specific and pooled - D4.1.3 of the HEIMTSA project (draft final)". University of Bath, p. 87. Online available at: <u>http://www.heimtsa.eu/LinkClick.aspx?fileticket=XidEPtxYdqs%3d&tabid=2937</u> &mid=6403&language=en-GB&forcedownload=true
- Mallet, V., Quélo, D., Sportisse, B., Ahmed de Biasi, M., Debry, É., Korsakissok, I., Wu, L., Roustan, Y., Sartelet, K., Tombette, M., Foudhil, H. (2007), "Technical Note: The air quality modeling system Polyphemus". *Atmos. Chem. Phys.*, 7 (20), p. 5479-5487.
- Mansfield, C., Sinha, P., Henrion, M. (2009), "Influence Analysis in Support of Characterizing Uncertainty in Human Health Benefits Analysis - Final Report". Prepared for U.S. EPA, Office of Air Quality Planning and Standards, Online available at: <u>https://www3.epa.gov/ttn/ecas/regdata/Benefits/influence_analysis_final_rep</u> ort_psg.pdf
- Markakis, K., Valari, M., Colette, A., Sanchez, O., Perrussel, O., Honore, C., Vautard, R., Klimont, Z., Rao, S. (2014), "Air quality in the mid-21st century for the city of Paris under two climate scenarios; from the regional to local scale". *Atmos. Chem. Phys.*, 14 (14), p. 7323-7340.

- Markandya, A., Ortiz, R.A., 2011. General introduction to valuation of human health risks. In: Nriagu, J.O. (Ed.). Encyclopedia of Environmental Health. Elsevier, pp. 871-878. ISBN: 9780444522733.
- MEEM (2016), "Aide à la décision pour l'élaboration du PREPA Rapport Principal". CITEPA, INERIS, ENERGIES DEMAIN, AJDB, Paris, France. p. 189. Online available at: https://www.ecologique-solidaire.gouv.fr/politiques-publiques-reduirepollution-lair#e5
- Mendelsohn, R. (1980), "An economic analysis of air pollution from coal-fired power plants". *Journal of Environmental Economics and Management*, 7 (1), p. 30-43.
- Miller, B. (2013), "IOMLIFET version 2013 Spreadsheets for life-table calculations". Institute of Occupational Medicine (IOM), Edinburgh, Scotland. p. 14. Online available at: <u>http://www.iom-world.org/research/researchexpertise/statistical-services/iomlifet/</u>
- Miller, B., Hurley, F., Shafrir, A. (2011), "Health Impact Assessment for the National Emissions Ceiling Directive (NECD) – Methodological Issues". Institute of Occupational Medicine (IOM), Edinburgh, Scotland. p. 74. Online available at: <u>http://ec.europa.eu/environment/air/pollutants/pdf/IOM%20Report%20meth</u> <u>odology%20NECD%20HIA.pdf</u>
- Mindell, J., Ison, E., Joffe, M. (2003), "A glossary for health impact assessment". *Journal* of Epidemiology and Community Health, 57 (9), p. 647-651.
- Muller, N.Z., Mendelsohn, R. (2007), "Measuring the damages of air pollution in the United States". *Journal of Environmental Economics and Management*, 54 (1), p. 1-14.
- Navrud, S., Ready, R. (2007), *Environmental Value Transfer: Issues and Methods*. Springer Netherlands. ISBN: 978-1-4020-4081-8.
- Nenes, A., Pandis, S., Pilinis, C. (1998), "ISORROPIA: A New Thermodynamic Equilibrium Model for Multiphase Multicomponent Inorganic Aerosols". *Aquatic Geochemistry*, 4 (1), p. 123-152.
- Neumayer, E. (2010), Weak Versus Strong Sustainability: Exploring the Limits of Two Opposing Paradigms. Edward Elgar Publishing. ISBN: 1849805431.
- Nitsch, J., Pregger, T., Naegler, T., Heide, D., Tena, D.L.d., Trieb, F., Scholz, Y., Nienhaus, K., Gerhardt, N., Sterner, M., Trost, T., Oehsen, A.v., Schwinn, R., Pape, C., Hahn, H., Wickert, M., Wenzel, B. (2012), "Langfristszenarien und Strategien für den Ausbau der erneuerbaren Energien in Deutschland bei Berücksichtigung der Entwicklung in Europa und global". Deutsches Zentrum für Luft- und Raumfahrt (DLR), Stuttgart; Fraunhofer Institut für Windenergie und Energiesystemtechnik (IWES), Kassel; Ingenieurbüro für neue Energien (IFNE), Teltow, p. 331. Online available at: http://www.dlr.de/dlr/Portaldata/1/Resources/bilder/portal/portal_2012_1/lei

tstudie2011_bf.pdf

- Nunge, S. (2001), "Der Referenzanlagenansatz zur Ableitung von Luftreinhaltestrategien

 Ein Ansatz zur techno-ökonomischen Charakterisierung von
 Emissionsminderungsmaßnahmen". VDI Verlag GmbH, Düsseldorf, Germany.
 p. 296.
- Oak Ridge National Laboratory and Resources for the Future (1992), "U.S. EC Fuel Cycle Study: Background Document to the Approach and Issues, Report No. 1". Oak Ridge National Laboratory, Oak Ridge, Tennessee. Online available at: <u>http://web.ornl.gov/info/reports/1992/3445604259157.pdf</u>
- OECD (2002), Handbook of Biodiversity Valuation: A Guide for Policy Makers. Organisation for Economic Co-operation and Development (OECD), Paris. ISBN: 92-64-17579-2, 92-64-19731-1.
- OECD (2012), "Mortality Risk Valuation in Environment, Health and Transport Policies". OECD Publishing, p. 139. Online available at: <u>http://dx.doi.org/10.1787/9789264130807-en</u>
- OECD (2013), Inventory of Estimated Budgetary Support and Tax Expenditures for Fossil Fuels 2013. OECD Publishing.
- OECD (2016), The Economic Consequences of Outdoor Air Pollution. OECD Publishing.
- Ostro, B.D., Rothschild, S. (1989), "Air pollution and acute respiratory morbidity: An observational study of multiple pollutants". *Environmental Research*, 50 (2), p. 238-247.
- Ott, K., 2014. Ethik und Naturschutz. Handbuch Naturschutz und Landschaftspflege. Wiley-VCH Verlag GmbH & Co. KGaA. ISBN: 9783527678471. <u>http://dx.doi.org/10.1002/9783527678471.hbnl1999016</u>.
- Ott, W., Baur, M., Kaufmann, Y., Frischknecht, R., Steiner, R. (2006), "Assessment of Biodiversity Losses, Delivery n° 4.2 - RS 1b of the NEEDS project". econcept AG, Zürich; ESU-services, Uster, p. 138. Online available at: <u>http://www.needsproject.org/docs/results/RS1b_D4.2.pdf</u>
- Pappin, A.J., Mesbah, S.M., Hakami, A., Schott, S. (2015), "Diminishing Returns or Compounding Benefits of Air Pollution Control? The Case of NOx and Ozone". *Environmental Science & Technology*, 49 (16), p. 9548-9556.
- Pearce, D. (2002), "An Intellectual History of Environmental Economics". *Annual Review* of Energy and the Environment, 27 (1), p. 57-81.
- Pearce, D., Atkinson, G., Mourato, S. (2006), *Cost-Benefit Analysis and the Environment*. OECD, Paris. ISBN: 92-64-01004-1.
- Pearce, D.W., Cline, W.R., Achanta, A.N., Fankhauser, S., Pachauri, R.K., Tol, R.S.J.,
 Vellinga, P., 1996. The social costs of climate change: greenhouse damage and the benefits of control. In: Bruce, J.P., Lee, H., Haites, E.F. (Eds.). Climate
 Change 1995: Economic and Social Dimensions of Climate Change. Cambridge
 University Press, Cambridge, pp. 179-224. ISBN: 0-521-56051-9, 0-521-56854-4.

Pigou, A.C. (1912), Wealth and Welfare. London.

- Pigou, A.C. (1932), *The Economics of Welfare (4th edition)*. Library of Economics and Liberty.
- Preiss, P., Friedrich, R., Klotz, V., 2008. External costs database (Euro/ton values) -[D.2.2] of the CASES project. available at: <u>http://www.feem-</u> project.net/cases/documents/deliverables/ExternalCosts per unit emission 0 80821.xls. Accessed on 12/07/09.
- Preiss, P., Klotz, V. (2008), "EcoSenseWeb V1.3 User's Manual & "Description of Updated and Extended Draft Tools for the Detailed Site-dependent Assessment of External Costs"". Institute of Energy Economics and the Rational Use of Energy (IER), University of Stuttgart, Germany. p. 63.
- Quinet, E., Baumstark, L., Bonnet, J., Croq, A., Ducos, G., Meunier, D., Rigard-Cerison, A., Roquigny, Q. (2013), "L'évaluation socioéconomique des investissements publics". Commissariat Général à la Stratégie et à la Prospective, p. 352. Online available at: <u>http://www.strategie.gouv.fr/sites/strategie.gouv.fr/files/atoms/files/cgsp_eva</u> luation socioeconomique 29072014.pdf
- Rabl, A., Holland, M. (2008), "Environmental Assessment Framework for Policy Applications: Life Cycle Assessment, External Costs and Multi-criteria Analysis". *Journal of Environmental Planning and Management*, 51 (1), p. 81-105.
- Rabl, A., Spadaro, J.V. (1999), "Damages and costs of air pollution: An analysis of uncertainties". *Environment International*, 25 (1), p. 29-46.
- Rabl, A., Spadaro, J.V. (2000), "PUBLIC HEALTH IMPACT OF AIR POLLUTION AND IMPLICATIONS FOR THE ENERGY SYSTEM". Annual Review of Energy and the Environment, 25 (1), p. 601-627.
- Rabl, A., Spadaro, J.V., Bachmann, T.M., 2011. Monetary valuation of trace pollutants. In: Nriagu, J.O. (Ed.). Encyclopedia of Environmental Health. Elsevier, pp. 856-869. ISBN: 9780444522733.
- Rabl, A., Spadaro, J.V., Holland, M. (2014), How Much is Clean Air Worth? Calculating the Benefits of Pollution Control. Cambridge University Press, Cambridge, United Kingdom. ISBN: 978-1-107-04313-8.
- Rabl, A., Spadaro, J.V., van der Zwaan, B. (2004), "Uncertainty of Air Pollution Cost Estimates: To What Extent Does It Matter?". *Environmental Science & Technology*, 39 (2), p. 399-408.
- Ready, R., Navrud, S., Day, B., Dubourg, R., Machado, F., Mourato, S., Spanninks, F., Rodriquez, M. (2004), "Benefit Transfer in Europe: How Reliable Are Transfers between Countries?". *Environmental and Resource Economics*, 29 (1), p. 67-82.
- Reiss, R., Anderson, E.L., Cross, C.E., Hidy, G., Hoel, D., McClellan, R., Moolgavkar, S. (2007), "Evidence of health impacts of sulfate-and nitrate-containing particles in ambient air". *Inhal Toxicol*, 19 (5), p. 419-449.
- Rentz, O. (1979), *Techno-Ökonomie betrieblicher Emissionsminderungsmaßnahmen*. Erich Schmidt Verlag, Berlin, Germany. ISBN: 3-503-01824-7.

- Ricardo AEA (2017), "IED Derogation Cost-Benefit Analysis Tool". UK Environment Agency. Online available at: <u>https://www.gov.uk/government/publications/industrial-emissions-directive-derogation-cost-benefit-analysis-tool</u>.
- Samuelson, P.A., Nordhaus, W.D. (2007), Volkswirtschaftslehre: das internationale Standardwerk der Makro- und Mikroökonomie. mi-Fachverl., Redline. ISBN: 9783636031136.
- Savitz, D.A. (1988), "Human Studies of Human Health Hazards: Comparison of Epidemiology and Toxicology". *Statistical Science*, 3 (3), p. 306-313.
- Schröder, A., Kunz, F., Meiss, J., Mendelevitch, R., von Hirschhausen, C. (2013), "Current and Prospective Costs of Electricity Generation until 2050 ". DIW, Reiner Lemoine Institute, Berlin University of Technology, Berlin, Germany. p. 94. Online available at: http://www.diw.de/sixcms/detail.php?id=diw_01.c.424588.de
- Schultmann, F., Jochum, R., Rentz, O. (2001), "A methodological approach for the economic assessment of best available techniques demonstrated for a case study from the steel industry". *The International Journal of Life Cycle Assessment*, 6 (1), p. 19-27.
- Schulz, M., Gauss, M., Benedictow, A., Jonson, J.E., Tsyro, S., Nyıri, A., Simpson, D., Steensen, B.M., Klein, H., Valdebenito, A., Wind, P., Kirkevag, A., Griesfeller, J., Bartnicki, J., Olivie, D., Grini, A., Iversen, T., Seland, Ø., Semeena, V.S., Fagerli, H., Aas, W., Hjellbrekke, A.-G., Mareckova, K., Wankmuller, R., Schneider, P., Solberg, S., Svendby, T., Liu, L., Posh, M., Vieno, M., Reis, S., Kryza, M., Werner, M., Walaszek, K. (2013), "Transboundary acidification, eutrophication and ground level ozone in Europe in 2011 - EMEP Status Report 01/2013". 1504-6109. Norwegian Meteorological Institute, Norway. Online available at: http://emep.int/publ/reports/2013/EMEP_status_report_1_2013.pdf
- Schwermer, S. (2007), "Ökonomische Bewertung von Umweltschäden -Methodenkonvention zur Schätzung externer Umweltkosten". Umweltbundesamt, Dessau. p. 115.
- Schwermer, S., Preiss, P., Müller, W. (2014), "Best-Practice-Kostensätze für Luftschadstoffe, Verkehr, Strom- und Wärmeerzeugung - Anhang B der "Methodenkonvention 2.0 zur Schätzung von Umweltkosten"". Umweltbundesamt (UBA), Dessau-Roßlau, Germany. p. 44. Online available at: <u>https://www.umweltbundesamt.de/publikationen/methodenkonvention-20zur-schaetzung-von-1</u>
- Scott, V., Gilfillan, S., Markusson, N., Chalmers, H., Haszeldine, R.S. (2013), "Last chance for carbon capture and storage". *Nature Clim. Change*, 3 (2), p. 105-111.
- Sensfuß, F., Ragwitz, M., Genoese, M. (2008), "The merit-order effect: A detailed analysis of the price effect of renewable electricity generation on spot market prices in Germany". *Energy Policy*, 36 (8), p. 3086-3094.

- Silver, J.D., Christensen, J.H., Kahnert, M., Robertson, L., Rayner, P.J., Brandt, J. (2016), "Multi-species chemical data assimilation with the Danish Eulerian hemispheric model: system description and verification". *Journal of Atmospheric Chemistry*, 73 (3), p. 261-302.
- Smeets, E., Weterings, R. (1999), "Environmental indicators: Typology and overview (Technical report No. 25)". European Environment Agency (EEA), Copenhagen.
 p. 19. Online available at: <u>http://www.eea.europa.eu/publications/TEC25</u>
- Snyder, E.G., Watkins, T.H., Solomon, P.A., Thoma, E.D., Williams, R.W., Hagler, G.S.W., Shelow, D., Hindin, D.A., Kilaru, V.J., Preuss, P.W. (2013), "The Changing Paradigm of Air Pollution Monitoring". *Environmental Science & Technology*, 47 (20), p. 11369-11377.
- Solow, R.M. (1974), "Intergenerational Equity and Exhaustible Resources". *Review of Economic Studies*, 41 (5), p. 29.
- Spadaro, J.V. (2004), "RiskPoll Manual and Reference Documentation (Version 1.05) : Impact assessment tools to estimate the health and environmental risks from exposure to routine atmospheric emissions". p. 39.
- Spadaro, J.V., Rabl, A. (2002), "Air pollution damage estimates: the cost per kilogram of pollutant". International Journal of Risk Assessment and Management, 3 (1), p. 75-98.
- Spadaro, J.V., Rabl, A. (2008), "Estimating the uncertainty of damage costs of pollution: A simple transparent method and typical results". *Environmental Impact Assessment Review*, 28 (2-3), p. 166-183.
- Sportisse, B. (2007), "Modélisation et simulation de la pollution atmosphérique". Université Pierre et Marie Curie. p. 81. Online available at: https://www.inria.fr/content/download/12971/432646/version/2/file/Bruno.S portisse-hdr.fr.pdf.
- Stanek, L.W., Sacks, J.D., Dutton, S.J., Dubois, J.-J.B. (2011), "Attributing health effects to apportioned components and sources of particulate matter: An evaluation of collective results". *Atmospheric Environment*, 45 (32), p. 5655-5663.
- Steinbach, J., Staniaszek, D. (2015), "Discount rates in energy system analysis -Discussion Paper". Fraunhofer ISI, Buildings Performance Institute Europe (BPIE), p. 18. Online available at: <u>http://bpie.eu/wpcontent/uploads/2015/10/Discount rates in energy systemdiscussion_paper_2015_ISI_BPIE.pdf</u>
- Steven, M. (1994), "Produktion und Umweltschutz : Ansatzpunkte f
 ür die Integration von Umweltschutzma
 ßnahmen in die Produktionstheorie".Zugl.: Bielefeld, Univ., Habil.-Schr., 1992. Wiesbaden. p. 214. ISBN: 3-409-13767-X.
- Swart, R., Bernstein, L., Ha-Duong, M., Petersen, A. (2009), "Agreeing to disagree: uncertainty management in assessing climate change, impacts and responses by the IPCC". *Climatic Change*, 92 (1), p. 1-29.

- Tarrasón, L. (2009), "Report on deliveries of source-receptor matrices with the regional EMEP Unified model - Technical Paper n° 1.2 - RS 1b of the NEEDS project". Norwegian Meteorological Institute (MET.NO), p. 11.
- TEEB (2010), "The Economics of Ecosystems and Biodiversity: Mainstreaming the Economics of Nature: A synthesis of the approach, conclusions and recommendations of TEEB". UNEP, p. 36.
- TFTEI (2015), "Estimation of costs of reduction techniques for LCP methodology". Task Force on Techno Economic Issues, p. 85. Online available at: <u>http://tftei.citepa.org/images/files/2015-02-</u> <u>16/TFTEI cost calculation methodology.pdf</u>
- Thompson, T.M., Saari, R.K., Selin, N.E. (2014), "Air quality resolution for health impact assessment: influence of regional characteristics". *Atmos. Chem. Phys.*, 14 (2), p. 969-978.
- Tombette, M., Sportisse, B. (2007), "Aerosol modeling at a regional scale: Model-to-data comparison and sensitivity analysis over Greater Paris". *Atmospheric Environment*, 41 (33), p. 6941-6950.
- Torfs, R., Hurley, F., Miller, B., Rabl, A. (2007), "A set of concentration-response functions - Final report (RS1b D3.7) for the NEEDS project". Mol, BE. Edinburgh, UK. Paris, FR. p. 162. Online available at: <u>http://www.needsproject.org/docs/results/RS1b/NEEDS_Rs1b_D3.7.pdf</u>
- Turner, R.K. (2007), "Limits to CBA in UK and European environmental policy: retrospects and future prospects". *Environmental and Resource Economics*, 37 (1), p. 253-269.
- U.S. Energy Information Administration (2015), "Annual Energy Outlook 2015 with Projections to 2040". U.S. Department of Energy, Washington, DC. p. 154. Online available at: <u>http://www.eia.gov/forecasts/aeo/index.cfm</u>
- U.S. EPA (2002), "EPA Air Pollution Control Cost Manual (6th edition)". Office of Air Quality Planning and Standards, United States Environmental Protection Agency, p. 752.
- U.S. EPA (2009), "Integrated Science Assessment for Particulate Matter (Final Report)". U.S. Environmental Protection Agency, Washington, DC. p. 2228. Online available at: http://www.epa.gov/ncea/pdfs/partmatt/Dec2009/PM ISA full.pdf
- U.S. EPA (2010), "Guidelines for Preparing Economic Analyses". National Center for Environmental Economics, Office of Policy, US EPA, Online available at: <u>http://yosemite.epa.gov/ee/epa/eerm.nsf/vwAN/EE-0568-50.pdf/\$file/EE-0568-50.pdf</u>
- U.S. EPA (2011a), "The Benefits and Costs of the Clean Air Act from 1990 to 2020: Final Report - Rev. A". U.S. Environmental Protection Agency: Office of Air and Radiation, p. 238. Online available at: <u>http://www.epa.gov/air/sect812/feb11/fullreport_rev_a.pdf</u>

- U.S. EPA (2011b), "Exposure Factors Handbook: 2011 Edition". National Center for Environmental Assessment, Washington, DC. Online available at: <u>https://cfpub.epa.gov/ncea/risk/recordisplay.cfm?deid=236252&CFID=705691</u> 24&CFTOKEN=28555954
- U.S. EPA (2012a), "BenMAP User's Manual". Office of Air Quality Planning and Standards, p. 272. Online available at: http://www.epa.gov/air/benmap/models/BenMAPManualOct2012.pdf
- U.S. EPA (2012b), "BenMAP User's Manual Appendices". Office of Air Quality Planning and Standards, p. 255. Online available at: <u>http://www.epa.gov/airquality/benmap/models/BenMAPAppendicesOct2012.</u> <u>pdf</u>
- U.S. EPA (2012c), "Regulatory Impact Analysis for the Final Revisions to the National Ambient Air Quality Standards for Particulate Matter". Office of Air Quality Planning and Standards, United States Environmental Protection Agency, p. 474. Online available at: http://www.epa.gov/ttnecas1/regdata/RIAs/finalria.pdf
- UBA (Ed.), 2008. Economic Valuation of Environmental Damage: Methodological Convention for Estimates of Environmental Externalities. Umweltbundesamt (UBA), Dessau. p. 85. <u>http://www.umweltdaten.de/publikationen/fpdf-</u> <u>I/3482.pdf</u>.
- Ueckerdt, F., Hirth, L., Luderer, G., Edenhofer, O. (2013), "System LCOE: What are the costs of variable renewables?". *Energy*, 63, p. 61-75.
- Umweltbundesamt (Ed.), 2012. Ökonomische Bewertung von Umweltschäden: Methodenkonvention 2.0 zur Schätzung von Umweltkosten. Umweltbundesamt (UBA), Dessau. p. 74. <u>https://www.umweltbundesamt.de/publikationen/oekonomische-bewertung-von-umweltschaeden-0</u>.
- van der Kamp, J., Bachmann, T.M. (2015), "Health-Related External Cost Assessment in Europe: Methodological Developments from ExternE to the 2013 Clean Air Policy Package". *Environmental Science & Technology*, 49 (5), p. 2929-2938.
- van der Kamp, J., Bachmann, T.M., Preiss, P., Sutter, D., Bieler, C. (forthcoming),
 "Methodenkonvention 3.0 Weiterentwicklung und Erweiterung der Methodenkonvention zur Schätzung von Umweltkosten, Sachstandspapier AP 1: Umweltkosten durch Luftschadstoffemissionen". EIFER und INFRAS, im Auftrag des Umweltbundesamtes (UBA), Dessau-Roßlau, Germany.
- VGB, EURELECTRIC (2010), "European Wide Sector Specific Calculation Method for Reporting to the European Pollutant Release and Transfer Register - VGB / EURELECTRIC Recommendations, 2nd Edition". VGB European Working Group "EPRTR", p. 45.

- von Böventer, E. (1995), "Einführung in die Mikroökonomie 8. vollständig neu bearbeitete und erweiterte Auflage". Oldenbourg, München Online available at: <u>https://epub.ub.uni-muenchen.de/3424/1/3424.pdf</u>.
- Walton, H., Dajnak, D., Beevers, S., Williams, M., Watkiss, P., Hunt, A. (2015),
 "Understanding the Health Impacts of Air Pollution in London". King's College London, London, UK. p. 129. Online available at:
 <u>https://www.london.gov.uk/sites/default/files/HIAinLondon_KingsReport_140</u> 72015_final_0.pdf
- Walz, R., 2009. Ethische Herausforderungen für Umweltökonomen. In: Maring, M. (Ed.). Verantwortung in Technik und Ökonomie. Universitätsverlag Karlsruhe (Schriftenreihe des Zentrums für Technik- und Wirtschaftsethik am Karlsruher Institut für Technologie 1), Karlsruhe, Germany, pp. 185-206.
 ISBN: 978-3-86644-296-2.
- Watkiss, P. (2007), "Deliverable 12a: Final Technical Report on Methods of the project METHODEX (Methods and data on environmental and health externalities: harmonising and sharing of operational estimates, EU FP6, contract number: GOCE-CT-2003-505368 - METHODEX)". AEA Technology Environment, Harwell. p. 291. Online available at: http://www.methodex.org/Methodex 20Deliverable 2012 5Fv4.pdf
- Watkiss, P., Pye, S., Holland, M. (2005), "CAFE CBA: Baseline Analysis 2000 to 2020". AEA Technology Environment, Didcot, Oxon, UK. p. 112. Online available at: http://www.cafe-cba.org/assets/baseline_analysis_2000-2020_05-05.pdf
- Weather Research and Forecasting Model, 2015. WRF Users Page WRF Model Version 3.6. available at: <u>http://www2.mmm.ucar.edu/wrf/users/wrfv3.6/wrf_model.html</u>. Accessed on 2015-02-19.
- WHO (1999), "Health Impact Assessment: main concepts and suggested approach".
 WHO, European Centre for Health Policy, Brussels, Belgium. p. 10. Online available at: <u>http://www.apho.org.uk/resource/view.aspx?RID=44163</u>
- WHO (2007), "Health relevance of particulate matter from various sources Report on a WHO Workshop, Bonn, Germany, 26-27 March 2007". World Health Organization, p. 21. Online available at: http://www.euro.who.int/ data/assets/pdf file/0007/78658/E90672.pdf
- WHO (2008), "Part 1: Guidance Document on Characterizing and Communicating Uncertainty in Exposure Assessment - Harmonization Project Document No.
 6.". World Health Organization, Geneva, Switzerland. p. 138. Online available at:

http://www.who.int/ipcs/methods/harmonization/areas/uncertainty%20.pdf

- WHO (2013a), "Health risks of air pollution in Europe HRAPIE project Recommendations for concentration–response functions for cost–benefit analysis of particulate matter, ozone and nitrogen dioxide". World Health Organization, Regional Office for Europe, p. 54. Online available at: <u>http://www.euro.who.int/en/health-topics/environment-and-health/airquality/publications</u>
- WHO (2013b), "Review of evidence on health aspects of air pollution REVIHAAP
 Project. Technical Report". The WHO European Centre for Environment and
 Health, Bonn, WHO Regional Office for Europe, Copenhagen, Denmark. p. 302.
- WHO, 2014. European detailed mortality database (DMDB). available at: <u>http://www.euro.who.int/en/data-and-evidence/databases/european-</u> <u>detailed-mortality-database-dmdb2</u>. Accessed on 2014-05-20.
- WHO, 2016. Databases. available at: <u>http://www.euro.who.int/en/data-and-evidence/databases</u>. Accessed on 2016-03-30.
- WHO Regional Office for Europe, OECD (2015), "Economic cost of the health impact of air pollution in Europe: Clean air, health and wealth". WHO Regional Office for Europe, Copenhagen. p. 54. Online available at: http://www.euro.who.int/en/media-centre/events/2015/04/ehp-mid-term-review/publications/economic-cost-of-the-health-impact-of-air-pollution-in-europe
- Zhang, Y., Sartelet, K., Zhu, S., Wang, W., Wu, S.Y., Zhang, X., Wang, K., Tran, P., Seigneur, C., Wang, Z.F. (2013), "Application of WRF/Chem-MADRID and WRF/Polyphemus in Europe – Part 2: Evaluation of chemical concentrations and sensitivity simulations". *Atmos. Chem. Phys.*, 13 (14), p. 6845-6875.
- Zhuang, J., Liang, Z., Lin, T., Guzman, F.D. (2007), "Theory and Practice in the Choice of Social Discount Rate for Cost-Benefit Analysis: A Survey". Asian Development Bank (ADB), Manila, Philippines. p. 40. Online available at: <u>https://www.adb.org/sites/default/files/publication/28360/wp094.pdf</u>



Exposure to ambient air pollution increases the risk for humans of developing diseases as well as of dving prematurely, thus imposing costs on society. Welfare-oriented policy-making aims at integrating these costs into market prices, e.g. by reducing harmful atmospheric emissions from industrial sources. Against this background, social cost-benefit analysis (CBA) serves to identify the "optimal" level of pollution control. Social CBA related to air pollution is well developed in the public policy context. Yet, approaches at the level of industrial point emission sources are rare and existing models have important shortcomings. This work aims at enhancing economic methods for social CBA of emission control measures at site level, notably by developing a new health damage cost assessment framework and by evaluating the influence of selected methodological aspects.

Temporal and spatial modelling resolution, in conjunction with emission source characteristics and population distribution, are shown to be key site-dependent influencing factors on health damage costs. More generally, methodological choices regarding the assessment of premature mortality from long-term particulate matter exposure are particularly influential. The results underline the importance of methodological guidance from official bodies to avoid arbitrary choices in social CBA, thus facilitating the integration of health damage costs into decision-making.

Gedruckt auf FSC-zertifiziertem Papier